Conformity in the selected print media content as a threat to democracy: The case of the Schabir Shaik trial Patience Thembeka Ntuli **June 2010 Prof Nixon Kariithi** (Supervisor) A thesis submitted to the Faculty of Humanities, University of the Witwatersrand, Johannesburg, in fulfilment of the Degree of Master of Arts in Journalism and **Media Studies**

· i

DECLARATION

I declare that this research report submitted for the degree of Master of Arts in Journalism and Media Studies at the University of the Witwatersrand, Johannesburg is my own original work and has not been submitted to any other institution of higher education or for any examination in any form. I further declare that all sources cited or quoted are indicated and acknowledged by means of a comprehensive list of references.

| Patience Thembeka Ntuli | Date |
|-------------------------|------|

Copyright University of the Witwatersrand 2010

· ii

ACKOWLEDGEMENTS

To my God, thank you, Lord, for always keeping your promise of "never leaving nor forsaking me" throughout the time of writing this thesis. Lord, this is once again the living testimony that you are God. Thank you for allowing me to continue to enjoy your blessings, everlasting love and protection.

I wish to extend my heartfelt gratitude to my supervisor, Prof Nixon Kariithi, for his continued support and encouragement throughout this challenging period and for ensuring that I complete this project. Prof, without your expertise and experience, this project would never have seen the light.

I am indebted to my employer, College of Human Sciences UNISA, for all the study leave, financial support and encouragement. Special thanks must go to Professor Mokgethi Motlhabi; Professor Rosemary Moeketsi; Professor Kuzvinetsa Dzvimbo; Professor Greg Curbertson and Professor Oupa Mashile. I am particularly grateful to Professor de Kock and Dr Barry Hanyane for providing constructive criticism of my work.

This dissertation would also not have been possible without the following people: my librarian at Unisa, Dawie Malan without whose support I would never have been able to juggle around the database; Marlette van der Merwe for the initial edit and Lyn Voigt for the final edit.

I acknowledge the prayers and support of all family members, mentors and friends. Thank you for allowing me to spend most of my time with my books. May God bless and protect all of you. Mama, it is at times like these that I miss you the most. Your teachings, especially the one about the importance of education continue to carry me. I wish you were here to "brag about me"! I miss you but I find joy in knowing that you love me and would have been very proud of me.

· iii

Dedication.

To the lovely blessing from God, my daughter *Sphesihle Chuleza Nomahlwani Mpapama*, who was born a week after I had received the news about passing the MA.

· iv

UNIVERSITY OF WITWATERSRAND, JOHANNESBURG ABSTRACT

$\mathbf{B}\mathbf{y}$

Thembeka Ntuli

It generally appears that headlines, story content, photographs and sources in different South African newspapers are often the same. This conformity poses a major threat to democracy as diversity in all facets of the media is important in order to help the public make informed choices. Using the case of the Schabir Shaik trial, the study investigated the extent of conformity in the news output of selected newspapers: *The Citizen, The Star, Sowetan and Natal Witness* from January 2001 to June 2005. Only selected front page stories and editorials of the newspapers were studied.

Thematic content analysis, quantitative and qualitative methods were used to explore the subject. The study was predominantly informed by theories of news production which posit that news is not the actual truth; it is in fact a selective representation of the world. The notion of objectivity was extensively debated and it was observed that complete "objectivity" as an ideal is unattainable especially when one considers that news-making processes are complex and influenced by a number of diverse factors. The study also argues that sources play an important role in shaping news content because they set parameters and define terms of reference. It emerged in the results that the sharing of news and reliance on similar sources across publications breed homogeneity; that most news is covered in an episodic/dramatic fashion for the purpose of profit and that objectivity is frequently not practised in newspaper editorials. The study concludes that media should promote diversity in all facets and strive for balance when covering stories.

v

"They look like newspapers; they feel like newspapers; they even leave ink on your finger tip. But they are not really newspapers" (Guy Berger, *Sunday Times*, 1 May 2005, p.3)

· vi

TABLE OF CONTENTS

| | | PAGE |
|------------------|--|-------|
| DEC | LARATION | (ii) |
| ACKNOWLEDGEMENTS | | (iii) |
| | MICATION | (iv) |
| | TRACT | (v) |
| | | () |
| СНА | APTER 1 INTRODUCTION | 1 |
| 1.0 | INTRODUCTION | 1 |
| 1.1 | Chapter Outline | 1 |
| 1.2 | Historical Background of the Schabir Shaik Trial | 3 |
| 1.3 | Research Questions | 5 |
| СНА | APTER 2 MEDIA AND DEMOCRACY | 6 |
| 2.0 | INTRODUCTION | 6 |
| 2.1 | Defining the Concept of Democracy | 6 |
| 2.2 | The Role of the Media in a Democracy | 8 |
| 2.3 | The Concept of Public Sphere | 13 |
| | 2.3.1 Criticism of the Habermasian theory | 15 |
| 2.4 | African Media and Democracy as they relate to the Public Sphere. | 19 |
| 2.5 | Conclusion | 23 |
| СНА | APTER 3 NEWS PRODUCTION AS A SOCIAL CONSTRUCTIO | N OF |
| REA | LITY | 24 |
| 3.0 | INTRODUCTION | 24 |
| 3.1 | Sociology of News Production | 24 |

· vii

| | 3.1.1 | News framing and values | 27 |
|-------|---|--|----|
| 3.2 | Socialisation of Media Workers | | |
| 3.3 | Reliance on Sources | | |
| 3.4 | Objectivity in the Media | | |
| | 3.4.1 | The issue of balance | 40 |
| 3.5 | Gatekeeping Theory | | 41 |
| 3.6 | Media Ownership | | 42 |
| 3.7 | Influence of Advertisers on Content | | 44 |
| 3.8 | Commercialisation | | 46 |
| | 3.8.1 | Market driven journalism | 47 |
| | 3.8.2 | Tabloid journalism | 50 |
| 3.9 | Concl | usion | 50 |
| | | | |
| CHAI | PTER 4 | THE SOUTH AFRICAN PRESS IN CONTEXT | 51 |
| | | | |
| 4.0 | INTRO | ODUCTION | 51 |
| 4.1 | Histor | y of South Africa Press | 53 |
| | 4.1.1 | English press | 57 |
| | 4.1.2 | Afrikaans press | 57 |
| | 4.1.3 | The Black press | 59 |
| 4.1.4 | The Pa | rotest press/Alternative press | 60 |
| 4.2 | Newsp | paper Firms in South Africa | 62 |
| | 4.2.1 | Naspers | 63 |
| | 4.2.2 | Independent News and Media | 63 |
| | 4.2.3 | Avusa Limited | 65 |
| | 4.2.4 | CTP/Caxton Publishers and Printers LTD | 66 |
| | 4.2.5 | M&G Media | 66 |
| 4.3 | Challe | inges Faced by South African Media, and Developments | 67 |
| 4.4 | Media State Relationship after Apartheid 69 | | 69 |
| 4.5 | Conclusions | | 71 |

· viii

| CHA | HAPTER 5 METHODOLOGY | | |
|-----|---|-----|--|
| 5.0 | INTRODUCTION | 72 | |
| 5.1 | Methodology | 72 | |
| 5.2 | Thematic Analysis | 76 | |
| 5.3 | Discourse Analysis | 77 | |
| 5.4 | Time Frame | 79 | |
| 5.5 | Selection of Articles | 79 | |
| 5.6 | Data Collection | 80 | |
| 5.7 | Conclusion | 81 | |
| СНА | APTER 6 FINDINGS AND DISCUSSION | 82 | |
| 6.0 | INTRODUCTION | 82 | |
| 6.1 | Data Collection and Data Analysis | 84 | |
| | 6.1.1 Analysis of editorials | 87 | |
| | 6.1.2 Analysis of front page articles | 94 | |
| | 6.1.3 Analysis of sources used in the front page articles | 100 | |
| | 6.1.4 Analysis of metaphors | 103 | |
| | 6.1.5 Analysis of news values/frames | 106 | |
| 6.2 | Conclusions | 107 | |
| | | | |
| CHA | APTER 7 CONCLUSION | 109 | |
| 7.0 | INTRODUCTION | 109 | |
| 7.1 | Conclusions on Literature Review | 109 | |
| 7.2 | Conclusion on Methodology | 113 | |
| 7.3 | Conclusions on Findings | 114 | |
| 7.4 | Limitations of the Study | | |
| 7.5 | On Further Research | 118 | |

· ix

LIST OF TABLES

| 3.1 | Values in news | 30 |
|-------|--|-----|
| 4.1 | Independent Newspapers | 64 |
| 5.1 | Readership profile of the four newspapers studied | 78 |
| 6.2 | Overall coverage of the Schabir Shaik trial by the four newspapers studied | 87 |
| | | |
| BIBLI | OGRAPHY | 120 |
| APPEN | NDICES | 134 |

 \cdot X

CHAPTER 1

CONFORMITY IN THE SELECTED PRINT MEDIA CONTENT AS A THREAT TO DEMOCRACY: THE CASE OF THE SCHABIR SHAIK TRIAL¹

1.0 Introduction

With the end of apartheid after 1990, the new South African Interim Constitution explicitly guaranteed press freedom with the view that this would encourage diversity of ownership and therefore diversity of opinion. Over the past few years the researcher has observed incidents in which different South African newspapers have come up with headlines, story contents, and photographs and used sources that are strikingly similar². Schabir Shaik trial has been chosen because of the following reasons: it is arguably South Africa's biggest post-apartheid corruption trial. The Schabir Shaik trial was the subject of intense media attention owing to the involvement of several high profile members of the South African government. The number of newspapers referred to it as a biggest court case in a democracy. It has had an influence in the sucking in of the deputy president, Jacob Zuma and is having a negative impact on his (Jacob Zuma's) political career and lastly the very same trial has also led to the biggest division within the African National Congress in its 95-year history. It is therefore interesting for the researcher to use it as the case study for this project. The Schabir Shaik trial was punted to be the trial of the decade. The Star, 06 Oct 2004:12, editorial. And was labelled the same by Natal Witness (02 June 2005:1).

South African media often comes out with close, sometimes even identical headlines, and uses the same pictures for stories especially in major news stories. For example, when the South African swimming team won Gold, the same picture was used in all four of the following newspapers: *The Star, Sowetan, The Citizen*, and *Business Day*. Also, when

1

Jacob Zuma was cleared of rape charges on the 9 May 2006, three newspapers had strikingly similar headings and pictures. Most newspapers used the picture of him and his daughter. This poses a major threat to democracy because the mass media in any modern society is central to the process of democratisation, as it represents a major resource for effective citizenship without which people cannot make informed decisions that are crucial to democracy. The central theme of liberal democracy is the need for diversity in all facets of the media (Croteau & Hoynes, 2003:98).

Ideally, responsible journalism calls for media that provides the public with diverse items of information, thereby enabling them (the public) to make informed choices. According to liberal theory, meaningful political discourse can only take place if the public is provided with a wide range of information (Curran 2002:217). The major argument about newspapers is that they pander to the lowest common denominator of public taste; they "generally fail to provide information that helps citizens make informed political judgments" (Strelitz and Steenveld 2005:265).

Continuing the debate, McQuail (1994:71) adds that media freedom calls for originality of the press, diversity in the news supply, critical stance of the press and independence of the media. When observing the way in which the media covers some of the big stories, for example, the Schabir Shaik corruption trial, it generally appears that there is a diversity decline, specifically in the print media content, and this poses a threat not only to the independent functioning of the media, but also to democracy at large.

According to Curran (1991:92), some newspapers venture into business to make a profit and not to provide news as expected, but one needs to consider the fact that media outlets are not the same as other businesses. By law, media has an important mandate from government not only to provide information, but also to be critical in doing so (Curran, 1991:92). The question is then: To what extent are journalists fulfilling that role if their motive is now more focused on profit?

In an effort to answer this question, Curran (1991:99) makes an important contribution by introducing a professional responsibility model which postulates that the journalist's first duty is to serve the public. This entails ignoring all the internal and external forces that may threaten good journalism. In view of all the pressures facing journalists, one can agree with Curran (1991:87) that media outlets are, therefore, not independent watchdogs of the state but are instead "self-seeking" and using their muscle to promote their interest

Graber (2003) argues that journalists do not find the mandate given to them by government compelling, and this is why there has been a decrease in stories where a journalist would, for example, investigate and find facts on his/her own. Nowadays, journalists wait for leaks, tips and news packs from public relations officials, something that is not supposed to happen in a democratic country like South Africa. As a result, stories end up becoming similar across all newspapers; this is the key focus of this study.

In view of the background provided, the main problem to be addressed by this study is conformity in the selected print media content. Using the Schabir Shaik trial, the study will critically analyse the extent of content conformity in the daily newspapers: namely, *The Citizen, The Star, Sowetan, and Natal Witness.* Second, the study will explore the implications of such conformity for democracy in South Africa. Third, it is expected that the findings will suggest strategies for increasing diversity in the selected print media content³.

1.1 Research Questions

Three research questions arise in relation to this topic:

1. To what extent did the selected print media in South Africa conform in the coverage of the Schabir Shaik trial?

³ Citizen, Sowetan, The Star, and Natal Witness were used. The selection of these three newspapers was influenced by the fact that they are South Africa's biggest national daily newspapers in circulation and, more importantly, are from different stables. Their availability on the Sabinet online database was also an influencing factor.

- 2. What are the implications of such conformity for democracy?
- 3. What can be done to increase content diversity in the print media?

1.2 Historical Background of the Schabir Shaik Trial

The historical background of the story is important in order to ensure that readers of this study understand why this particular study was chosen. The trial of Schabir Shaik⁴ began in a packed court in Durban and first made headlines in January 2001 when The *Mail and Guardian* (Nov 16-22, 2001, p.2) revealed that Durban businessman, Schabir Shaik, was the brother of Chippy Shaik, a government arms procurer, Jacob Zuma's struggle comrade and financial adviser, and a director of the company that had won a R400 million tender in the arms deal. This indicated how the watchdog function of the media can assist in fighting corruption by government officials.

One of the reasons for choosing the Schabir Shaik trial was that it was subject to intense media coverage owing to the involvement of several high-profile members of the South African government. In October 2001, Schabir Shaik was arrested by the Scorpions and charged with the theft of Cabinet documents, including minutes containing information on the arms deal procurement process. He pleaded not guilty to both the charges of fraud and corruption. The matter was then referred to the regional court for trial on the 27th of May 2002. During the trial, Durban businessman Schabir Shaik, sometimes charming, sometimes volatile and always impeccably dressed, arrived on time every morning. On occasion he greeted everybody around him.

Though Shaik claimed that his financial dealings were legitimate, he asserted in his plea explanation that the State had misunderstood his dealings with Zuma. He agreed that there was a financial relationship between Zuma and him, but it was not a corrupt one as payments he made to Zuma were "loans made in friendship". In spite of all his claims, on

. 4

-

⁴ The Schabir Shaik trial was one of the most prominent court trials in post-apartheid South Africa. The case, tried in the Durban High Court before Hilary Squires, proved the fraudulent and corrupt relationship between Durban-based businessman Schabir Shaik and South African politician and the anti-apartheid leader, Jacob Zuma.

the 30th May 2005, the Durban High Court handed down its final judgment. He was pronounced guilty of corruption for paying Jacob Zuma R1.2 million to further their relationship and for soliciting a bribe from French arms company Thomson-CSF, as well as guilty of fraud for writing off more than R1 million of Zuma's unpaid debts. Judge Squires sentenced Shaik to two terms of 15 years for corruption and one term of three years for fraud, to be served concurrently.

While Jacob Zuma was not in court, the judge found that he and Schabir Shaik had a "generally corrupt relationship". Following the decision, Zuma stepped down from his seat in parliament but remained the president of the ANC. This case attracted more media attention as Schabir Shaik attempted appeals at the Supreme Court of Appeal. However, all five judges under President Craig Howie unanimously rejected his appeal and agreed that the correct sentence had been given. Furthermore, Schabir Shaik was later given parole for medical reasons. On the 7th of April 2009, this was followed by the National Prosecuting Authority (NPA) officially dropping all the charges against Jacob Zuma who was later elected as the South African president⁵.

1.2 Chapter Outline

Chapter one provides a general introduction to the study, the problem statement, and the description of the scope and overall objective of the study. The chapter contains a brief motivation for choosing print media (newspapers) and for using the Schabir Shaik trial as the case study.

Chapter two presents the **t**heoretical framework and literature review. This chapter looks at the relationship between the two concepts: media and democracy. In the first section, a detailed definition of democracy is provided, followed by a discussion on the role of media in democracy. The third section discusses the concept of public sphere in detail as a platform through which citizens can actively take part in a democracy. The chapter ends with a section that provides a detailed discussion of African media and democracy as they

5

⁵ http://en.wikipedia.org/wiki/Schabir_Shaik_trial

relate to the concept of public sphere.

Chapter three covers news production theories. It presents an analysis of different theoretical views of the sociology of news production theories, as well as other relevant media theories. It argues that there are various factors that threaten democracy. It posits that the supply of information is considered to be the "media's most important function". It is, therefore, important to determine what is considered as news and how it supplies the information processes involved in this process of supplying information. The major argument is that news has been constructed by the media. This chapter provides a detailed analysis of the concept of the objectivity of news.

Chapter four outlines the history of SA print media and posits that historical discussion of the press is important because it provides a broader political and economic context in which the South African media operated. The chapter notes that it is characterised by different newspaper firms that are largely conglomerates. It also notes that before democracy, the media was more focused on fighting apartheid.

Chapter five discusses the research methodology. It examines content analysis as the research method employed in this study. The focus is the justification of the use of this particular method instead of others.

Chapter six presents the findings and data analysis of the study. It covers the integration of data collected, as well as a review and analysis of the same. This chapter also provides answers to key research questions posed at the beginning of the study. It emerges that sharing of news and reliance on similar sources across publications breed homogeneity. Also emergent is that most news is covered in a more episodic fashion. The analysis of the editorials reveals that objectivity is not practised by some newspapers.

Chapter seven concludes and provides recommendations for this research. Conclusions based on the findings of the investigation as well as the literature review are presented. The report establishes that media should be encouraged to promote diversity of voices;

editorials should remain objective and not be partisan. Recommendations regarding increasing diversity in the media content are made. Possible research projects that could emanate from this research are suggested.

7

CHAPTER 2

MEDIA AND DEMOCRACY

2.0 Introduction

Mass media plays an important role in society – hence it is always referred to as a fourth estate that checks on government (Bennet 1982:32; Nwanko 2000:3; Curran 2002:217 and Wasserman and de Beer 2005:37). Mass media shapes public opinion, but it is also influenced and manipulated by different interest groups in society. It must also be noted that mass media can be used to promote democracy; it is actually referred to as being central to democracy. However, the very same media can pose a threat to the healthy functioning of democracy, in that it can cripple democratic consolidation.

This chapter looks at the relationship between the two concepts: media and democracy. In the first section, a detailed definition of democracy is provided, followed by a discussion on the role of media in democracy. The third section discusses the concept of public sphere in detail as a platform through which citizens can take part actively in a democracy. The chapter ends with a section that provides a detailed discussion of the African media and democracy as they relate to the concept of public sphere.

2.1 Defining the Concept of Democracy

Dahl (2005:191) states that for many it is very difficult to understand what is meant by the concept of democracy. To prevent confusion, it is therefore important for the researcher to begin this chapter by defining what is meant by the concept of democracy.

Various definitions are provided in this regard. Keane (1991:168) posits that the struggle to control the definition of democracy is an intrinsic feature of modern societies. Yet, democracy is not a word that can be made to mean whatever one chooses it to mean. Keane (1991:168) defines democracy as a concept that "comprises procedures for

8

arriving at collective decisions in a way which secures the fullest possible and qualitatively best participation of interested parties". As democracy requires citizens to take part, it is important for citizens to be capacitated to engage in the decision-making process.

Continuing with definitions, Dahl (1989:112) asserts that "each citizen ought to have adequate and equal opportunities for discovering and validating the choice on a matter to be decided that would best serve the citizens' interests". Dahl (1989:109) also points out that "throughout the process of making binding decisions, citizens ought to have an adequate opportunity and equal opportunity to express their preferences". What is noted here is that the quality of citizen participation is as important as participation itself.

Dahl (2005:187) and Jacobs (2002:10) note that democracy is a concept that requires broader popular inclusion and effective participation in government and political life. Dahl (2005:188) further states that large-scale democracy requires the following criteria: elected officials; free, fair and frequent elections; freedom of expression; alternative sources of information; associational autonomy and inclusive citizenship. Garnham (1986:250) points out that for proper democracy to survive, public participation is important but acknowledges that it is impossible and undesirable to require all citizens to participate or debate.

Steenveld (2002:119) and Splichal (1993:4) comment that "democracy denotes a form of government that ensures to the people political equality and involvement in decision making about public affairs".

It is clear from all these definitions that the literature does not offer only one definition of the concept of democracy. Common to all the provided definitions is that democracy is based on the understanding that there should be active participation of citizens. It has also been proved that media plays a fundamental role in the workings of democracy (Dahl 1971:2 & Keane 1991:168).

A discussion of the role of the media is central to this study for several reasons. Any critique of any publication needs to be based on pre-determined performance criteria so that it may be easy to determine whether the publication is fulfilling its role as expected. The next section discusses the role that must be played by the media in a democracy and most of the issues discussed will be revisited later in the study.

2.2 The Role of the Media in a Democracy

Wasserman and De Beer (2005:37) state that the role of the media in a democracy is a topic surrounded by many controversies and involving much debate, as it is understood differently by different role players. According to Wasserman and De Beer (2005:37), all the discrepancies in defining the role of the media in a democracy are largely attributed to the fact that scholars often approach this issue from different theoretical viewpoints. These approaches/viewpoints may be functional, critical or comparative evaluations.

The role of the media is outlined using two theories: namely, the liberal pluralist approach and the political economy approach. The two theories posit different expectations of the role played by the media. The liberal-pluralist approach tends to stress the importance of the media's watchdog role while the political-economy approach focuses on the media as an instrument of the public sphere.

According to the liberal theory, one of the primary roles of the media in a democratic country is to act as a watchdog (Curran 2002:217). Contemporary democratic theory appreciates the watchdog role in ensuring that the government is held accountable for all its actions. It is argued that the government cannot be held accountable if citizens are ill-informed about the actions of officials and institutions. This role, therefore, requires the media to act as a guardian of the public interest, warning the citizens against those who are doing them harm⁶.

⁶ http://safrica.info/ess info/sa glance/constitution/news.htm

It must be mentioned that for the media to perform this role effectively, it must be independent of the government and of all other pressures that might influence its operation. It is feared that once the media is subject to state regulation, it becomes difficult to fulfil the watchdog role (Curran 2002:217). The watchdog role requires that the media not only "monitor the full range of state activities but also fearlessly expose abuses of those in power" (Wasserman and De Beer 2005:37). Additionally, the media must ensure that the government and politicians do not abuse the office power granted to them or exceed their mandates.

The watchdog role of the media ensures that the media keep a strong eye on any public process especially elections. During elections, many things can go wrong and one of those is rigging by some of the political parties or those in government. (Tettey 2001:7) notes that the media's watchful eye helps to minimise if not eliminate rigging and bring transparency to the process to ensure that elections are conducted freely and fairly. This normally increases the credibility of the results and is essential in ensuring credible democracy.

Continuing the argument, Curran (2000:121) posits that the watchdog role overrides all the other important functions of the media and, despite various criticisms, the watchdog role is still considered in most of the literature as an important role for media in any democracy. Having said that, Curran warns readers not to read too much into this view – it is simply based on the position in the 18th century when all media was largely controlled by the press.

Wasserman and De Beer (2005:37) hold similar views on the importance of the watchdog role of the media but argue that this function no longer lies with the state, but with the media itself. Continuing the debate, they argue that the media cripples its own functioning by focusing too much on light news so that sales can be increased. Secondly, it is a conglomeration and homogenisation, and that can only result in poor news content. Lastly, the media no longer exercises its independence because it is faced with the

responsibility of adhering to management practices (Wasserman and De Beer 2005:37).

The second key function of the media for the healthy functioning of democracy according to the liberal theory, is to facilitate debate and provide necessary information so that the public may make informed decisions (Curran 2000:128, Baker 2002:155). Oosthuizen (2001:447) agrees that supplying information is the media's second most important function. As a tool for information, media also helps the public make informed choices by providing a forum for public debates.

Curran (2002:225) posits that it is the duty of the media to ensure that there is a free flow of information between the governed and the government. Thus, mass media plays an important role in terms of linking the government and the public. Murdock (1993:21) agrees that in order for citizens to exercise their full rights, they must have access to the information, advice, and analysis that enables them to know what their personal rights are and allows them to pursue them effectively.

Historically, mass media has fulfilled a variety of functions: namely, information, education, mobilisation and entertainment. It is expected that some functions are more important than others. Stokke (1971:11) argues that mass media plays an important informational role in that it has helped to develop countries. He adds that it then becomes important for scholars to ask questions such as who gets to decide the relevant importance of functions and who decides which function the media should perform, for example, whether the mass media should perform a watchdog, political or administrative function.

Pillay (2004:169) asserts that the "mass media in liberal democracies is often seen as a conduit of information and ideas that empowers ordinary citizens". However, the media does not always do this successfully. To a large extent the media often neglects the views, struggles and aspirations of the poor and marginalised. In any democratic country the media is expected to provide diversity in the news and information. Meaningful political discourse can only take place if the public is provided with a wide range of

information. As a tool for information the media helps the public to make informed choices by providing a forum for public debate. Access to information is essential to the healthy functioning of democracy, because it ensures that citizens make well-informed decisions. For any democratic country the media is an important body that must provide diversity in the news and information (Pillay 2004:169).

Following on from the role of providing information and analysis, the media is also expected to interpret and analyse information for citizens. In a democracy ordinary citizens, according to Waldhal (2004:13), are not always in a position to understand and interpret what is happening in the political sphere. Journalists are sometimes required to elucidate complicated matters to a level that may be understood by ordinary citizens. The researcher has decided to discuss these two roles together because there is usually a very thin line between the two and they are normally debated in the same context.

Linked with the above role, the media also has a duty to ensure that the public engages in public debate and discussion normally by devoting columns to letters from readers and giving prominence to editorials and other commentaries. Such practice ensures that the public play some form of role in the agenda-setting of the political discourse. Some newspaper agencies, however, tend to allow mostly influential people to air their opinions. Such actions have implications in that often such individuals use the publicity to push their own agendas.

The continuous investigation by scholars and researchers into what constitutes high quality journalism is not merely aimed at planting the seed of doubt in the media, but such criticism is aimed at ensuring that democracy is not short-lived (Harber 2004:79). McNair (2000:197) argues that one cannot run away from the fact that journalism is being blamed for undermining the integrity of democracy because some are not truly independent in their functioning. From this one can ask the question: What does the independence of the media mean in a young democracy?

The answers to the above questions need to begin with what we mean by the independence of the media. It is generally assumed that independence means media that functions without any interference from government. During the apartheid period, there was a great deal of government interference when it came to the media performing its watchdog role. Interference took place in the form of censorship and the closing down of some newspapers.

Most of the government intervention was to a large extent caused by the fact that many newspapers were receiving government subsidies, thus making them vulnerable to manipulation. Gillwald (1993:74) posits that subsidies tend to lay publications open to manipulation. Sparks (1992:48) argues that "[j]ust as subsidies do not necessarily mean government intervention in the content of the press, neither does its absence guarantee non-interference". News organisations must, therefore, move toward financial viability so that they can be sure of their independence. A study on media in Central and Eastern Europe and the newly independent states of the former Soviet Union recommends that news organisations be given training in financial management as well as assistance in setting up advertising and business departments (South Africa 2007). The study states: "Media owners need forums to work without arrangements such as circulation audits, advertising rates and production and distribution networks".

It is also important to note that these problems do not exist only in Africa. It is for that reason that McNair (2000:198) maintains that while it is clear that the media has not yet reached its highest level of operation as compared with that of other democratic countries such as Great Britain, "the picture is far from as bleak as some critics portray it to be". Most of the international audits conducted, such as the 2001 Mayne and Chil Study in Germany, and Drok's 2002 survey in the Netherlands, show that poor functioning of the media is a worldwide trend. The findings indicate a lack of practical skills among reporters and journalism teachers not being involved in the actual practical practice.

⁷ Freedom House, Media responses to corruption in emerging democracies: Bulgaria, Hungary, Romany, Ukraine p.10 http://freedomhouse.org/reports/mediatxt.html.

The fact that practising journalists are not receiving enough training is also another reason for concern (Steyn and De Beer, 2004:387). Training is important in ensuring that journalists fulfil their tasks effectively in a democracy. Journalists need to be taught how to move away from reliance on sources such as press releases and press conferences. Journalists must learn how to write with depth and insight. It generally appears that most of these accusations are being aimed at the print media, but McNair (2000:198) maintains that the broadcast media is obviously also guilty of the same charge.

Given all the developments and changes discussed in this research report, perhaps it is important to agree with Berger (2001:172) when he states that the media has played a significant role in terms of providing a forum, promoting an informed citizenry and a common public sphere. In case of the print media, Berger (2001:173) posits that indeed the media has helped in terms of keeping government accountable. It is, however, safe to say that the role of the print media in terms of providing diverse opinions is still highly debatable. The next section deals with the concept of public sphere as a useful concept in understanding the role of the media in a democracy.

2.3 The Concept of Public Sphere

A good starting point for understanding the democratic role of the media is considering the 1962 work of Jurgen Habermas. The concept of public sphere is largely rooted in the history of capitalism. Habermas wrote the first text on the public sphere in 1969. It, however, was not translated until 1989. Calhoun (1992) and Dalgreen were the first critiques. Keane and Thompson came later. Public Sphere is considered useful in understanding the role of the media in democracy. Dahlgren (2002:196) conceptualises public sphere as "a realm of social life where exchange of information and views on questions of common concern can take place so that public opinion can be obtained".

Public sphere as a concept is also known as public space in which opinion is formed (Curran 1991). Keane (1993:77) defines sphere as a

particular type of spatial relationship between two or more people, usually connected by certain means of communication (television, radio, satellite, fax, telephone etc.), in which non-violent controversies erupt, for a brief or more extended period of time, concerning power relations operating within their given milieu of interaction and/or within the wider milieux of social and political structures within which the disputants are situated.

Dahlgren (2002:196) posits that "public sphere takes place when citizens, exercising the right of assembly and association, gather as public bodies to discuss issues of the day, specifically those of political concern".

Continuing with the definition, Calhoun (1992:21) asserts that the concept of public sphere is made up of "private individuals gathering together as a public and articulating the needs of society with the state". Both Schudson (1992:147) and Calhoun (1992:2) concur that public sphere becomes a reality only when "private individuals assemble to form a public body". Habermas (1974:49) defines "public body" as that body consisting of citizens who have the freedom to express and publish their opinions about matters of general interest". Curran (2000:134) states that it is important to note that these private individuals in the public sphere debate matters of common interest.

Thompson (1993:257) points out that the concept of public sphere is largely an "ideology conception". He further notes that "it is based on the idea that individuals come together in a shared locale and engage in a dialogue with one another, as equal participants in a face-to-face conversation". McKee (2005:9) posits that the public sphere enables society to indicate whether they are happy with government or not. Furthermore, the public sphere enables ordinary citizens to understand what their role is, especially in the matter of the running of government.

Dahlgren (1995:9) claims that democracy requires participation and participating entails mechanisms or channels that could be used by the people in voicing opinions and formulating public agenda. He posits that public sphere takes place when citizens,

exercising the rights of assembly and association, gather as public bodies to discuss issues of the day. Elsewhere, Dahlgren (2002:198) sees the concept of public sphere as the "fulfilment of communicational requirements of a viable democracy

Given all the above definitions, it appears that the concept of public sphere can best be understood as a platform whereby people come together, discuss issues, form public opinion and shape policy preferences.

2.3.1 Criticism of the Habermasian theory

Media scholars have criticised the public sphere concept over what they see as its critical failures. The entire book by Calhoun entitled *Habermas and Public sphere* (1992) criticises Habermas on various issues. Calhoun (1992:38) raises four broad concerns about the Habermasian theory. First, he argues that Habermas fails to address issues of power relations: "... the networks of communication, the topography of issues, and the structure of influence of the public sphere except in very general terms of faction and parties". Second, he posits that Habermas overestimates degeneration of the public sphere. He believes that things are not as negative as Habermas suggests in his theory. Third, (1992:38) he claims that the concept of public sphere fails to address various issues in the theory, particularly the issues of power relations. Calhoun (1992:37) adds that the unified public sphere is actually more relevant to white and rich men. He "takes issue with Habermas's neglect of social movements by his conforming too closely to the liberal bourgeois ideal in imagining the public sphere simply as a realm into which individuals bring their ideas and critiques. He argues that social movements are crucial to reorientating the agenda of public discourse, bringing new issues to the fore".

Habermas's study of the media as the public sphere was based on the middle class public sphere of the 18th century. This involved private individuals who used to gather and discuss matters of public importance. According to Habermas (1989:118), only men and property-owning private individuals were admitted to the public sphere. Dahlgren confirms that in the past the public sphere was inclusive of men and took place only in certain settings such as salons, coffee houses and clubs (2002:197). Thompson

(1993:253) also notes that the bourgeois public sphere was based on the principle of universal access. In practice this theory was restricted to individuals who had education and financial means and was dominated by males.

This provoked a lot of criticism from feminists. Fraser (1990) brings our attention to the exclusion of women from the official public sphere and the privatisation of gender politics. For her, Harbermas's account reveals mostly a bourgeois, masculinist conception of the public sphere which also subordinated workers, people of colour, and gays and lesbians. Other critics argue that the exclusion of women was not a contingent historical circumstance but was constitutive of the very nature of the public sphere (Thompson 1993:254).

Calhoun (1992:3) further criticises Habermas when he argues that Habermas's demands for equal and unrestricted participation is in a way paradoxical because in its early days, the public sphere was mainly restricted to the elite and to educated males.

Dahlgren (2002:195) also criticises Habermas for excluding other segments of society. According to him, the public sphere operated largely in the sophisticated media such as intellectual journals, pamphlets and newspapers which could only be read by the educated segment. This is not good for democracy because true democracy is about the public participation of citizens regardless of their class (Dahlgren 2002:196). He (2002:198) offers three areas when conceptualising the public sphere: namely, institutions, media representation, socio-cultural interaction. Dahlgren (2002:199) posits that the dimension of media institution and that of media representation receive more attention because both are the centre of policy debate.

The dimension of social structure refers to the "political ecology" of the media, setting boundaries for the media's institutional and organisational profile. The dimension that the public sphere cannot be seen as a space operating in isolation from all other social, political and economic domains, as if it were a self-contained entity, is one of Habermas's central points.

The last dimension that links well with social structure is socio-cultural interaction. It generally refers to "non-mediated, face-to-face encounters between citizens, to relevant aspects of subjectivity and identity processes, and also to the interface of media and citizens, that is, the process of reception" (Dahlgren 2002:198).

In his defence, Habermas (1992:152), in a detailed note in Structural Transformation of the Public Sphere, expresses his appreciation to all scholars, not only for the interest shown in his work, but for critical insights provoked by it. The Structural Transformation of the Public Sphere (1989) offers a historical narrative of the changing forms of public life which is, in many ways, quite compelling. Continuing his response, he concedes that his earlier account would have to be substantially revised today. He further acknowledges that there is a need for a "more flexible approach to popular movements and popular cultural forms". Structural Transformation of the Public Sphere has been widely criticised (Calhoun 1992, Fraser, 1992, Garnham, 1992, Schudson 1992) Harbermas has attended to many concerns raised by critics in his recent works but Thompson (1993:253) finds it intriguing that Habermas, although informed about the marginalisation of women, still does not give this issue full attention in his latest work, Structural Difference.

Hallin (1994:2) also rejects Habermas's conception. He opposes Habermas's portrayal of the history of journalists as the decline of the golden age of the public sphere. Dahlgren (2005:199) also came out with a book addressing some of the problems pointed out by various critics. Supporting Dahlgren's viewpoints are Calhoun (1992), Hallin (1994) and Fraser (1990) who state that Habermas overlooked many issues.

The other problem of Habermas's conception of public sphere concerns refeudilisation of the public sphere. Thompson (1993:255) defines refeudilisation as "showiness characteristics of mediated politics today, and its concern to cultivate personal aura rather than critical debate". In his theory, Harbermas fails to acknowledge the impact that the mass media has in the modern world. Thompson (1993:252-53) notes that the adequacy and plausibility of Habermas's account of the emergence of the bourgeois public sphere

without acknowledging that during the 18th and 19th centuries there were other forms of public discourses is a problem. She posits that the bourgeois public sphere "was from the outset, embedded in the field of confidential social relations which shaped its formation and development". Thompson argues that popular social movements were very important during the 18th century and it is a pity that even in the *Structural Transformation* work, Harbermas concentrated largely on the "liberal model".

The Habermas concept fails to acknowledge the importance of mass media in public debates and for initiating discussions about matters of general concern (Grbesa (2004:115). The Habermas concept has failed to acknowledge that media structures are central to democracy. In his concept, Habermas actually sees the mass media as "distorting the public sphere and only providing false impressions of it". (Dahlgren 2002:195). Supporting this notion, Elliot (1982:262) concurs that print was the medium that underpinned the concept of public sphere by providing an arena for politics.

In his revised work, Habermas (1992:163-65) recognises the new forms of communication such as radio and television but the role of the media in the public sphere is not properly outlined and he does not acknowledge new forms of communication such as television chat shows, panel discussions and other debate shows. He, however, argues that all the mentioned forms are in no way comparable to the critical rational debate that was constituted in the bourgeois public sphere. Thompson (1993:257) notes that this conception does not outline the importance of the kinds of communication that have been established and sustained by the media and does not outline clearly that the press assisted in the creation of the public sphere. Instead it portrays the press as an entity that was simply interwoven into the life of clubs and coffee houses.

Elliot (1982:262) argues that the newspapers celebrated by Habermas were engines of propaganda for the bourgeoisie rather than the embodiment of disinterested rationality and their version of reason was challenged by radical papers whose other project was developing a set of ideas that generalised the interest of a class excluded from the political system.

Despite all the criticism directed against the concept of public sphere, the literature reviewed generally shows that the Habermasian conception of public sphere is still essential in understanding the role that must be played by the media in democracy. Authors like Garnham (1992:394) still believe that the concept of public sphere provides "a sounder basis for critical analysis of current developments both in the media and democratic politics and for the analysis and political action necessary to rebuild systems of both communication and the concept of public sphere touches on the importance of exchanging views and discussing issues of importance so that decisions can be made" Media such as newspapers, radio and television played as a forum in discussing these matters. Dahlgren (2002:198) supports Garnham's viewpoint, arguing that the concept of public sphere is still valid and helpful because it touches on relevant issues.

Based on all the above arguments, it can generally be concluded that the concept of public sphere is still essential in providing direction for the role that must be played by the media within democracy (Berger 2000:25). The next section discusses African media and democracy as they relate to the concept of public sphere. It must, however, be noted that literature in the African press is scant compared to that focusing on other parts of the world.

2.4 African Media and Democracy as they relate to the Public Sphere

This study requires the researcher to conceptualise the role of the media from the African media perspective. Some argue that there is a need for a different approach to the role of the media in Africa. According to authors like Curran (1991:82) "new times call for new thinking". From this one can ask whether the concept of public sphere can be used when analysing the role of the media in a democracy within the African media context. The following section is an attempt to answer these questions.

The analysis of African media as it relates to the concept of public sphere is very important. Berger (2002:31) posits the following:

[t]he analysis of the public sphere in African countries therefore needs to take into account not just who is included and with what effects ... Africa's media does not exist in a sealed compartment but includes much international content ... from outside newsmakers, content suppliers and donor funders.

From this it is, therefore, important to ask the question: Who then dominates the public sphere? And who determines what issues to discuss in the public sphere? The issue of power relations again comes to the fore. Note, however, that Calhoun (1992:38) argues that the concept of public sphere fails to address various issues, particularly the issues of power relations.

In Africa the main problem that the media is faced with is the decline of the public sphere. Factors such as government interference in media operations have played a role in the decline of the public sphere. In his analysis of challenges facing the African media, Kupe (2005:194) posits that African media is faced with various problems such as journalists' ability to report perspective or views or alternatives. Media has not facilitated meaningful debate. One can say in a democracy, media is expected to encourage citizens to take part in meaningful debates. This is called participatory communication. Kupe (2005:195) defines participatory communication as the "ability of the media to play two critical roles: to provide the public with information and analysis, and to open forums for debate and discussion".

It is important to note that in Africa there are countries where government policies still inform how the media must function. For example, Tahsoh (2000:141) asserts that the Cameroonian government's interference with the media is important because the press is seen as a powerful tool. He argues there is a strong fear that the media will serve as a disruptive force if left uncontrolled (Tahsoh 2000:141).

Ramaprasad (2001:539-555) develops a profile of journalists in post-independence Tanzania, and argues that the concept of "nation building" is crucial to understanding the

country's fledgling media sector. He further points out that the Tanzanian government

believes that all resources – including the mass media – should play an important role in realising this goal. In Tanzania, media is responsible for "initiating discussions on any subject relevant to the development of a socialist and democratic society" (Ramaprasad 2001:541).

Continuing the debate, Ramaprasad (2001:541) claims that in Tanzania, independence of the media is crucial and notes that "the press must be free to report, comment and criticise any faults in society without any government intervention". Kasoma takes a different view, contending that government interference is needed in Africa. He believes that the existence of "media unprofessional conduct including the use of newspapers as political opposition, bad advocacy journalism and overuse of anonymous sources" requires the government's interference so that the media can be more responsible (Kasoma 1997:296-7).

Writing mainly in reference to Kenya, Mak' Ochieng (1993) posits that most literature on democracy and media in Africa borrows too heavily on the outdated four theories of Siebert et al. (1956). According to him, this should not happen because the four theories are largely criticisms. He asserts that the concept of public sphere is the viable alternative for Kenya and other African countries.

Fombad (2002:650) argues that Botswana's public service media "face many challenges particularly from competing and abundant alternative sources of information made available by independent institutions that are part of the broader democratisation process designed to render government more transparent, accountable and closer to the people". Jacobs (2002:1-15) also finds that South Africa's new democracy has resulted in many developments, such as media freedom, freedom to criticise the government, unprecedented access to state-held information and the breaking of the state's monopoly over broadcasting. But Jacobs warns that some critical issues in the mass media were still being overlooked, and cites conformity as a concern for democracy.

Similarly, Zegeye and Harris (2002:239) posit that South Africa is a highly diverse society that is continually undergoing "complex, far-reaching and multifaceted" change in which mass media are expected to play a significant role. The role of the media is not only to provide public information but also to provide a forum for public debate about the matters of public interest for the purpose of the formation of public opinion. Thus, media plays an important role in a democracy. According to Zegeye and Harris, South Africa is a country rich with groups with diverse interests. These different groups with their interests are represented by mass media in the public sphere.

Second, citizens must have access to the broadest possible range of information, interpretation and debate on areas that involve public political choice. Third, citizens must be able to recognise themselves and their aspirations in the range of representations on offer within the central communications sectors. Murdock (1993:21) argues that in order to meet the above criteria, communication systems need to be diverse and open to all citizens. The question is how then to organize the communication systems so as to extend diversity and openness and thereby provide adequate resources for full citizenship.

Another factor that contributes to the decline of public spheres in Africa is that in some countries, some citizens cannot afford media resources. Language is also a problem, because language inadequacy in South Africa, for instance, limits the citizens from participating in public debates. This is mainly because most of South Africa's new media especially the print media are published in English or Afrikaans. Some people cannot even read. According to Megwa (2001:282), this creates a perception that "whites make informed political decisions and intelligently challenge the government whereas blacks wallow in ignorance and blindly support the government even when it blunders".

It must be noted that in Africa, Habermas's notion of public sphere becomes undermined because there is a concentration of media ownership, a dominance of advertising and public relations which sells ideas instead of debating them (Gillwald 1993:67). It is, however, equally important to note that the concept of the transformation of the public

sphere is still important and applicable.

Common to most African countries are the concerns from various authors about the decline of the public sphere, owing to government interference in countries like South Africa. It is, however, noted that in countries like Cameroon and Tanzania, government intervention is considered important because it addresses issues such as bad journalism and overuse of anonymous sources can also be properly addressed if the media is accountable to government (Kasoma 1997:297).

2.5 Conclusion

This chapter noted that media can play a positive role in democracy, only if there is an enabling environment to do so. This chapter also indicated that journalists and writers need to be trained so that they can perform the in-depth reporting that democracy requires. It argued that the public sphere is important and serves as a necessary part of a democratic society. The chapter also highlighted the importance of media's role in a democracy and noted that active participation of citizens is important in a properly functioning democracy. It also became apparent in this chapter that there is a need for a more scholarly attention insofar as the role of the media in democracy is concerned. The next chapter looks at the theories of news production, as well as some of the influences that cause conformity in the selected print media.

CHAPTER 3 NEWS PRODUCTION AS A SOCIAL CONSTRUCTION OF REALITY

3.0 Introduction

The previous chapter outlined the role that is expected to be played by the media in a democracy and the concept of democracy was defined in detail. The section on public sphere illustrated that public space is important in ensuring active participation in a democracy. If news plays such an important role in our culture, then journalists and their work deserve some form of scrutiny so that we may understand the process through which news is produced. This chapter begins by discussing the sociology of news production. It argues that there are other factors that threaten democracy. Oosthuizen (2001:447) posits that supply of information is considered to be the "media's most important function". It is, therefore, important to determine what is considered as news and how the information processes involved in the process of supplying information function. The major argument is that news is constructed by the media (Schudson 1992:151-155). The other argument in this chapter is that news is not an objective reflection of reality.

3.1 Sociology of News Production

The study is predominantly informed by theories of news production. The underlying assumption of the social construction theory is that news is not the actual truth; it is, in fact, a selective representation of the world (Schudson 2003:3). This assumption is supported by Williams (2003:165) who states that newspapers give reports as seen through the eyes of their staff, and docketed, assessed and interpreted by the news workers. Simply put: journalists "remain messengers when reporting news" (Schudson 2003:28). Manning (2001:50) has a similar view when he posits that news does not mirror reality but is instead more of a manufacturing and fabrication process. Elaborating

on this view he argues that journalists do not deliberately fabricate stories or lies but the production of news involves the routine gathering and assembling of certain constituent elements which are fashioned to construct an account of the particular news events.

Sociologists hold a view that news is merely the product of certain working arrangements. Fishman (1982:219-20) offers a clear description of this view.

The practical accomplishment perspective says that journalists' routine methods for producing news – that is the process of newsgathering – construct an image of reality. In distortion of reality because either of these characterisations implies that news can record what is out there. News stories if they reflect anything, reflect the practices of the workers in the organisations that produce news.

The social construction theory posits that the practice of news journalism is far from mirroring reality but it is more of a manufacturing and fabrication process (Manning 2001:50). How then is news produced on a day-to-day level? Theorists suggest that there is no such thing as "reality" (Croteau and Hoynes 2007). The social construction approach to news has stimulated a major debate between social scientists and journalists. Sociologists hold the opinion that news is not truth, but a reality that has been manufactured by the journalists (Schudson 2000:176). Journalists vigorously defend themselves by saying that they do not manufacture news, but only report on the facts. In a study of literacy of editors, most editors claim that they do not select news but are only responsible for responding to events that are already happening.

Interestingly enough, the findings of the study contradict editors' claims (Stokes, 2003:65). The findings confirm that editors do play a major role in the selection of news, which means that they actually do construct reality. Editors select the news by the mere fact that they (editors) decide on what to print, in what position, how much space to allocate and with what emphasis (Lippmann 2000). In another study conducted by Schlesinger (1978), it was found that the existence of editorial policies implies that news organisations construct news in identifiable ways in terms not only of selection of news but also their angling and mode of presentation. The issue of news selection is well

outlined by Staab (1990:423-443), who argues that news selection is both "a conscious and unconscious decision made by an editor when writing news". Staab (1990:428) outlines the Instrumental Actualisation Theory invented by Kepplinger, which posits that news selection does not happen because of certain factors but is an intentional act by journalists which is done for the purpose of achieving certain organisational objectives. The assumption in this theory is that "news selection is understood as a direct response to stimuli" (Staab 1990:428).

Manning (2001:50) contributes this idea and seeks to clarify the point that journalists do not fabricate stories but are required to work according to certain routines and editorial policies when compiling stories. Shoemaker and Reese (1996:105) argue that for people to understand the media content, it is important to examine the routines and organisational practices that are followed when writing the news. Routines are defined as "those patterned, routinized and repeated practices and forms media workers use when gathering the news" (Shoemaker & Reese 1996:105). Bennet (1996:129) offers the advantages and disadvantages of routines. The notable advantages of using routines are that they enable newspapers to capture the same readers, which automatically leads into their catching similar advertisers. Another advantage is that using routine reporting increases objectivity and consistency of facts across newspapers. The main disadvantage of routine reporting is that it decreases content diversity in the media, which automatically results in the principles of democracy being undermined. Using routines is considered by critics as a bad thing in the journalism profession (Shoemaker & Reese 1991:107).

It is important to outline what different authors understand as news. Studies on news production like that of Lester (1980: 984) posit that news is a product of reality-making activities and not simply reality-describing ones. Fowler (1991:4) believes that news is the representation of a world in language. Through language, individuals are given an opportunity to explore and understand their own individuality and through language individuals gain access to social relationships (Hartley 1982:14).

According to Oosthuizen (2001:453), news is primarily comprised of the events that are selected as newsworthy by journalists and which are eventually published and/or broadcast. This writer further argues that news is semiotic in nature. Fourie (2001:327) defines semiotics "as the science of signs" and codes. The main aim of media semiotics is "to sharpen our critical awareness of the ways in which the media reflects, represents and imitates reality or aspects of reality with the purpose of conveying specific meaning usually in support of an underlying ideology, point of view, ideal, argument and attitude". This, therefore, is in agreement with the theory of social construction that news is not the reality in itself but a representation and an imitation of reality.

3.1.1 News framing and values

Fowler (1991:11-13) similarly maintains that news is not something that just happens on its own but journalists more or less unconsciously make use of socially constructed categories called news values. He (1991:12-15) posits that news workers use frames to organise reality applying variables such as time, space, facticity and presentation. The concept of frame analysis was first proposed by Goffman (1974). When people interpret what is going on in their world they do so through the framework or schemata of interpretations they have which Goffman calls the primary framework. News framing is the process of filtering and transmitting news through an angle or "frame" in order to support specific ideologies, stimulate widespread attention or persuade an audience. Schudson (2003:35) points out that to acknowledge that news stories frame reality is also to acknowledge that it would be humanly impossible to avoid framing. Production of news involves routine gathering and assembling of certain constituent elements which are fashioned to construct an account of the particular news events.

The findings of this study will be largely based on the news framing literature. Another study that has used the frame analysis approach is that of Tuchman (1978) who studied news reporting. His findings indicate that news reporting consists of events framed to organise everyday reality. A similar work is provided by Entman (1993) who develops Goffman's idea and clearly conceptualises framing and how frames work. He posits that

framing involves "selection" and "salience". These, therefore, mean that framing is about deliberately selecting some pieces of information to report and obviating others.

As stated, news workers work according to frames. The first frame that news workers work according to is time. Media workers are required to meet certain deadlines which mean that every story has a certain life span before becoming obsolete. Watson (2007:108) asserts that time is a dominant frame. News workers often find themselves having to conform to deadlines and "deadlines cut things short and drop things out". Oosthuizen (2001:457) notes that time frames determine what type of news one eventually reads or sees/watches in the news media.

The second frame that guides the news media is space. As with time, media staff impose a structure upon space to enable them to accomplish their daily work and plan across days (Oosthuizen 2001:457). As it is practically impossible for news workers to be in every place, many news organisations rely heavily on the correspondents of bigger news organisations that are able to put their news reporters everywhere rather than rely on stringers. It appears, however, that most editors prefer to publish material produced by their own staff than rely solely on stringers/agencies.

Third, journalists are largely guided by the facticity frame. This frame deals primarily with the way in which journalists find facts. According to Tuchman (1978:93), it is important for journalists to ensure that they follow professional procedures to reflect and separate facts from comments. This is usually done through the use of quotation marks.

Lastly, Oosthuizen (2001:458) asserts that the final frame is actual presentation of the story. This tool is used by journalists to transform events into news by using specific codes. This frame primarily deals with the fact that for news to be news it needs to be presented in a certain format.

The issue of frames clearly outlines that news does not select itself and reinforces the view of sociologists that authors and journalists are guided by various frames and do not

present news as it is. Similarly, the study conducted by Schlesinger (1978) found that the existence of editorial policies implies that news organisations construct news in identifiable ways in terms not only of the selection of news but also in their angling and mode of presentation. Watson (2007:112) argues that news values provide/serve as rules for news selection. Schudson (1979:19) maintains that news values are "commonly assumed, unnoticed background assumptions through which news is gathered and within which it is framed".

It is also important to state that news is produced according to news values: namely, frequency, threshold, unambiguity and relevancy among others. Each value will be discussed briefly in the table below.

It is argued that "events are likely to meet the criteria of newsworthiness if they satisfy the conditions of frequency, intensity, unambiguity (clarity), cultural proximity, relevance, consonance, predictability, unpredictability, continuity, composition, elite persons and elite nations, negativity and ethnocentricity" (Kisuke 2004, Galtung and Ruge 1965:53-60). For the purpose of this study, the researcher will not cover all the news values; only the ones relevant to the study as seen in the table.

Table 3.1: Values in news

| News value | Description of the value |
|-------------------------|---|
| 1 Frequency | Frequency refers to the time span taken by |
| | an event (Hartley 1982:76). News stories |
| | differ according to their importance. Some |
| | stories take very little time while others |
| | take very much longer to unfold and to be |
| | made meaningful. The case of the Schabir |
| | Shaik trial suited the daily newspapers well |
| | because people/readers had to be updated |
| | continually on what was happening. |
| 2 Threshold | Hartley (1982:76) posits that threshold |
| | refers to the size of the event. The bigger |
| | the story, the more added drama is needed |
| | to keep it going. Crime reporting is an |
| | example of this. During the Schabir Shaik |
| | trial, many small trials unfolded. <i>Natal</i> |
| | Witness reported as follows: Shaik has |
| | following – out with his legal team (Natal |
| | Witness, 25 Feb 2005:1). |
| | I will sort you out (Natal Witness, 12 |
| | March 2005:1). |
| 3 Unambiguity (Clarity) | Unambiguity is one of the important news |
| | values. Hartley (1982:77) maintains that |
| | clarity of the story is very important and |
| | range of possible meanings must be |
| | limited. In news it is important for a story |
| | not to generate too many meanings but |
| | Hartley (1982:77) posits that "news events |
| | that accord with the cultural background of |
| | news gatherers will be seen as more |

| | meaningful than others and so more liable |
|------------------|---|
| | to be selected". |
| 4 Relevance | Relevance is another important news value. |
| | Relating to cultural proximity, it is |
| | important for news reports to cover issues |
| | of local importance. |
| | |
| 5 Consonance | Consonance means that the media expects |
| | that something will happen. Hartley |
| | (1982:77) posits that the danger of |
| | reporting like this is that in most cases |
| | important issues are ignored while special |
| | attention is given to one anticipated issue. |
| | |
| 6 Unexpectedness | The unpredictability and the rarity of |
| | events are important in news reports. |
| | According to Hartley (1982:78) unexpected |
| | news is normally unleashed via expected |
| | contexts. |
| | |
| 7 Continuity | Hartley (1982:78) writes that it is important |
| | for a story to have a long life span. The |
| | case under investigation is the classical |
| | example of a running story. |
| | |
| 8 Composition | Composition refers to different kinds of |
| | events. It is important for a newspaper to |
| | be packed in a balanced fashion (Hartley |
| | 1982:78). |
| | |
| | People with economic means are |

| 9 Reference to elite persons | considered to be of more value than |
|-------------------------------|--|
| | ordinary citizens. Stories concerning rich |
| | people are covered far more than those |
| | about ordinary people. The Schabir Shaik |
| | trial falls under this value. This case |
| | touches on high profile people in the land |
| | like Jacob Zuma. Gans (1979:26) notes that |
| | ordinary working citizens are only given |
| | media coverage when they are victims of |
| | accidents such as mine disasters and |
| | construction accidents. |
| 10 Reference to elite nations | Hartley (1982:78) posits that stories |
| | involving influential countries are more |
| | likely to be reported than events in poor |
| | countries. For example, if something |
| | happens in the USA, it is more likely to be |
| | reported than something that occurs in |
| | Zambia. |
| | |
| 11 Ethnocentrism | Ethnocentrism assumes that news reporters |
| | give more attention/coverage to news that |
| | is relevant to their country – stories about |
| | domestic conditions (Gans 1979:42). |
| 12 Altruistic democracy | This notion generally assumes that news |
| | media should pursue stories that are largely |
| | based on public interest and public service. |
| | Journalists tend to cover stories in a way |
| | that differentiates winners from losers. |
| | According to Gans (1979:43), news media |
| | tends to have little patience with losers. |
| | |

| | News media also pays attention to racial |
|---------------|--|
| | integration. It is generally assumed that by |
| | paying special attention to racial |
| | integration, the media adheres to the norms |
| | that are viewed as expressions of public |
| | interest (Gans 1979:44). This value further |
| | requires that the media constantly report on |
| | political and legal failures to achieve |
| | altruistic and official democracy. |
| 13 Negativity | Hartley (1982:79) maintains that bad news |
| | is actually good news. News that is bad is |
| | more likely to be covered than purely good |
| | news. Good news tends to be boring; hence |
| | does not get enough coverage. Hartley |
| | (1982:79) maintains that news values are |
| | able to provide a good idea of a kind of |
| | event that will survive the selection |
| | process. Values further provide the "clues |
| | as to the priority different stories will be |
| | given. |
| | |

McManus contributes to the social construction theory debate by discussing various steps of news production. According to McManus (1994:88) the first step of news production is uncovering potentially newsworthy issues. The major problem, however, is that significant news is often expensive to discover because powerful interests want it hidden.

McManus (1994:98) posits that the second stage of news production is "choosing among those events and issues uncovered in step one a subset for inclusion in the newspapers". This is a difficult process because journalists are faced with the choice of having to choose journalism needs and market demands.

The final stage of news production is the actual reporting of the story (McManus 1994:89). When reporting, a journalist is normally faced with various choices such as which angle to emphasise and the angling of news constitutes bias. McManus (1994:89) further notes that newspapers report according to their needs. If, for example, the aim of the newspaper is increased profit, then less care may be taken with information. It is generally assumed that too much information tends to be very boring.

Additionally, it is important to note that news is not selected and constructed in a vacuum. Most authors posit that with every profession, including that of journalism, there are limitations, pressure, structures and norms under which journalists are expected to work (Tuchman 1978, Gans 1978 and Oosthuizen 2001:463). Institutions and individuals under which the media works largely influence the nature of news. Gassaway (1984:77) asserts that just as in any profession, news workers are influenced by many things when performing their function – culture, family, schools, churches and others that they deal with daily.

Oosthuizen (2001b:164) believes that media institutions function within specific societies and any society provides the framework for the expression of expectation about media functioning. And most of the time a society is guided by a set of fundamental values, for example, in a democracy values like freedom, equality and order are important.

3.2 Socialisation of Media Workers

Another important aspect to consider when studying conformity in the print media output is the socialisation of media workers. Croteau and Hoynes (2000:14) define socialisation as "a process whereby we learn and internalise the values, beliefs and norms of our culture and in so doing develop a sense of self". The socialisation process influences journalists when performing their functions. Journalists behave in ways that are defined by Croteau and Hoynes as "socially acceptable ways".

As in any other career, journalists not only socialise, but also rely on one another for story ideas. In as far as the media is concerned, Croteau and Hoynes (2000:20) identify three types of social relations: namely, relationships between institutions such as interactions between the media and government; relationships within an institution as well as relationships between an institution and the public. This involves the influence of mass media on readers.

The relationship between the media and other social institutions is another relationship to consider. For people to understand how media functions, it is essential to take into consideration that media exists within a social, economic and political context which plays a big role in its operation.

Yet other relationships to understand are the relationships within the media industry, such as the relationships among journalists from different newspapers. As journalists normally function within similar standards, it is inevitable that they will copy one another's styles of reporting. Within the media industry standard journalistic practices normally shape the process of news reporting and even the content of news.

Socialisation is again, is bound to inject content conformity into the print media. An example of socialisation where reporters end up sharing information is provided by Tuchman (1978:75), where a *Times* reporter is witnessed sharing information with a *News* reporter. Tuchman (1978:78) also observed that when John F. Kennedy was shot, journalists were sharing information with everyone, including their competitors. The sad result for democracy is that journalists do not regard such practice as a problem; instead they view it as a form of cooperation that is used to help one other get a story out on time (Tuchman 1978:78).

One of the causes of content conformity is this habit of journalists relying heavily on one another for ideas, and this reliance decreases content diversity (Shoemaker & Reese 1991:101). This practice – closely related to socialisation – is pack journalism, whereby journalists covering the same beat or the same story tend to emphasise the same angle

and adopt the same viewpoint (Schudson 2003:139). Pack journalism also refers to a notion whereby journalists gather and report news in groups, promoting the sharing of ideas and news judgments with other journalists (Schudson 2000:188; Shoemaker and Reese 1991:221). Reporters are in many cases expected to cover news in "packs" so that they can obtain stories similar to those of their competitors. Failure to cover stories with a similar angle is perceived as the reporter having "missed".

Underwood (1995:146) points out that the main reason behind this trend is the pressure to sell. Reporters are in most cases under pressure from their editors to conform to other news organisations. According to Shoemaker and Reese (1996:170), news workers consider this behaviour a good way of reporting, because stories are accepted easily by editors. Reporters are, therefore, often not prepared to differ or to approach the stories from a different angle, because differing will increase the chances of the stories not being approved by their editors. One must mention that such pressure is against the norms of journalism and poses a threat to the healthy functioning of democracy.

According to McManus (1994:105) the lack of investigation units in news rooms is another reason for concern. Stories are not followed up and some media houses cannot afford to finance a unit dedicated specifically to the investigation of stories. It is important to note that investigative journalism is costly, and many media managers are not willing to spend for it. Investigative journalism also requires journalists who are willing to take risks. It is also worth noting that many journalists have neither the skills nor the training that investigative journalism requires (Steyn & De Beer 2004: 387). Limited budgets restrict journalists from performing the good reporting that democracy requires. In some cases media houses do not invest in training journalists. Large budgets are normally used for buying resources, rather than for actual reportage.

Tuchman (1978:90) adds that the lack of investigative journalism in the news media is caused by the pressure to meet deadlines. It is clear that having an investigative unit is not popular amongst newspapers. For example, *KULG* in the United States was the first television station in the nation to have a team that specifically focuses on investigations

(McManus 1994:105). Investigative journalism can assist in achieving the highest standards in journalism, and it is important to note that it is only through investigative journalism that the media ensures that its role as a watchdog is fully realised.

3.3 Reliance on Sources

Another reason for content conformity in the media output is the use of one source by both print and electronic media and heavy reliance on external news providers. McManus 2004:111) posits that as it is practically impossible for reporters to cover every story, most media houses rely largely on their reporters, correspondents and news agencies which are responsible for covering events as they happen around the world. It is assumed that facts are self-validating; people with more facts are, therefore, likely to be more accurate. Legitimate organisations are regarded as being inherently more correct because of the procedures they follow to protect the institution and people who come into contact with them. According to Vermeer (1995:9), over-reliance on these limited news sources is a constraint on the independent functioning of the media.

The researcher believes that over-reliance on the same limited news sources does not only inject content conformity into the media, but may also expose the media to manipulation by the very same sources. In contrast to the researcher's opinion is the fact that the use of news services seems to be useful to reporters. According to Shoemaker and Reese (1996:119), news services help the media organisations operate smoothly, enabling journalists to meet deadlines. Furthermore, most journalists rely on pictures, copies of speeches and other handouts that are provided by public relations practitioners and spin doctors.

Given all the above factors influencing the media content, can the media remain objective in its reporting? The following section discusses the issue of objectivity in the media.

3.4 Objectivity in the Media

Objectivity means that news comes to the consumer unattained by conscious bias or external influence that could make it appear but what it is. There are various opinions relating to the existence of objectivity when producing news. Croteau and Hoynes (1997:106) state that objectivity is more than simply getting the facts straight. It is "a doctrine that perceives the separation of facts and values as a messy business that requires the use of practices to ensure their separation". Croteau and Hoynes (1997:108) posit that objectivity is "an ideal that is unattainable".

It is important to note that not all journalists subscribe to the notion of objectivity. For example, David Brinkley was one of the several panellists on the ABC-TV viewpoint programme considering the role of the media covering US presidential election campaigns. Responding to the question from the audience on whether journalists can remain objective or not, David Brinkley responded as follows: "Well, first of all, I would question the use of the word objectivity. In my opinion, there is no such thing. Possibly a computer is objective. A human being who is objective would have no standards, no values. He would be in favour of nothing. And such a person, if there is one, probably should be locked up" (Koppel 1983:10).

Legun (1971:206) strongly asserts that "there is no such thing as objective news" but admits that there are things such as neutral facts. These, according to him, are mostly statements about simple events. He continues (1971:206) that being "neutral can be done up to a certain level". As soon as the reporter moves into writing a story with different angles, however, it is inevitable that he/she will be required to "select" one angle. Selectivity according to him is more to do with the process of the reporter choosing what to take and what to leave out. This process results in the news losing the claim of being absolutely objective because it involves value judgments by the reporter.

Although Legun (1971:206-207) admits that it is difficult to attain objective reporting, he strongly suggests that "balanced reporting should be the aim of every serious journalist ... a committed journalist should attempt to describe as fully as possible the total situation". Manning (2001:68) supports Legun when he argues that objectivity cannot be achieved by producing value-free and comprehensive accounts of real events. The term rather describes a set of practices that journalists can defend as objective. McNair (1998:70) concurs with this viewpoint as he writes that news, like all human observations, is not truly objective. He argues that achieving objectivity is just an ideal for journalists.

A similar argument is presented by Jim Kuypers, in his book *Press Bias and Politics*, where he compares the text of six speeches made by political figures about race and homosexuality with the ways that the news media covered the events. He argues that the comparisons show that "only a narrow brand of liberal thought is supported by the press; all other positions are denigrated" (Kuypers 2002:19).

Hallin (1995), Lippmann (2000) and Lichtenberg (2000) discuss the issue of objectivity in the news at greater length. Hallin (1995) and Lippmann (2000) state categorically that there is no such thing as objective standards of journalism. Everything that the public reads is the result of various selections and choices that are made by journalists. Continuing the argument, the two critics state that objective standards in journalism can never be achieved, because editors are faced on a daily basis with the task of deciding what should be printed, in what position, in how much space and with what emphasis.

Judith Lichtenberg (2000:238) defends objectivity and looks at it from various perspectives. She attacks critics such as Lippmann (2000) and Hallin (1995) who firmly believe that there is no such thing as objective standards in journalism. Lichtenberg (2000:239) refers to these claims as "not making sense" and adds that the assertion that objectivity is not desirable makes the complaint that journalism is not objective senseless. According to Lichtenberg (2000:239), if objectivity were impossible to achieve in

41

journalism, critics would not complain that some piece of journalism was not objective; to doubt objectivity is more like saying it is not possible to know reality.

Critics such as Schudson (2003:34) hold a different view: there *is* an objective way of representing each event and believe that failure to be objective can only mean bias. Schudson believes that journalists are in a better position to know the reality but they deliberately make efforts not to present reality as it is but colour it in the way that will serve their particular interests.

From the above different viewpoints it is very difficult to tell if the media can truly remain objective in its reporting.

It is important to note that my having discussed only the print media does not mean that television reports objectively. Electronic broadcasting especially television is considered to be a very biased medium when compared to newspapers because on television the audience does not have the same opportunity of thoroughly looking/reading the story. Moreover, the structure of television does not allow for balance and in-depth coverage of facts. Time is very limited. Reporters are prevented from doing the kind of job that a person in print could do.

3.4.1 The issue of balance

The issue of objectivity links well with the concern about the lack of balance in the news media. McManus (1994:146) defines a balanced story as "a story where all major sides of the story are given appropriate access to the audience". Kariithi (2005:154-56) attributes the lack-of-balance problem to the fact that daily newspapers tend to be more reactive than proactive when reporting stories. He defines proactive journalism as that kind of journalism that defines news content and news agenda and reactive journalism as "often restricted and uncritical", explaining that reactive reporting is widely seen in weekly publications and not much in weekend newspapers. Kariithi (2005:154) ascribes this to the fact that weekend newspapers have more time to interact with particular stories. He

(2005:154) criticises this type of reporting as bad reporting because it does not provide in-depth coverage of stories.

Kobre (1981:3) supports Kariithi when he posits that in-depth and balanced reporting is essential in a democracy. For a person to be a good citizen, it is important to be well informed. It has been noted that argument is based on the fact that citizens are required to vote intelligently. Garnham (1986:248) points out that the answer to this is to ensure that there are trained journalists and self-sustaining newspapers. Garnham (1986:248) claims that high quality information is expensive to find and in most cases powerful elites want it hidden. Garnham (1986:248) further find it strange that the media, under the banner of balance and objectivity, claims to serve the public but is not accountable to the populace.

3.5 Gatekeeping Theory

The objectivity argument links up well with Shoemaker's gatekeeping model. The gatekeeping model was invented by Kurt Lewin and studied in 1950 by David Manning. It emphasises that for news to be news there need to be people who are responsible for selecting it. These people are called gatekeepers (Shoemaker, 1997:57). This theory argues that the selection /of stories depends largely on editors, meaning that editors and journalists select stories according to what they prefer. In the study conducted by David Manning, results indicate that editors are responsible for making the final decisions as far as news selection is concerned (Schudson 1997:9; Shoemaker 1997:57). These findings confirm the notion that news is far from mirroring reality and is more of a manufacturing and fabrication process.

This theory is not without its critics. The gatekeeping approach to news fails to acknowledge that news gathering is a process that includes many facets such as sources and news routines. The theory also fails to consider that editors and advertisers are not the only ones responsible for determining what goes into the paper. It does not explore the impact of external factors such as the influence of advertisers. The gatekeeping model merely portrays the news media as the only active participants in the news production process (Shoemaker 1997:57).

Closely related to the gatekeeping theory is the revised work of Schudson (2000:175-195) in which he looks at three perspectives of news production. Among these he employs the view of the political economy of the media, which posits that the capitalist class places pressure on editors and reporters to report their desires. The political economy approach emphasises that the outcome of the news process is shaped by the structure of the state and the economic foundation of that particular news organisation (Schudson 2000:178).

Furthermore, the political economy of the media approach has shown that professional journalistic ideology tends to be undermined because newspapers are both directly and indirectly obliged to reflect not only the interests of the media owners and advertisers, but the socially powerful forces (Golding & Murdock 2000:73; Schudson 2000:179; Williams 2003:56).

3.6 Media Ownership

In the light of the above dynamics, the researcher attempted to unbundle the question of the impact of ownership on the news content. There seems to be a direct link between ownership and news content. The underlying assumption is that the media is an instrument that reflects the position of those who control it. It is therefore important for the researcher to look at the issues of ownership in more detail.

The position of many authors is that newspaper owners and advertisers put pressure on editors and reporters as to which stories are "suitable" for the newspaper (Schudson 1991:143; Williams 2003:74-86; McQuail 2005:99-100; Croteau & Hoynes 2003:58). Failure to adhere to the needs of a newspaper owner often results in serious action being taken against a particular editor or reporter.

Doyle (2002:173) concurs that media ownership plays an important role in the functioning of the media and ultimately determines news output. **A** study conducted in 2001 of 97 countries, commissioned by the World Bank and conducted by Djankov, McLeish, Nenova and Shleifer, states:

In our sample of 97 countries, only four per cent have other ownership structures – a mere two per cent of our sample – and family-controlled newspapers account for 57 per cent of our sample, and families control 34 per cent of television stations. State ownership is vast. On average, the state controls approximately 29 per cent of newspapers and 60 per cent of television stations. The state owns a huge share – 72 per cent of radio stations. The media is therefore owned overwhelmingly by parties most likely to extract private benefits of control. (Djankov, McLeish, Nenova & Shleifer 2001).

The results of both studies above support the notion that ownership is still a problem in most countries and is concentrated in the hands of the few.

McNair (1998:101) gives a detailed analysis of how economic forces impact on journalism. He argues as follows:

Journalism is in a real sense the private property of these individuals and conglomerates and they are free within the law to dispose of and use it as they like. Journalists are employees, strongly influenced by those who own or control their organisations. They are subject to direct economic power in the same way as other employees of capital enterprises, although some journalists with "star" value bring their own reputations to the market and use them to buy autonomy.

McNair further contends that journalism is an industry, meaning that it has a right to do whatever is necessary in order to increase its chances of survival in the crowded market.

The foregoing discussion highlights conformity in media content at the global context. While still systematically untested, there is reason to believe that the South African media also suffers from these tendencies. There are numerous anecdotal reasons to subscribe to this point of view. First, the South African media is highly globalised, by ownership, media training, exchange programmes, and socialisation. Berger (1993, 2005) highlights

ownership structures in the media, while Strelitz and Steenveld (2005) offer some discussion of the globalisation character of the local mass media. There is ample evidence of globalisation patterns in formal training and professional exchange programmes: all major journalism schools have formal links with partner programmes in North America and Europe, and many South African editors are alumni of major global exchange programmes like Harvard University's Nieman Fellowship. This anecdotal evidence, if demonstrated through systematic research, would most likely bring to our attention the urgent need to understand and deal with conformity in the local media. A failure to test these persuasive patterns might not only mean that South African audiences are inadequately served by the mass media, but that significantly high levels of conformity could even jeopardise the country's young democracy. These observations undergird the purpose of this study.

Croteau and Hoynes (2003:51) argue that the media, which is owned by the few, often has content that lacks diversity. Croteau and Hoynes (2003:52) believe that this decline in diversity in newspapers poses a threat to the ideal free press. In his book, *Media Monopoly*, Ben Bagdikian (2004:27-54) asserts that media ownership is becoming concentrated in the hands of conglomerates. Illustrating this point, Bagdikian (2004:27-54) points to the "Big Five" in USA as good examples of conglomeration, positing that the Big Five are simultaneously engaged in both print, electronic media, movie production and even music industries. Machesney (1998: 1-25) concurs with Bagdikian when he claims that the media no longer focuses on one thing but consists of big business with various interests. These mergers are bad for democracy, however, because they discourage competition among newspapers; sometimes one conglomerate owns up to six newspapers.

The results of a content analysis study of newspapers conducted by Entman (1993) reveals that competition can instead encourage newspapers to produce news with similar content in order to attract the same readers and catch the same advertisers. Competition

breeds sensationalism⁸. In most cases journalists are expected to cover similar stories, and it is difficult to make a profit under such circumstances. Reporters end up sensationalising stories with the hope of increasing readership figures and selling more newspapers than their competitors.

3.7 Influence of Advertisers on Content

Linked with the issue of ownership influence is the influence of advertisers. It is generally assumed that some advertisers insist on a particular type of news presentation and threaten to withdraw their advertising should the newspapers fail to comply (Curran 1991:87). Schudson (2003:125) conducted a study with 118 television directors and the results indicate that advertisers do indeed pressure editors either to run positive stories or kill the negative ones.

Another example which supports Schudson's findings is the case of a tobacco company in the United States which immediately withdrew its advertisement in the *Mother Jones Magazine* after that magazine ran a negative story insisting that tobacco is the major cause of lung cancer (Parenti 1993:32). These results indicate that advertisers do indeed indirectly control the media. In his book (2004), Bagdikian (1983:61) argues that news organs are often beholden to advertisers, who may be potential subjects of critical stories. He adds that in spite of that, journalists and editors still have a duty to produce news fearlessly and without any coercion from outside forces from business or from their own media sources.

Given the above evidence it is, therefore, safe to say that over-reliance on advertising is bad for media output because it is difficult to imagine these newspapers standing on their own by publishing news articles that may be of public interest but have the potential to antagonize an advertiser and the publication's potential future. Schudson (1986:114) posits that advertising is not good for a healthy democracy. Herman and Chomsky (1994: x) concur that over-reliance on advertising as the primary source of income is a problem

⁸ The term "sensationalisation" is used to refer to the form of reporting that is not objective (McQuail 2005:357). Sensationalisation is used by newspapers to sell more newspapers. Sensationalised news appeals more to a wider audience; on the other hand, normal news is seen as boring to the readers.

because it inevitable directly affects media content. They argue that advertisers exert two kinds of pressure: namely, selecting and packaging of news.

Murdock (2004:247) identifies three more problems associated with over-reliance on advertising. He argues that dependence on advertising revenue subjects newspapers to editorial pressures from large advertisers wishing to promote their corporate interests. Second, dependence on advertisers limits space for more voices. Third, advertisements lead newspapers to sensationalism. Given this argument, it is equally important to note as argued by Bagdikian (2004:247) that 80% of the profit for most newspapers comes from advertisers. Murdock (1982:145) also admits that the costs of producing a newspaper are covered largely by advertisements and not by the public. It is for that reason that editors, in most cases, find themselves not knowing whether to adhere to journalism norms or create an environment that is good for selling. Any newspaper will think twice about publishing a story that might upset its biggest advertiser (Herman and Chomsky 1994: x; Jackson 1993:76).

Murdock (1982:145), however, reasons that although the press obtains most of its income/profit from advertisements and not from the public, this does not mean that advertisers are more important than the public. Golding and Murdock (2000:75) contribute to this debate by contending that the public are as important as advertisers in terms of bringing in profit for the newspaper, because the price that advertisers pay for advertising depends entirely on the size and the composition of the readers.

The viewpoint of Sparks (2000:268) is different as he argues that newspapers do need advertisers to make a profit. He states that generating profit is important for a newspaper so that it can carry out its daily duties but the only problem is that sometimes editors are faced with indirect pressure from advertisers to act according to what the advertisers want. He further notes that without profit there can be no fourth estate, no investigative journalism and even no newspaper at all. Despite his viewpoint, however, Sparks (2000:268) admits that profit-generating newspapers pose a threat to democracy. Negrine (1989:88) posits that without readers, no newspapers can survive; without sufficient

readers but with willing advertisers, the chances of survival are greater but the medium still has to prove itself. Unless it does so, it is likely to lose both readers and advertisers.

3.8 Commercialisation

Over-reliance on advertisers leads to commercialisation which is discussed in more detail in this section. Ross (1997:4) defines commercialisation of the press as the drifting of ultimate control into the hands of men with business motives. It is apparent that mass media has become too commercialised and this commercialisation of news undermines responsible reporting (Croteau & Hoynes 2003:58 & Hallin 2000:259). The underlying assumption of the concept of commercialisation is that media normally produces whatever is sold the most and that the content of news is packaged in a way that encourages readers to buy.

This is what Bagdikian (2004:242) calls "buying mood". The search for profit compromises quality and according to McKee (2005:68) quality is compromised mainly in that a newspaper avoids complex language and "climbs down". Furthermore, commercialised news is bad for democracy because it suppresses diversity of viewpoints (McManus 1994:34). Keane (1991:53) holds a similar view when he writes that the media should not only be aimed at making a profit but should operate for the benefit of all citizens; failure to cater for all members of the public poses a threat to democracy.

From a public sphere perspective, the concern is whether commercialisation compromises public service obligations. It is clear that commercial pressure in the print media sector has resulted in many problems such as "tabloidisation", reduction of staff, and a "juniorisation" of newsrooms (Wasserman and De Beer 2005:39).

3.8.1 Market driven journalism

Since the early eighties there has been a worldwide tendency to follow a marketing approach in the newspaper business and this tendency is generally referred to as market-driven journalism. McManus (1994:194) posits that commercialisation analyses news as a commodity in which normative journalism ideals take a back seat in subordination to

market driven journalism. In essence, this notion means that newspapers are doing what they must to respond to the needs of the market, so that profit can be generated (Hallin 2000:221; Underwood 1995:19). The concept of market driven journalism assumes that media owners put pressure on journalists to think and act like marketing managers. Some newspapers even go to the extent of employing marketing managers from retail industries for the purpose of introducing a business for their newspapers firms. An example in the United States of America is Mark Wile, who became chief at Times Mirror Co. in 1995 after a career with food industry giant, General Mills, and Jim Rosse, an economist who was selected to head Freedom Communications in 1992. This was done to put the newspaper in a better position to achieve its commercial objectives.

Gillmor (2004:24) posits that market driven journalism delineates the limits and the extent to which the media can be open to dissident voices and encourage genuine debate. This view is supported by Bagdikian (2004:247) when he argues that market-driven journalism gathers an audience "not to inform it, but to sell it to advertisers". This is not good for democracy in that the public loses an opportunity to engage in constructive debate and readers consequently become what Gillmor (2004:24) calls "a mass of shallow citizenry".

In a market driven journalism environment, public interest is subordinated to commercially favourable content. Bagdikian (2004:247) notes that the sole purpose of market driven journalism is to gather an audience not for the purpose of informing them but selling and making a profit. McManus (1994:191) further argues that market driven journalism is not good for democracy because it causes society to suffer from "information poverty" and he refers to "junk journalism". This viewpoint is reinforced by Shoemaker and Reese (1991:162) who posit that "Today the editor's job is more function of management than editing". Most experienced journalists are unhappy with this form of journalism.

According to McManus, the author of the market driven journalism theory (1994:21), most news is produced by profit-making business in America. The pressure to generate

profit conflicts with journalism norms (McManus 2004:34). Journalists' standards call for independent, fair and complete coverage of events, diversity and representative opinions (McManus 1994:35). In the study conducted by McManus to assess the quality of news gathering, one news director indicated that it is very frustrating to balance the demands of the market and those of journalism; the results indicate that putting either one first, compromises the other (McManus 1994:35).

Market driven journalism encourages competition in the media industry which automatically results in journalists monitoring one another's coverage (McManus 1994:70). This is contrary to the standards of good journalism which require that the media must work independently of one another. Underwood's study (1995:118) to determine the impact of market driven journalism in news organisations indicates that most journalists are unhappy with this form of journalism. In addition, results show that market driven journalism has resulted in media outlets changing policy from public-oriented journalism to profit-oriented journalism. It is further noted in Underwood (1995:19) that most editors label market driven journalism as an "unhealthy trend" that creates friction in the newsrooms. McManus (1994:197) claims that market driven journalism poses a threat to democracy. He elaborates (1995: xx) that newspapers engaging in profit-driven journalism are in a danger of losing the true spirit of the

⁹ Quality is a concept that can mean different things to different people. Picard (2004:61) writes that in business terms "Quality involves providing value for money". In journalism quality is not about achieving profit but "it is a central element in achieving the social, political and cultural goals asserted for journalism and democratic societies". Continuing with the definition, Picard (2004:61) posits that "Newspaper quality invoices the content and methods of journalism and information. It is providing comprehensive coverage and information that produces understanding; it is evidenced in the expertise of personnel, in reportage rather than in having personnel at places where news is made; in articles that provide context and background and in articles based on thoughtful analysis. Quality also involves the operational activities associated with the newspaper, such as consistent availability of the newspaper at newsstands and few delivery errors; the technical effectiveness of the organisation; printing excellence; ease of the use of the paper; accuracy in billing and quick problem resolution. The above stance captures most of the aspects that

define a quality newspaper. The question is to what extent newspapers are adhering to the provided

. 51

description of quality.

journalistic mission – the commitment to provide information. McManus notes that market driven journalism and journalism norms cannot co-exist as these two concepts are "mutually exclusive".

The issue of market driven journalism then poses a question of how newspapers can serve the interests of the public in the light of the market-journalism concept. McManus (1994:1) states that most media cooperations (as discussed in chapter 4) are businesses that are listed on the stock exchanges. He further explains that it is normal that a shareholder's or investor's primary interest will be to generate profit. It, therefore, goes without saying that the media's objective of pursuing truth will be a secondary objective.

Doyle (2002:13) takes a different view when he argues that commercialisation should not be seen as a threat to good journalism but needs to be viewed as "a breath of fresh air in the news market". McManus (1994:2) supports this view when he contends that market forces can be a mixed blessing which possess the potential to invigorate journalism that was previously "too serious, sanctimonious and often plain boring".

3.8.2 Tabloid journalism

Tabloid journalism may have many faults as pointed out by critics, but Ornebring and Jonsson (2008:33) posit that it can be seen as an alternative arena for public discourse. It has an ability to broaden the reading public, giving news access to groups that previously have not been targeted by the prestige press. Tabloid journalism is also said to give rise to new forms of journalistic discourse that may be more accessible than traditional authority. Ornebring and Jonsson (2004:33) also note that tabloid journalism can help in stimulating political participation by speaking to the senses and feelings as well as the rational mind.

Ornebring and Jonson (2004:285) observe that there is actually a need for an alternative public sphere because most newspapers have been largely aimed at covering stories that cater for elite groups. According to these authors, tabloidisation has the ability to broaden the public, giving news access to groups that have not been previously targeted by the prestige press. Sparks (1991:63) asserts that the British press caters for different kinds of publics and it is, therefore, not surprising that the content of newspapers directed to the working class should be different from that directed to the ruling class.

Picard (2004: 60) notes that over the past 30 years the number of newspapers has exceeded 60 per cent: "The majority of editorial material does not classify as news. In most newspapers, three-fourths or more of the material is made up of reviews, comics, advice, puzzles and automatically entertainment, lifestyle and other sections that are not truly news". Notably, most newspapers are keen to cover non-local content rather than local content. Picard (2004:61) posits that the reason for this practice is that non-local content is easy to obtain because most of it comes from news services, whereas local content requires editorial staffing that is far more expensive. This, therefore, means that local and equality stories are often ignored in favour of stories that are more appealing to advertisers and entertaining to a larger number of readers.

Roberts (2001:15) concurring with Picard, points out that the end product of market driven journalism is "thinner and blander news & reports". Newspapers are becoming less distinctive institutions and journalistic and community achievements remain secondary. This point is well illustrated in Underwood and Stamm's survey (1992:307) which shows that increased reader and market orientation of newspapers manifests itself in loss of depth in stories and displacement of informative news with "softer" reporting.

¹⁰ "Tabloid journalism" is the term used to define the tendency of the media to report stories that interest the public instead of writing quality news. Ornebring and Jonson (2004:23) posit that "tabloid journals lower the standards of public discourse or even worse, tabloid journalism may even actually be a threat to democracy breeding cynicism and a lack of interest".

In this study it is argued that there is a shift of interest. Newspapers are more interested in pleasing advertising interests than informing readers. Hadland (2005:13) believes that both market driven journalism and commercialisation pose a threat to editorial independence because both lead to suppression of news that may upset advertisers. With similar view is Bagdikian (2000:XV) when he observes that serious journalism has been neglected in favour of trivial and banal content and, consequently, there is superficialisation of public discourse that results in degradation of news.

Picard (2004:54-64) explores the diminution of the importance of public service through commercialism. He posits that newspaper content has diverted its attention from journalism to activities primarily related to business interests of the press. He further states that "there is a growing conflict between the role of newspapers as the servants of readers and the exploitation of readers to seek additional gain. It is sad for democracy – as pointed out by Picard (2004:55) – that many publishers have adopted strategies that make commercial considerations equal to or in some cases more important than editorial quality. Such practice is harmful to the social service role of newspapers.

Picard (2004:55) further comments that it is not fair to blame newspapers for attending to commercial aspects. He acknowledges the danger of focusing on commercialisation but notes that "newspaper companies, like all firms, have economic self interests and can be expected to attend to them". He, however, warns that too much emphasis on economic interests by some newspapers poses a reason for concern.

Based on the above arguments it is, therefore, clear that both commercialisation and market driven journalism undermine the role of the press in the media and have worrying implications for democracy. Both commercialisation and market driven journalism are, therefore, a "diminution of quality and diversity of quality and diversity as commercial media emphasis on serving broad audiences and providing content that will not lead to reductions in audiences and advertisers".

3.9 Conclusion

The chapter highlighted the fact that news production is a process that involves many facets that ultimately affect the output. The chapter asserted that news is largely shaped by various factors. It also argued that newspapers tend to rely too much on advertisers and that this is not good for democracy as advertisers tend to be more concerned about profit than the production of quality news. In addition, it showed that news values are criteria used in newsroom practice to select items that qualify for inclusion in the news from the several contending items that are potentially news.

The following chapter considers the South African press as the context of the study.

55

CHAPTER 4: THE SOUTH AFRICAN PRESS IN CONTEXT

4.0 INTRODUCTION

The context of the study emphasises the importance of diversity of opinions in the media. Not only does there need to be a diverse and generally representative range of views and interests represented within the media, but all South Africans need to have avenues through which to express their views and ideas in the media (Kisuke 2004). An array of electronic and printed media in South Africa provides an opportunity to express views. The question here is to what extent the media fulfils this requirement.

This chapter posits that historical discussion of the press is important because it provides a broader political and economic context in which the South African media operate. The history of the South African media is important in this study because as Wigston (2001:4) points out, the present media environment is largely influenced by what happened in the past. Wigston (2001:5) admits that studying media history is never easy because there is still "no usable model to enable us to undertake an all-encompassing study of the media environment".

4.1 History of the South African Press

Historically, the *Cape Town Gazette*, an official government publication was first published on 16th August 1800, by Alexander Walker and John Robertson (Wigston 2001:35; Kareithi 2005:2 & Jackson 1993:17). The paper appeared mainly in English and was largely used for official notices and advertisements which were controlled by the governor. This resulted in the newspaper carrying only the interests of the colonial rulers. Wigston (2001:35) notes that the title of the newspaper was changed to *Kaapche Courant*, but reverted to its original bilingual title in 1806 when the British re-occupied the Cape. According to Ocitti (1999:13), the paper's appearance in Cape Town was to prove only a journalist teaser. It is, however, noted that the lifespan of this publication

was very short. Within just three months of its existence it had to close down and "there is no record in the literature as to why Africa's first newspaper suffered such a sudden death". Ocitti (1999:13) believes that one of the possible reasons for its early demise was that it was run by people without any journalistic background: "... these were probably amateurs who had no experience whatsoever in the successful press".

Expanding on this view, Ocitti (1999:13) posits that the proprietors of the newspaper might have found out that publication of a newspaper requires much more than they anticipated. It is suspected that problems such as profitability and the quality of the newspaper might have led to its sudden death. Kareithi (2005:2) in support of Ocitti's work, states that the main reasons for the closure of this newspaper were the lack of economic or technical resources and government persecution of the newspaper through severe forms of censorship. It is equally interesting to note that the publication of this newspaper was nearly 150 years after Dutch settlers first arrived at the Cape of Good Hope in South Africa. From this, one can conclude that the press was not a priority of white settlers in South Africa as they survived for 150 years without a newspaper.

Two months after its closure, Thomas Pringle and John Fairbairn started the bilingual periodical known as *The South African Journal (Het Nederduitsch Zuid-Afrikaansch Tydschrift)* together with Reverend Abraham Faure (Wigston 2001:35). It is important to point out that this journal became unpopular because the content was largely about constitutional matters, such as curbing despotic power, and in later issues it contained issues on the freedom of the press. The journal was eventually closed down in 1824 (Wigston 2001:36). The first non-government newspaper was *The South African Commercial Advertiser*, which first appeared on 7th January 1824 in Cape Town (Wigston 2001:35). Reporting on Africa, Legun (1971:29) notes that major newspapers were wholly or largely in the hands of the powerful and were primarily responsible for defending their interests.

With the coming of democracy in 1994, South Africa's newspapers were freed from all restrictions. The new constitution safeguards freedom of the media; freedom to receive or impart information or ideas; freedom of artistic creativity; academic freedom and freedom of scientific research (Stokke 2000).

The study is not based on broadcasting, but it is, however, important briefly to discuss the history of broadcasting. Teer Tomaselli (2005:203) tells us that broadcasting was largely dominated by the South African Broadcasting Corporation (SABC) which was formed on 1st August 1936.

SABC radio is dominant in South African broadcasting. AMPS indicates that over 78% of South Africa's adult radio listeners tune in to SABC radio. The on-line news service, SABCnews.com attracts an average of 600 000 site visits a month. Radio is mostly the domain of the state broadcaster, the SABC, although there are also growing numbers of community stations. There are currently three main players in television: the SABC, which has three channels (SABC1, SABC2 and SABC3); the free to air broadcaster, etv and Multichoice, which has sewn up the pay channel market. The SABC was incorporated into the network of cultural, economic and political institutions through which the Afrikaner nationalism of the National Party secured its hegemony from the 1950s onwards.

According to Barnett (1999: 4), things have not been easy for the SABC because it has operated as a commercial broadcaster since the 1950s. Given its relation to the institutional frameworks with Afrikaner nationalism on one hand and its financial dependence on advertising revenues on the other hand, the SABC stood in a highly compromised relationship with respect to the imperatives of both the state and the market. As the result of all that, the SABC is guilty of having fallen short of the principles of public service broadcasting as they have been understood over the past half century (Teer Tomaselli 1996).

The SABC played an important role in both constructing and supporting apartheid structures of pre-1991 South Africa (Tomaselli and Tomaselli 2001:124). Some authors posit that during the 1980s the SABC was used to support the interests of the ruling party and it is for this reason that it was eventually perceived as the government mouthpiece (Tomaselli and Tomaselli 2001:124; Wigston (2001:3). Tomaselli and Tomaselli 2001 (124-125) note that during apartheid there were two broad "camps"; namely, the ANC camp and the National Party camp. The National Party camp was made up of parastatal organisations such as the SABC, the Electricity Supply Commission (ESKOM), South African Posts and Telecommunications (SAPT/Telkom), Human Sciences Research Council and many others. The ANC camp on the other hand consisted of Film and Allied Workers' Organisation (FAWO) and protest press such as *Weekly Mail, Rand Daily Mail, Vrye Weekblad, New Nation* and *South*.

For over 40 years the apartheid state tried to gag the country's newspapers, using legislation, harassment and imprisonment, culminating in the late-1980s States of Emergency. Through all of this, South Africa's press continued to report on all the news they could (Stokke 2000). In April 1994, nationwide non-racial elections marked the formal end of apartheid in South Africa, leading to many changes such as the changes in the way media operated specifically in programming, diversification of ownership as well as expansion of services.

Currently, South Africa has a listing of 20 daily and 13 weekly newspapers, most in English. Some 14.5 million South Africans buy the urban dailies, while community newspapers have a circulation of 5.5 million. There is also a range of general and specialised news websites which, in terms of the speed and breadth of their coverage, are on a par with the best in the world (ABC 2008/9).

Jackson (1993:34) posits that in addition to the daily and weekend newspapers, the country has a number of regional, provincial and suburban press. Most of these newspapers are produced in suburbs of large metropolitan n areas or in small towns around the country. Other media include extensive consumer magazines which do not

produce political news but are specifically aimed at the consumers. Jackson (1993:34) contends that hundreds of other magazines and journals cater for professional, trade and special interest groups. In these publications, political issues are largely ignored. In the early years South African media were clustered into categories: namely, English Press, the Afrikaans Press, the Black Press and Protest Press. These categories are also important when outlining the history of the South African press. The following section discusses each category in more detail.

4.1. 1. The English press

The target audience of English Press is largely rooted in the mining industry. The English press was formed by mining tycoons through Johannesburg Consolidated Investments, subsidiary of mining giant Anglo American. The English press was seen as a weapon for fighting apartheid (Tomaselli 2001:134). Jackson (1993:39) posits that the English press is one of the newspaper groups that are generally perceived as having some form of credibility. The Independent Newspapers group that falls under the English Press and is South Africa's largest newspaper group, traces its history back to 1889 when Francis Dormer established the Argus Printing Company, which had close links with mining magnate Cecil Rhodes.

4.1.2 Afrikaans press

According to Wigston (2001:37), "The Dutch/Afrikaans press was established largely as a reaction to the liberal views of Fairbairn in the *South African Commercial Advertiser*, particularly regarding issues such as slavery, the tensions between the Dutch farmers and the Xhosa and the work of the missionaries". The newspapers survived for 22 years before closing down owing to financial complications. The *South African Commercial Advertiser* was shortly followed by a publication called *De Zuid-Afrikaan*, the first to represent the interest of the Dutch/Afrikaners. This publication was established on 9th April 1830 by Christoffel Joseph Brand, an advocate unpopular with local authorities for his ability to defend cases against government in law courts with some success. In summary, this publication was all about championing the rights of the Afrikaner.

The Afrikaans press, mainly aimed at Afrikaans-speaking people, originated because there were people who were not happy with the English press and felt that they needed something to meet their needs. The Afrikaans press was seen as a cultural and political weapon for the promotion of the Afrikaans language (Tomaselli 2001: 134). As it was upfront about protecting Afrikaans cultural interests, it contributed immensely to the apartheid era in South Africa and generally enjoyed good relations with the government of the time. Jackson (1993:33) observes that the "Afrikaans papers have never pretended to be anything but a party press to promote the cultural and political interest of Afrikanerdom". A former editor of *Die Burger*, once described the relationship between his paper and the National Party government as "a sort of marriage in which partners never really think in terms of divorce, but do think, sometimes, in terms of murder" (Jackson 1993:33). Literature shows that until relatively recently, the government and the Afrikaans press traditionally regarded the English papers with a mix of wariness, anger or outright hostility.

The first newspaper in *Natal* was *De Natalier*, a four-page weekly, mostly in Dutch with some smatterings of English which first appeared on 15th March 1844 (Wigston 2001:39). This newspaper was edited by Boniface, previously with *De Zuid-Afrikaan*, who left the Cape hastily when faced with a charge of libel.

The first newspaper to appear in *Gauteng*, formerly known as the Transvaal, was the *Government Gazette* which was published by Cornelius Moll in Potchefstroom, beginning on the 25th September 1857 (Wigston 2001:42; Muller 1987:120). This publication was renamed many times. First, within just two years of its existence, it was changed to the *Gouvernements Courant der Zar*. Soon after this publication many other publications came into existence and collapsed. Some of the newspapers of both languages worth mentioning include *Emigrant*, which was established on 15th October 1859, De *Oude Emigrant*, *Staats Courant*, *Transvaal Argus*, *Die Volkstem*, *Mining Argus*, *The Standard and Transvaal Mining Chronicle*. Note, however, that *The Standard* and *Transvaal Mining* amalgamated with the *Diggers News* to form the *Transvaal Observer*. Soon after this string of newspapers a newspaper called *Eastern Star* was formed and was

later known as *The Star* on 3rd April 1889. This newspaper was the largest selling daily newspaper in the late 1800s and was overtaken by *Sowetan* which is in turn currently being overtaken by the *Daily Sun*. According to AMPS *Daily Sun* is the biggest selling daily newspaper with 507,328 readers (Audit Bureau of Circulation -January 2009 - March 2009)

Some of the Afrikaans newspapers that existed include *Het Volksblad* which was the first to appear in March, 1915 having evolved from the Dutch weekly *Het Western* which began publication in 1904 (Wigston 2001:42; Muller 1987:120). This publication was later followed *De Burger* which first appeared in July 1915 in Cape Town. *De Burger* was later renamed as Die Burger. This newspaper was considered as "the official mouthpiece of the National Party" (Oosthuizen 2001:142).

There were four anti-apartheid Afrikaans newspapers: namely, *Die Zuid-Afrikaan*, *Vrye Weekblad*, *Saamstaan* and *Namaqua Nuus*. All the mentioned newspapers participated in the struggle for a democratic, non-racial South Africa. Under the leadership of Max du Preez, *Vrye Weekblad was* launched on 4th November 1988 and is one of the newspapers that were committed to objective journalism. *Vrye Weekblad* enabled people to read about Nelson Mandela without his being branded a violent terrorist (Wigston 2001:42; Muller 1987:120).

4. 1. 3 The Black press

The history of the South African press will not be complete without a brief analysis of the black press. The literature attributes its inception to missionaries. The first black newspaper *Umshumayeli Wendaba* which literally means "publisher of news" was printed at the Wesleyan Mission Society in Grahamstown from 1837 to 1841. Soon after that, another newspaper followed known as *Imvo Zabantsundu* which means *The African Opinion*. This newspaper – the first written for, owned and controlled by black people – was started by John Jabavu in 1884 in King William's Town. Another newspaper that is worth mentioning, *Indian Opinion*, was formed in 1903 by Mahatma Ghandi, the founder of the Indian National Congress. This newspaper was written by an Indian, for Indian

people who are also considered to be black. Switzer (2000:44) posits that the *Sowetan* grew out of Bantu World which was founded in 1932 with the ANC leader as the editor. It was original intended for Africans residing in Soweto and other urban townships in South Africa's industrial heartland. It still exists.

4. 1. 4 The Protest press/Alternative press

Any historical study of the South African print media is not complete without mention of the protest press, also known as the resistance press. Keyan Tomaselli (2000:378) defines the term alternative press as a serial publication supportive of a broader democratic tendency of the period, organised mainly via the United Democratic Front (UDF) and the Mass Democratic Movement (MDM). A number of newspapers evolved during the apartheid years, some lasting for only a few issues, while others are still in publication. Some of those newspapers include *Weekly Mail* – founded after the liberal *Rand Daily Mail* was closed down – *Vrye Weekblad and New Nation* and *South*. These newspapers had a wider national readership and were able to secure advertising.

The anti-press was also made up of smaller newspapers, produced by organisations and educational institutions. One of these was *Sash*, originally known as the *Black Sash*, produced by the Black Sash organisation from 1956-1994. The bad part, however, is that these newspapers were always under the threat of being closed down by the government of that time. Most of those newspapers were banned, one example being *Grassroots*, *Grassroots* was started in 1980, the first of a series of anti-apartheid community newspapers, with a circulation that grew to 20 000. The paper struggled financially, but was helped by small donations. Another newspaper falling under protest press was *Work in Progress*; this was a trade union publication that was founded by *University of Witwatersrand* postgraduate students. This publication was able to challenge a number of bannings and made inroads into the Publications Act committee declaring its material undesirable 11.

¹¹ http://safrica.info/ess info/sa glance/constitution/971558.htm

South is another newspaper that was established in 1987 by a group of coloured journalists who formerly worked for Cape Argus. Founded by Alan Boesak, the former UDF leader and activist, South was established to fight "racism in the newsroom". Adhikari (2000:327-369) has written extensively about South and posits that South was unique in such a way that "it was the only left-wing commercial newspaper in the region in the last decades of the apartheid era. It is further stated that South lifted media activism to a higher plain". According to Adhikari (2000:346), this newspaper gave a lot of effort but failed to sustain circulation figures of more than 10 000 and, struggling to make a profit, finally closed.

Other newspapers falling under the Alternative Press include *Indicator* and *New Nation*. *Indicator* was published biweekly and was largely aimed at a broader black community while *New Nation* was more African oriented and openly analysed national politics (Jackson 1993:47. Also falling under this category is *New Nation*, *New African*, *Grassroots*, *Samstaan*, *Work in Progress*, *Staff Rider* and *Sapsu Nation* (a student publication of that time). *Grassroots* was mainly concerned with expressing the views and aspirations of communities and workers, was aimed at the coloured and African readership in the Cape Peninsula. *Grassroots* formed part of the new alternative press.

The *Weekly Mail* and the *Vrye Weekblad* both based in Johannesburg were unusual in that they were both intended for a white audience. Another alternative press paper was *Mail*, respected as the most professionally produced paper (Jackson 1993:47). Democracy in South Africa saw the death of the liberation movement support media. Some of the newspapers that died include *South*, *Vrye Weekblad*, *New Nation* and *Work in Progress* (a magazine). Newspapers like *Mail & Guardian and Sowetan* are still in existence (Berger 2001:153). Most of the other newspapers were closed mainly because of problems relating to funding and limited advertising resources (Jackson 1993:61-63).

According to Switzer (2000:39) most of these newspapers were shut down because "they were unable to survive without foreign subsidies". Ownership of newspapers has also changed to a large extent. The papers that previously belonged to the government are

now privatised allowing them an opportunity to serve their watchdog role. This change is also visible in the newsroom. There is also a considerable difference in editorial independency allowed to editors.

4.2 **Newspaper Firms in South Africa**

South African media is currently dominated by a handful of large corporations, with their interests ranging from newspapers to magazines and the Internet. Major media owners are Media 24, Independent News and Media, Avusa and Caxton and the CTP Group. These four own almost all the major newspapers and community newspapers, most of the consumer magazine titles, a slew of specialist magazines and have a finger in Internet and broadcast pies. 12

Berger (2001:152) posits that media prior to 1993 "was entirely a state monopoly", with ownership limited to only four white-owned groups. Tomaselli and Tomaselli (2001:131) contend that during the 1994 elections the print media industry was largely dominated by two media groups, Argus Holdings Ltd, Times Media Ltd and two Afrikaans language groups, Perskor and Nasionale Media BPK. Berger (2001:153) argues that owing to foreign investment, South African print media was given a chance to end the white South African monopoly. As a result, competition and arguable diversity were also added.

The media in South Africa is not a neutral purveyor of information – writers and editors pre-select and construct information for specific audiences. South African media can be divided into commercial media, which is mainly sold to make profit, state-funded public service media, and community media (Duncan & Seleone 1998:10). The South African media is characterised by its lack of diversity in all respects. The debate in this study is situated within this context¹³. The following section discusses each newspaper firm in detail.

65

¹² www.southafrica.co.za13 www.SouthAfrica.info.

4.2.1 Naspers

Nasionale Media BPK also known as Naspers, is a multinational media group with its principal operations in print media – newspapers, magazines, printing, book publishing and private education and electronic media such as pay television and Internet service provision. Formed in 1915, the first newspaper of Naspers, *Die Burger*, was the official mouthpiece of the National Party in the Cape.

Print media comprises two segments: Media24 and book publishing and private education. Media24 is the largest publisher and publishes five national dailies: *Daily Sun, Die Burger, Beeld, Volksblad, Sondag, Son* (daily and weekly), *Witness, City Press, Beeld, Rapport and Soccer Laduma*. The company is owned by Naspers, which owns the MIH group, the owner of Multichoice.

Daily Sun, the largest daily newspaper in South Africa, competes directly with the Sowetan. It is the first South African tabloid aimed at the black working class. Initially met with disdain by the established press, its huge sales – and the fact that it has made new newspaper readers out of millions of South Africans – have earned it some respect. It is sold in Gauteng, Limpopo, Mpumalanga and North West. Average daily sales: 301 800; average daily readers: 2.29 million. On Sundays, Naspers, publishes Rapport, City Press and Sunday Sun, printed in four cities and distributed nationally 14.

4.2.2 Independent News and Media

Formerly known as Independent Media or Independent Newspapers, Independent News and Media owns 14 national and regional newspapers in most of the major cities such as *The Star, Cape Argus, Isolwezwe, Daily News, Daily Voice, Cape Times, The Mercury, Pretoria News, Diamond Field Advertiser, Business Report, The Post, Sunday Tribune, Independent on Saturday, Sunday Independent and the voice (www.iol.co.za).*

¹⁴ http://safrica.info./ess info/sa glance/constitution/news.htm

Independent Media is owned by Irish tycoon Tony O'Reilly who first acquired a stake in the local newspapers in 1995 when he took control of Argus Newspapers in the 1990s and renamed it Independent Newspapers. The company has some presence in the community newspapers and magazines markets. Its Internet presence is IOL, which carries news, classifieds and information from all its newspapers.

As noted in Table 4.1 below, *The Star* is published in Johannesburg and distributed throughout South Africa, with most sales in Gauteng. Once aimed exclusively at the white market, today over 50% of *The Star*'s readers are black. It is owned by Independent Newspapers. Launched in Grahamstown in the Eastern Cape in 1887 as the *Eastern Star*, the paper moved to Johannesburg in 1889. The paper is the flagship daily newspaper, enjoys average daily sales of 2.8 million copies, reaches about 63% of English newspaper readers and receives about 48% of total advertising spend in the newspaper market¹⁵.

15 http://www.mediclubsouthafrica.com

Table 4.1: Independent Newspapers

| Type of Publication | City/Region | | |
|------------------------------|--|--|--|
| Daily | Cape Town – Western Cape | | |
| Daily | Johannesburg – Gauteng | | |
| Daily | Durban – KwaZulu-Natal | | |
| Daily | Durban – KwaZulu-Natal | | |
| Daily | Cape Town – Western Province | | |
| Daily | Durban – KwaZulu Natal | | |
| Daily | Pretoria – Gauteng | | |
| Daily | Kimberley – Northern Cape | | |
| | | | |
| Daily (inserts in four daily | Durban, Johannesburg and Cape | | |
| papers) | Town | | |
| Bi-weekly | Durban – KwaZulu-Natal | | |
| Weekend | Durban – KwaZulu-Natal | | |
| Weekend | Durban – KwaZulu-Natal | | |
| | | | |
| Weekend | Johannesburg | | |
| Daily | Cape Town Western Cape | | |
| | Daily Daily Weekend Weekend Weekend | | |

Courtesy of Independent Newspapers (www.iol.co.za)

4.2.3 Avusa Limited

Avusa Limited, formerly known as Johnnic Communications/Johnnic Communications or Johncom, Avusa), its beginning can be traced back to 1902. Today, Avusa is the leading provider of news and information. It is owned by a coalition of black business groups and trade unions, the National Empowerment Consortium. The consortium bought TML from Anglo American's Omni Media Corporation in 1996. The National Empowerment Consortium's takeover of Johnnic Communications coincided with the establishment of a joint venture between British group Pearsons and TML, under which the former acquired half of TML's *Business Day* and *Financial Mail*. In Buffalo City in the Eastern Cape, the company publishes the *Daily Dispatch*, which was edited by Donald Woods from 1965 until his arrest and banning in 1977 for exposing government responsibility for the death of Steve Biko.

Some of the newspapers that are published by Avusa Limited are *Sunday Times* which is one of South Africa's biggest Sunday newspapers as well as the *Sowetan, Business Day, Sunday World, Daily Dispatch, The Herald, Weekend Post, Algoa Sun, Ilizwi* and *Our Times*. It is important to state that the group also owns music publishing company, Gallo Music, movie distributor Nu Metro, the Exclusive Books chain of Bookshops, and Struik, New Holland and Map Studio Publishers¹⁶ Its magazine division has titles in the consumer, specialist, business and medical sectors, while Picasso Headline publishes a range of titles and takes care of custom publishing for other organisations.

Note that Avusa is leaning towards broadcasting, with a stake in *Home Channel* and *Summit TV*, which are both carried through DStv. Its online presence is strong, with Internet sites for many of its newspapers. It also owns the citizen journalism site, Reporter.co.za and has I-Net Bridge, an electronic provider of data, news and applications to the professional investment community and corporate market.

¹⁶ www.southafrica.info; www.johnnic.co.za

4.2.4 CTP/Caxton Publishers and Printers LTD

Caxton and CTP Publishers and Printers are 38% owned by Johnnic Communications. The group publishes *The Citizen*. *The Citizen* is published six days a week and is distributed mainly in Gauteng. *The Citizen* has daily sales of 90 978 and an average of 466 000 daily readers. The newspaper division stables 88 free and sold Caxton owned or co-owned titles as well as 30 regional and community newspapers. According to the World Wide Web, the magazines published total 15 titles, including *Farmers Weekly*, the oldest magazine in South Africa. Just like other media companies, CTP is not only limited to printing magazines and newspapers, it is also involved in other printing business such as book printing and stationery¹⁷.

4.2.5 **M&G Media**

Although it is a small group, M&G Media must be included in any overview of the South African media. The company publishes the highly regarded *Mail and Guardian*, which began life in 1985 as the *Weekly Mail*, a newspaper that earned international respect for its fearless exposure of apartheid-era abuses. This is one of the few newspapers that managed to survive to see democracy. Its target audience remains the more serious reader.

The *Mail and Guardian* was also the first South African – and African – newspaper to have its own website, with its original editors, Irwin Manoim and Anton Harber, launching M&G Online in 1994. The site is now co-owned by Internet service provider MWeb. M&G media is the joint owner of the M&G online, along with the Internet service provider MWeb. M&G Media is 87.5% owned by Newtrust Company Botswana Limited, which is owned by a Zimbabwean with minority shareholders making up the rest.

¹⁷ www.southafrica.info

4.3 Challenges Faced by South African Media, and Developments

The historical background of South African media is not complete without discussing some of the challenges that are facing the media. This section looks at some of the developments as well as the media-state relationship.

Hindrances such as commercialisation and market driven journalism make it increasingly difficult for the news media to play the role it should in a democracy. Most media accept that South African media is commercialisation of news. This is well illustrated in the ANC document, Transformation of the Media:

The freedom of the South African media is today undermined not by the State, but by the various tendencies that arise from the commercial imperatives that drive the media. The concentration of ownership, particularly in the print media sector, has a particularly restrictive effect on the freedom of the media. This is not a particularly South African phenomenon. Around the world, consolidation of the media groups and the drive to maximise profit – have led to a global homogenisation of news. Despite protestation to the contrary, there is an increasing number of instances where the supposedly sacred separation between management and newsroom is breached, where commercialisation considerations influence editorial control.

(Transformation of the Media, ANC documents, 2007).

According to Steyn and De Beer (2004), skills decline is another reason for concern in the developing democracy. Other issues such as juniorisation of newsrooms and moving of senior journalists from writing to managing are some of the issues that are considered as contributory factors in the problems facing the media. Most of these problems were discussed in an annual general meeting of the South African National Editors Forum (Sanef). The same meeting also noted that no research had been conducted to measure the extent of these problems.

Sanef then undertook to conduct a situation analysis of journalism reporting, writing, and accuracy skills among the reporters of between 2 - 5 years experience (Steenveld 2002). According to Tsedu (2002), the Sanef audit was one way of determining how true, relevant and deep the problem was.

The results confirmed that there is a low level of competency within the media. The Sanef skills audit also discovered that in most reports only one source is used which leads to limited and shallow reporting. The reviewed literature in chapters 2 and 3 does not provide what should be covered by the media in a properly functioning democracy. Most of the literature reviewed covers what the media should cover during elections (Ramsden, 1996). Ramsden (1996) also points out that media scholars always stress the importance of freedom of the press in a democracy but little is said about what and how stories should be covered in order to fulfil that role.

Jacobs (2004:159) states that there is a strong need to acknowledge the role that was played by the media during the first elections in South Africa in 1994. He attributes peace during elections to the way the media covered the stories relating to the elections. He, however, points out that journalists of that time were largely white, male and middle class. According to Nelson Mandela in his address to the International Press Institute Congress, (1994), this resulted in their providing "a skewed picture of African experience" (Mandela 1994)¹⁸. Fourie (2002:20), concurring with Mandela, posits that during the apartheid era, African media operated according to the Western system and with Western news values resulting in African news being largely limited to famine, war and corruption. During the apartheid era there was little media freedom and Fourie (2002:22) comments that journalists were not even allowed to quote leaders of the anti-apartheid struggle such as Nelson Mandela. Journalists who broke these laws were either

¹⁸ Mandela, N (1994) Address to the International Press Institute Congress. Retrieved April 18, 2008, from http://www.anc.org.za/ancdoc/history/mandela/1994/sp40214.html

imprisoned or had their newspapers banned.

One of the most important achievements in transforming the South African media environment since 1994 has been the removal of censorship and the commitment of the new democratic order to guaranteeing the rights of South Africans to freedom of expression. However, the exercise of this right is constrained by the lack of media diversity and the domination of commercial considerations within the media (ANC Policy Documents, 2007).

4.4 Media-State Relationship after Apartheid

There were visible strains on the relationship between the post-apartheid media and government¹⁹. According to Wasserman and De Beer (2005:43), the initial years of democracy were largely marked by mutual mistrust between the media and government. Although elections on 27th April 1994 marked the official end of the apartheid system and the beginning of democracy, it is important to note that the very beginning of democracy was also the beginning of bad blood between the press and the government.

Statements made by the former South African presidents, Nelson Mandela and Thabo Mbeki, highlight the fragility of these relations which according to Wasserman and De Beer 2005:193 had been under severe strain on many occasions since the advent of democratic elections. Bad relations between the government and the press can also be traced back to the apartheid era, for example, F.W. de Klerk often accused English journalists of negativism towards the National Party.

The media and the government have been at loggerheads on several occasions and this sour relationship was highlighted by a two-day Sun City conference. The former

The former President of South Africa, Thabo Mbeki lashed out at the media for what he called "unprofessional reporting" (Megwa 2001:281-5). Johnston (2005:13) also notes some of the controversial issues highlighting bad blood between the government and the media: angry government reaction to newspaper speculation that Parks Mankahlane, former press aide to former South African President, Nelson Mandela, died owing to Aids-related causes. And court action for defamation by the Mail and Guardian newspapers against Jeff Radebe.

president of South Africa, Thabo Mbeki, members of the cabinet and members of Sanef initiated the dialogue between the journalists and the media. This conference was organised after a number of statements had been issued by the government voicing its concern about the media's apparent "reluctance to embrace the concept of national interest". Note that this meeting was not productive as planned because the media did not accept "national interest" as an adequate description of its role in post-apartheid South Africa. Instead, it asserted its role as being in the service of "public interest" in its submission to the Broadcasting Amendment Bill in 2003.

The Media Institute (MISA) stated it is our contention that the use of the term national interest in relation to newsgathering and dissemination is too restrictive and can have a narrow political connotation. Journalists work in the public interest, which is much wider. Politicians of the ruling party may decide that there should be secrecy over an issue but a sounder, and much wider base might override national interest. Chapter two of the Constitution protects the public interest". Wasserman and De Beer (2005:45) posit that these two concepts encapsulate the different perspectives from which the media and government view the media's role in society. It is important to mention that the meeting led to the establishment of a Presidential Press Corps which was meant to create a better flow of information from government to the public.

Johnston (2005: 13-14) argues that hostility between the press and the government is not something new and should not raise too much concern because in other democracies with free presses, government and media do not get along let alone trust each other. This is usually interpreted as healthy for democracy. Wigston (2001:3) notes that the relationship between the state and broadcasting is different from that between the state and the press.

This study focuses on print media only but it is significant that, since 1994, the South African Broadcasting Corporation (SABC) has been dramatically transformed with the proliferation of new radio and television services, diversification of new programming and restructuring. According to the Bill of Rights contained in South Africa's constitution

(Act 108 of 1996), everyone has the right to freedom of expression and freedom to receive or impart information or ideas. The media diversity in South Africa is regarded as a good starting point for democracy. It is, however, important to pose a question from this: Is there diversity in the content of the added media?

4.5 Conclusion

The chapter has briefly outlined the history of South African print media and noted that a great deal has been written on print media as compared to broadcasting. Reasons for this imbalance are not very clear but it safe to assume that most media scholars are more interested in the print media because of its rich history. The chapter noted that before democracy, the media was more focused on fighting apartheid and that South African print media is characterised by different newspaper firms that are largely conglomerates. It also noted that with the birth of democracy in 1994, the media environment underwent many changes. One of the notable ones was that the SABC was transformed from being a state broadcaster to being a public broadcaster. Other changes highlighted were that through democracy South Africa's newspapers were freed from all restrictions.

The next chapter discusses the methodology chosen for this study. The advantages and disadvantages of each method chosen are explored.

CHAPTER 5 RESEARCH METHODS

5.0 INTRODUCTION

This chapter presents the research methodology used in this study, discussing the research techniques as well as the design implemented for the study. The chapter also outlines the feasibility of the methods chosen. This is done by studying each method chosen, taking into account the advantages and disadvantages of each method.

The chapter later explores issues such as selection of articles, data collection, data coding and data analysis. As stated in chapter one, the focus of the study is to analyse all the front pages and editorials of the four newspapers: namely, *The Star, Sowetan, The Citizen* and *The Natal Witness*. The researcher analysed the articles from the beginning of the Schabir Shaik Trial to his final sentencing in court. The following section discusses the methods employed in this study as an effort to find answers to the research questions posed in chapter 1 of this project. It is important to be reminded that this research project is aimed at answering the following questions:

- To what extent did the selected print media in South Africa conform in the coverage of the Schabir Shaik trial?
- What are the implications of such conformity for democracy?
- What can be done to increase content diversity in the print media?

5.1 Methodology

Because of the complexity of this study, qualitative content analysis supported by thematic analysis and news discourse analysis was predominantly used as a data collection method. This combination of methods is referred to as triangulation. Denzin in Mouton (1996:156) defines triangulation as the "use of multiple methods of data collection". Wimmer and Dominick (1997:111) define the term as "the use of both quantitative and qualitative methods to fully understand the nature of the research

problem". Bertrand and Hughes (2005:239) maintain that it is the use of two or more research data gathering methods. This argument is elaborated by Van Dijk (1988:66) who posits that the combination of quantitative and qualitative analysis is the adequate approach to the study of mass media messages.

The advantage of using multiple methods (triangulation) is that it increases the reliability of the study (Mouton, 1996:156). According to Cresswell (2003:100), this integrated approach to document analysis converges both broad numeric trends from quantitative research methods and the details of qualitative methods. It is important to note that the combined use of qualitative and quantitative research methods in data collection is widely accepted for social sciences research. For instance, Morris Janowitz in his study (1952/1967) of the community press used both qualitative and quantitative research methods and they complemented each other well.

The researcher used quantitative content analysis, since it is imperative to group the themes and count how many articles and editorials adopted similar views and how many did not. Qualitative content analysis is the first method employed followed by thematic content analysis. Later, two or three themes will be selected for discourse analysis.

Gunter (2000:82) defines qualitative content analysis as the method that "emphasises the capacity of texts to convey multiple meanings, depending upon the receiver". He further elaborates that the central thrust of content analysis is to provide a descriptive account of what the text contains and to do so in a fashion that can be easily replicated by another researcher. Krippendorff (2004:88) posits that qualitative researchers search for multiples of interpretations by considering diverse voices and alternative perspectives. This method is considered by some scholars to be useful. For example, Tuchman (1991:79) states that the most valuable research done in his time is qualitative content analysis. Content analysis is "well suited to the study of communication" (Du Plooy 1995:152). According to Du Plooy (1995:165), one of the strengths of content analysis is that it combines well with other research methods. For example, in this study the researcher will combine qualitative content analysis with news discourse analysis.

Other advantages of using content analysis are that it is not expensive especially when the data being investigated is available. Content analysis is unobstructive; in other words, it does not interfere with people's lives, and problems such as people refusing to give interviews are prevented (Bertrand and Hughes 2005:184). Some of the other strengths of content analysis are that it is easy to find material from the past and it is considered the preferred method when it comes to managing large amounts of data (Bertrand and Hughes (2005:184).

A limitation of this method is that it is time consuming and relies heavily on "recorded" information (Du Plooy 1995:166). It is also limited in analysing data that has been already recorded, and sometimes it becomes difficult for the researcher to find all the data. As stated earlier, the researcher will use Sabinet, a comprehensive online database covering a wide range of information including media clippings. As such, no missing data is anticipated.

Content analysis involves ten important steps (Wimmer et al., 1997:116): namely, formulating the research question or hypothesis; defining the population in question; selecting an appropriate sample from the population; selecting and defining units of analysis; constructing categories of content to be analysed; establishing a quantification system; training of coders and conducting a pilot study; coding the content according to established definitions; data analysis, and drawing of conclusions.

The sampling in content analysis often takes place in more than one step (Gunter 2000:65). First, using the purposive sampling procedure, the researcher selects the newspaper titles to be used. Du Plooy (2001:114) defines purposive sampling as a "sampling method that relies entirely on the judgment of the researcher".

To ensure that the sample is representative of daily newspapers, the researcher will use four daily newspapers: namely, *The Citizen, The Star, Sowetan* and *The Natal Witness*. The selection of these four newspapers is also influenced by the fact that three of them

are South Africa's biggest national daily newspapers in circulation and, more importantly, these newspapers are from different stables. *The Citizen* is published six days a week and distributed mainly in Gauteng. The newspaper is co-owned by Johnnic Communications and CTP/Caxton. *The Citizen* has average daily sales of 90 978 and an average of 466 000 daily readers. *Sowetan* is *Daily Sun*'s main competition, also aimed at an English-literate black readership. Initially distributed as a weekly free sheet in Soweto, the paper was transformed into a daily in sales of 118 261 and average daily readers of 1.54 million.

The third newspaper that the researcher will use is *The Star*. It is published in Johannesburg and distributed throughout South Africa, with most sales in Gauteng. Once aimed exclusively at the white market, today over 50% of *The Star* readers are black. It is owned by Independent Newspapers. *The Star* has average sales of 171 542 and averages /616 000 daily readers.

The fourth newspaper chosen is *Natal Witness*. This is the South Africa's oldest newspaper and serves English readers throughout Kwazulu Natal with most of its readers in greater Pietermaritzburg and inland KwaZulu. Owned by Media24, it was formally known as the *Witness*. It has average daily sales of 23 804 and averages 167 000 readers. The reason for inclusion of the *Witness* is that it is KwaZulu Natal's major publication and as Shabir Shaik is a businessman from KwaZulu Natal, it will be interesting to see how the media in that area covered this particular case.

The researcher ensured that all the newspapers are from different publishing houses because it is documented that media ownership is one of the variables that can influence media content. There is speculation that the content of newspapers is largely determined by the ownership of the newspapers. It is alleged, for example, that newspapers from the same stable produce the same content (Schudson, 2000:178). None of the weekend newspapers such as The *Sunday Times* or *City Press* were used, as their news production processes are informed by different factors from those of the daily newspapers.

The second step of the sampling process was for the researcher to select manually the actual articles to be analysed. This step was guided by the duration of the trial. The Schabir Shaik trial began in October 2004 and ended in May 2005. Because many articles were published during the eight months' period of the trial, the researcher selected only front page stories and editorials. Front page stories were chosen because their visibility made them more likely to be read by many readers, and were deemed important by editors and by readers alike. Tuchman (1978:97) states that readers consider front pages as representing the "reality" of the story. Editorials, on the other hand, represent the internal position of a newspaper (Tuchman 1978). The researcher paid particular attention to the tone of the editorials to confirm whether the editors were fulfilling the media role of providing a forum for critical analysis.

In working out this methodology the researcher relied heavily on archival material as well as the Internet. All newspaper articles on the Schabir Shaik trial were secured from Sabinet using key words such as trial, corruption, Schabir Shaik. Sabinet offers research information from major South African media and Sapa databases. Media clippings are from 1997 onwards, with an archive backdated to 1978 (Sabinet 2005); as such, it is hoped that there was no missing data.

5.2 Thematic Analysis

Altheide (1996:42) defines thematic content analysis as the methodology that seeks to understand the process and character of social life and to arrive at meaning and process. Themes can also be referred to as issues, topics and events that the media selects when reporting. Neuendof (2002:34) defines thematic content analysis as "the scoring of messages for content, style or both for the purpose of assessing the characteristics or experiences of persons, groups or historical periods". Jensen (2002:251) defines this method as "an attempt to identify, compare and contrast meaning, as they emerge from and recur in several contexts. What distinguishes thematic coding from much quantitative content analysis is the emphasis on defining each of the elements in relation to their

context".

Kelly (1999:412) states that thematic development/analysis is a kind of pattern – finding process where we identify a type of occurrence by virtue of its being perceived as underlying "a common form" found in different contexts. Kelly (1999:412) further posits that in identifying themes it is important for one to overlook certain contextual differences in the things being compared. Altheide (1996:31) refers to themes as boundaries or parameters for discussing a particular event by bracketing or marking off something as one thing rather than another.

The first step in doing thematic analysis is to collect data. All data relating to the Schabir Shaik trial was collected. Patterns of reporting were then noted and listed. The next step to thematic analysis was to relate to already classified patterns. This step was followed by combining and cataloguing the related patterns into sub-themes. The next step was to build a valid argument for choosing the themes. This was done by reading the related literature. Once the themes were collected and the literature studied, the researcher was able to formulate theme statements to develop the story line. When the literature was interwoven with the findings, the arguments that the researcher constructed would be able to stand with merit.

The following section discusses the advantages and disadvantages of discourse analysis because it is another method that was used in this study.

5.3 Discourse Analysis

As noted earlier, news discourse analysis is another data collection method that is employed in this study. Discourse analysis postulates that news is a presentation of the world in language and it is best suited to analysing newspaper reports (Gunter, 2000:87). The underlying assumption of discourse analysis is that language plays an important role in the construction of reality. Van Dijk (1997:27) adds that discourse analysis is the method that analyses the deeper meaning by going to the extent of questioning the

definition itself. According to Bertrand & Hughes (2005:94), discourse analysis provides a way to understand the deeper structure of any text as well as its surface content. This, therefore, means that using this method will assist the researcher in assessing the extent of conformity /construction in the selected print media.

In most cases, discourse analysis is combined with other research methods (Gunter 2000:88). The researcher in this study has done exactly that. The advantage of using discourse analysis is that it deals with any personal bias stemming from the use of content analysis, and it furthermore helps to validate results obtained in the content analysis methodology.

For the purpose of this research, one or two themes were chosen and discourse analysis was employed. In using discourse analysis, the researcher is required to determine broad patterns of the data. Such procedure is also a general form of content analysis (Gunter 2000:88). As already stated, the main weak point of discourse analysis is that it tends to be very biased (Gunter 2000:88). The researcher will be using multiple methods, so that the content analysis will counteract this limitation.

Table 5.1: Readership profile of the four newspapers during the period of the study was as follows:

| | Total | The | Sowetan | The Star | Witness |
|-----------------------------|----------|---------|---------|----------|---------|
| TOTAL | Universe | Citizen | 2007 | 1001 | 10/ |
| TOTAL | 31305 | 496 | 2097 | 1081 | 126 |
| Gender | | | | | |
| Men | 15600 | 290 | 1300 | 685 | 68 |
| Women | 15705 | 206 | 797 | 396 | 58 |
| Age | | | | | |
| 16-24 | 8258 | 54 | 455 | 162 | 16 |
| 25-34 | 7524 | 106 | 614 | 313 | 37 |
| 35-49 | 8389 | 201 | 675 | 380 | 28 |
| 50+ | 7134 | 136 | 353 | 226 | 45 |
| Race | | | | | |
| Blacks | 23617 | 286 | 2023 | 732 | 42 |
| Coloureds | 2758 | 19 | 41 | 49 | 5 |
| Indians | 831 | 17 | 9 | 45 | 33 |
| Whites | 4099 | 174 | 25 | 254 | 46 |
| Level of education | | | | | |
| No School | 1044 | 0 | 8 | 0 | 0 |
| Some Primary | 2345 | 4 | 32 | 6 | 0 |
| Some High | 12462 | 115 | 672 | 199 | 33 |
| Matric | 8927 | 190 | 906 | 431 | 57 |
| Technikon Diploma Degree | 1741 | 63 | 176 | 193 | 12 |
| University Degree | 1075 | 53 | 114 | 126 | 13 |
| Other Post Matric | 1163 | 55 | 120 | 110 | 8 |
| Household Income | | | | | |
| R1-R299 HH -R499 | 906 | 3 | 43 | 0 | 0 |
| HH | | | | | |
| R500-R899 HH | 3161 | 14 | 87 | 10 | 3 |
| R900-R2499 HH | 8432 | 60 | 320 | 81 | 15 |
| R2500-R3999 HH | 4143 | 43 | 291 | 78 | 9 |
| R4000-R6999 HH | 4926 | 64 | 392 | 147 | 23 |
| R7000-R11999 HH | 4655 | 107 | 520 | 263 | 33 |
| R12000+ HH | 5082 | 204 | 443 | 502 | 43 |

Source: AMPS 2008 Main Branded AB (Jan

2008-Dec 2008) Weight: Population

5.4 Time Frame

The study analysed the four selected newspapers, from Jan 2001 until the time of sentencing on the 8th of June 2005. The researcher, however, included the 14th of June because some of the newspapers may have published their stories after this date. This time frame was chosen because the Schabir Shaik trial was the biggest corruption trial in the history of South Africa involving people in high positions such as Jacob Zuma. It has been labelled the trial of the decade (*Natal Witness*, 02 June 2005, p1).

5.5 Selection of Articles

As earlier mentioned, the focus of the study is threefold. First, using the case of the Shabir Shaik trial, the study seeks to investigate the extent of conformity in the news output of the selected newspapers. The study used purposeful sampling (Dominick and Wimmer 1987).

This section provides details of the articles that were selected. As previously stated, because so many articles were published during the eight-month period of the trial, the researcher selected only front page stories and editorials.

Front page stories were chosen because their visibility makes them more likely to be read by many readers, and they are deemed important by editors and readers alike. Tuchman (1978:97) states that front pages are considered by readers as representing the "reality" of the story. The selection of the front-page story is based on the claims that stories that appear on page one are more newsworthy than others. These claims were confirmed in the study by Kisuke (2004:62).He posits that "the selection of front page stories calls for a daily editorial conference involving a group of editors whose task is mutually to negotiate judgments of these stories and accomplish newsworthiness together".

Page one is considered as the newspaper's showplace and additionally a measure of a reporter's success. Croteau and Hoynes (2000:126) posit that the selection of the front page story "identifies the editors' selection of the most important event of the day". This influences which stories will be most visible.

As stated, the researcher also used editorials. Tuchman (1978) posits that editorials represent the internal position of a newspaper. The researcher paid particular attention to the tone of the editorials to see if the editors were fulfilling the media role of providing a forum for critical analysis.

5.6 Data Collection

Articles covering the Schabir Shaik trial were secured from Sabinet, a comprehensive online database. ²⁰ The articles on the Schabir Shaik trial were retrieved using keywords such as "trial" and "Shaik Schabir" and "corruption". "Schabir Shaik" was used as the key word meaning that any story making reference on Schabir Shaik was retrieved. Sabinet is structured professionally and allows the researcher to specify the name of the newspaper and one may even specify the name of the newspaper and the period that is needed. All the articles were retrieved from all four newspapers.

5.7 Conclusion

This chapter discussed the methods used for the study. Advantages and drawbacks of each chosen method were outlined. A combination of qualitative content analysis, thematic analysis and discourse analysis was discussed. For the purpose of this study the combination of the three methods was necessary to bring insight and perspective into how the case was covered.

The following chapter covers the integration of data collected, as well as a review and analysis of the same. It is also expected that the next chapter will answer the key research questions mentioned at the beginning of the study.

²⁰ Sabinet Online Academic and Library Division is dedicated to serving all academic institutions, Government Libraries and other library clients with value added access to information services, cataloguing and interlibrary support services. According to their website, they further strive to provide products and services of high quality as well as exceptional client service. http://aclib.sabinet.co.za/about.html.

CHAPTER 6

PRESENTATION OF FINDINGS AND DATA ANALYSIS

6.0 INTRODUCTION

This important chapter presents the results of the study and contains what the research is all about. It is through this chapter that the researcher determines whether the aims of the study were achieved. It kicks off by providing a brief history of the Schabir Shaik trial in order to offer an understanding of the context in which the study is happening. The second part contains an analysis of themes and categories found in the data, and is followed by an analysis of news values.

It must be noted that the findings and interpretation thereof belong to the researcher and they are, according to Wimmer and Dominick (1987), prone to bias/personal perception. It will, therefore, be incorrect to posit that results presented here are completely free from bias but they are useful in terms of understanding how the media functions. Most positivists claim objectivity when interpreting results and that is actually not possible according to Wimmer and Dominick (1987). To minimise the possibility of bias in interpreting the results, the researcher has adhered to the rules and procedures of the methodologies chosen.

This chapter presents answers to the key research questions posed at the beginning of the study – an assessment of the four newspapers to ascertain the degree to which they correctly reflect that designation and whether their performance is inimical to the growth of democracy or supportive of it. This research project was aimed at answering the three questions:

- o To what extent did the selected print media in South Africa conform in the coverage of the Schabir Shaik trial?
- o What are the implications of such conformity for democracy?

• What can be done to increase content diversity in the print media?

The researcher analysed all front page articles and editorials from January 2001 to the 14th of June 2005. Although Schabir Shaik was sentenced on the 8th of June 2005, receiving a sentence of 15 years in prison, the researcher did not conclude the analysis on the 8th but decided to continue to the 14th of June 2009 because media coverage of the sentencing was sustained.

6.1 Data Collection and Data Analysis

In order to answer the research questions, the researcher used both qualitative and quantitative analysis. The front page news articles about Schabir Shaik were all clustered according to what Rossman & Rallis (1998:171) refer to as chunks or categories.

Categories can also be called frames. It is argued that classification and labelling of events into discrete categories is a central part of most research in the social sciences because it enables investigators to identify and group similar instances of a phenomenon together for the systematic study and is important to any quantitative analysis. The success of content analysis generally lies in determining categories and ensuring that they do not overlap. Each news article was evaluated, itemised and placed in the appropriate column in one of the researcher's pre-defined categories. Articles appearing on any page other than the front page or editorial were not used. It is important to state that from the articles, the researcher checked the following:

• The types of sources used in the story. A closer analysis of the sources helped the researcher to determine whose voices were given certain prominence and which ones were not. The rationale for this analysis stems from a position that "foregrounded and backgrounded information" (Goatly 1997). This originated from Jan Mukarovsky, a Prague literacy scholar of the 1930s, who defines it as "The opposite of automatization, that is, the de-automatization of an act; the more an act is automatized, the less it is consciously executed; the more it is foregrounded, the more completely conscious does it become [A]utomatization

- schematizes an event; foregrounding means the violation of the scheme..."
 (Mukarovsky 1970:43). Many meanings are attributed to the term
 "foregrounding" but in this context the one that implies the act of reading a
 particular set of information is given a special prominence and "backgrounding"
 simply refers to the opposite of foregrounding.
- The researcher also checked articles covering the same event. The rationale behind this was to find out to what extent the media conforms when covering a story. As argued, by Radebe (2006:92), "diversity and differences in reporting should be encouraged as it is likely to promote divergent ideas which are necessary for creating diverse media with diverse view points, which they perceive as good for democracy".
- Headlines of the articles were also analysed. Hartley (1982:155) posits that headlines can be checked by studying their significance and pondering on what alternative possibilities one can suggest to "make sense" of the particular story or event. He adds that another method for analysing newspapers is comparison. He reasons that comparison brings out the distinctive features of the study by showing which characteristics are unique to themselves, which are common with others and what possibilities are absent altogether (Hartley (1982:155). Headlines are normally good starting points when checking issues like objectivity, bias, and market driven journalism. Jones (2005:154) argues that headlines perform four main functions: they summarise the story for the reader, in other words, serve a fundamental framing function; upgrade or downgrade topics; attract readers' attention and serve as a guide to the themes. For the purpose of this analysis exercise, the researcher selected some headlines.
- The researcher also checked the values that were prominent during the Shaik trial. Journalists share many societal values and understanding of news values (Gans 1979; McNair 1998:6-7). The question is then to what extent news values contribute to conformity in the media.

88

Table 6.1 Overall coverage (front page articles) of the Schabir Shaik trial by *Sowetan*, *The Star*, *The Citizen and Natal Witness*.

| Title | 2001 | 2002 | 2003 | 2004 | 2005 | Totals |
|---------------|------|------|------|------|------|--------|
| The Citizen | 3 | 2 | 3 | 8 | 5 | 22 |
| Sowetan | 1 | 1 | 9 | 0 | 0 | 11 |
| The Star | 2 | 1 | 18 | 22 | 19 | 62 |
| Natal Witness | 6 | 1 | 1 | 14 | 18 | 40 |

The bulk of the Schabir Shaik trial coverage was in 2004. This is due to the fact that he was arrested in that year for being in possession of illegal documents. Year 2002 and 2003 were quiet years but the story was still in the media with developments here and there. The trial began to make front page headlines again in May 2004 with intense coverage for the eight-month period from May 2004 to June 2005. It must be repeated that as this was a big case, the researcher chose to limit the data by focusing on only some front-page stories and editorials (January 2001 to 14th of June 2005). Sentencing took place on the 8th June 2005 but the analysis was stopped on the 14th because some of the newspapers might have published their articles about sentencing later.

It generally appears (see Table 6.1) that *The Star* had more articles (62) on Schabir Shaik than other newspapers. This was followed by *Natal Witness* which covered the case intensively with 40 front page articles. This may be attributed to the fact that Schabir Shaik is from Durban. *The Citizen* and *Sowetan* covered the Schabir Shaik trial reasonably. *The Citizen* had a total of 22 articles while the *Sowetan* had only 11 front page stories. All four newspapers wrote more articles about Schabir Shaik but as they were not front page stories they were disqualified from inclusion in the analysis.

6.1.1 Analysis of the editorials

Table 6.2: Editorial pieces on Schabir Shaik trial as covered by *Sowetan, The Star, The Citizen and Natal Witness*

| Title | 2001 | 2002 | 2003 | 2004 | 2005 | Totals |
|---------------|------|------|------|------|------|--------|
| Natal Witness | 0 | 0 | 1 | 4 | 3 | 8 |
| Sowetan | 0 | 0 | 0 | 2 | 4 | 6 |
| The Citizen | 1 | 0 | 0 | 1 | 2 | 4 |
| The Star | 1 | 0 | 0 | 2 | 0 | 2 |

The researcher analysed all the editorial pieces from all four newspapers. The same approach used for analysing front page articles was used in the analysis of editorials. Themes were extracted and then all articles were analysed by grouping them into "chunks" (Hartley 1982:155). Altheide (1996:31) refers to themes as boundaries or parameters for discussing a particular event by bracketing or marking off something as one thing rather than another. Grouping of articles into themes made it easy for the researcher to analyse all articles falling under one theme together instead of analysing each article separately.

As this was a case that aroused high public interest, all four newspapers published editorials that dealt solely with the case. As noted in the literature, the use of editorials was important in the study because they convey the views of the editor and (by extension) the newspaper as a whole and express opinions and frame situations by directing readers to a selected meaning. Analysis of editorials enabled the researcher to ascertain whether the media's role of providing critical analysis was well played out. As discussed in chapter five, the main function of editorials is the "expression and persuasive

90

communication of opinions". They also make up a relevant body of text for the examination of predominant ideological assumptions in a society.

The following section discusses the themes as seen in the editorials of the four newspapers.

Theme 1: Objectivity in the media

To what extent were editors objective in their analysis of the Schabir Shaik trial?

This study noted in chapter 3 that objectivity in the media is important though contested by many. Objectivity in the media must be promoted by allowing diversity of perspective. *The Citizen* editorial took a rather arguably subjective stand in as far as the coverage of this case is concerned. It is one of the newspapers that pre-empted the outcome of the case before hearing the judge's decision. *The Citizen* editorial commented as follows: "He (Jacob Zuma) is clearly unsuitable for his current job, never mind the presidency" (*The Citizen*, 04 June 2005, p.10).

In another editorial piece *The Citizen* editorial again repeated comments about not believing that Jacob Zuma was fit to lead. In the editorial entitled "A point against Zuma" (*The Citizen*, 20 Feb 2005, p.14), the editorial stated: "With each passing day it becomes more obvious that Jacob Zuma is unsuitable to be the next president of this country no matter what the ANC Youth league says". This is a negative editorial and looks as if *The Citizen* editorial was sentencing Jacob Zuma before he was given his chance in a court of law.

With regard to this matter, the *Sowetan* editorial took a very different position in an editorial entitled "Innocent until proven guilty". This editorial began: "So many things have been said about the relationship between Deputy President Jacob Zuma and his former financial adviser Schabir Shaik who goes on to trial today". It continued that the media had tried Zuma before even giving him a chance to present his case in court – "Like all South Africans, Zuma has a right to be presumed innocent until the State can

prove beyond reasonable doubt that he was guilty of any allegations made against him. If he is found to have acted corruptly in any of the financial transactions, Zuma will have to do the right thing – quit". It is very important – as stated above – for the editorial writer to be very careful about what he/she writes because as stated in the literature, editorials represent the internal position of a newspaper (Tuchman 1978).

These findings are in line with Croteau and Hoynes (1997:108) when positing that objectivity is "an ideal that is unattainable". McNair (1998:70) concurs with this viewpoint as he writes that news, like all human observations, is not truly objective. He argues that achieving objectivity is just an ideal for journalists.

Theme 2: Media independence

Research question: To what extent was media independence evident in the editorials?

Manning (2001) argues that there is an assumption in democratic societies that a free and independent media plays an important role in maintaining the flow of ideas and information upon which groups of people make choices. The editorial on whether to broadcast the trial live or not, with the headline "To TV or not to TV? – (*The Star*, 06 Oct 2004, p. 4) does not take any particular stance on this matter. The editorial begins by providing the advantages and disadvantages of covering the case live on air. It argues that television coverage would provide an "edited version, which offers sensational highlights that could misrepresent the facts (*The Star*, 06 Oct 2004, p12).

A similar article about media independence titled "Small triumph for openness" (Sowetan - 31 May 2005, p.10) comments that the Sowetan welcomes the decision to allow live media coverage of the Schabir Shaik trial as it considers it a small triumph for openness. The editorial expresses concern that there is a great increase in media intolerance in South Africa. It is important to note that this was a significant day in as far as the Schabir Shaik trial was concerned because it was the day on which judgment was handed to the accused.

In another headline entitled "Fraud's Vaseline unglues society" (Sowetan, 09 June 2005, p.12), it is clear that the stance of the *Sowetan* is that if you commit a crime, you must be punished. In fact the *Sowetan* is concerned that it seems as if the most powerful are able to get away with serious crimes. This editorial looks into the matter from both sides and concludes that Schabir Shaik without doubt committed a serious crime that needed to be punished; thus the sentencing to 15 years was correct. In the last paragraph, the editorial asserts that as a young nation, South Africa has to be vigilant about ensuring that those committing serious crimes are punished.

Theme 3: Democracy

Research question: Are the principles of democracy adhered to?

Schudson (2003:198) points out that the press is an important instrument of democracy. Both *The Star* and *Sowetan* expressed great concern that the Schabir Shaik trial had a great impact on who became South Africa's president in 2009, seeing that Jacob Zuma's name appeared on every charge sheet of the Schabir Shaik trial. In an article titled "Our future is also on trial" *The Star* editor, writing about the impact of the Schabir Shaik trial, reminded readers that it is wrong for the media to ignore the principle that a person is presumed innocent until proven guilty. (*The Star* - 08 October 2004, p. 2). In closing, the editorial reminded readers that the trial was due to start on the following Monday and stated that the trial itself would have a great impact on who became the next president of South Africa.

The *Sowetan* editorial covering the same subject was titled "Zuma trial tests our democracy". The editorial argued that should Jacob Zuma be sent to trial and lose the case, the ANC might find itself without a presidential candidate in 2009. *Sowetan* editorial went on to say "If Zuma is found guilty, it would be a tragic end to a sterling political career of a man who sacrificed his life to a noble struggle of his people". In view of what is being foregrounded and backgrounded, the above paragraph fails to offer a fair and balanced coverage/opinion. The use of terms such as it would be "tragic" is questionable. One wonders why it would be tragic, because another person might say it

would be triumphal in that justice would be served. The editorial, therefore, offers one side of the story. The editor further goes on to praise what Zuma has done and refers to him as a "a man who has sacrificed his entire life to noble struggle of the people". This type of writing again shows that the article offers one-sided arguments and largely emphasises how unfair it would be for this country to lose such a person.

Natal Witness (04 June 2005 p.6) referred to the same issue in an editorial entitled "Credibility crossroads". The headline of the editorial denotes that the Schabir Shaik trial and its outcome have brought South Africa to what is referred to as "Credibility Crossroads". In the first few paragraphs, the editorial indicates that the trial has revealed many implications that the outcome of the trial might have on the future president, Jacob Zuma. According to the editorial, this case has also put the credibility of this country under threat because the verdict itself has shaken the independency of the judiciary system.

Towards the end of the editorial, the position of *Natal Witness* is clear that there must be thorough investigation of the arms deal and Zuma's role in it, which may be illuminated by the study of the financial relationship of Schabir and Chippy Shaik. The editor goes on to say South Africa as a country can only take the right turn at the "credibility crossroads" if Zuma is compelled to appear in court and explain his involvement in the arms deal. If all that does not happen, it will mean that "we would have taken a wrong turning at the credibility crossroads and it will be difficult to go back and get it right" (*Natal Witness*, 04 June 2005, p.6).

The Citizen editorial (18 October 2004, p.14) covering the same subject is titled "Succession looking Shaiky" (*The Citizen*,). It is interesting to note the humorous dimension in some of the editorials. For instance, in this editorial; the editor has turned Schabir Shaik's surname into an adjective (Shaiky). The editorial is about the concern that it is not clear who will succeed Thabo Mbeki. The first paragraph states that "it is no coincidence that with the Schabir Shaik corruption trial underway, there has been a lot of interest in who will succeed Thabo Mbeki (*The Citizen*, 18 October 2004, p.14). The

following paragraphs state that Jacob Zuma and Schabir Shaik had a long-standing relationship but it is wrong to assume that friendship equals crime. In view of objectivity, the editorial has tried to provide a fair and a balanced picture by emphasising that it is wrong for the media and others to conclude that Schabir Shaik and Zuma were committing a crime just because they were friends. *The Citizen* editorial continues that "the crucial question was whether Shaik tried to buy political influence and how he went about doing so". The editorial comments that whether Shaik was found guilty or not Jacob Zuma's political standing has been largely dented. In the end the editor warns that there may be a bigger game being played by people seeking to control the politics of this country (*The Citizen*, 18 Oct 2004:4).

In an article by *Natal Witness* (08 Oct 2003, p.8) entitled "Seeking the truth", the editorial's main concern is that South Africa as a country stands a great chance of losing its credibility. The editorial goes on to say that "the murky convolutions of the arms deal have turned it to a burden and handicap for the South African government". The crux of this editorial is that it everyone should be treated equally before the law and that all people must be thoroughly investigated and punished if found guilty.

When covering the case, the editors of all four newspapers generally adopted similar positions in that they all raised a concern that the future of the country was also being tried. They were also worried about the fact that some of the people in senior positions like Jacob Zuma were not being charged for their wrongdoings. Their main concern was that justice was not fairly dispensed to all as expected. *The Star* (08 Oct 2004) took a very strong position stating a concern that with the Schabir Shaik underway, it was clear that our future was also on trial. It must, however, be noted that *The Star* did not have as many editorials on Schabir Shaik as *Natal Witness* which had more editorials covering the trial than any other newspaper. This may be attributed to the fact that *Natal Witness* is based in Kwa-Zulu Natal. *Sowetan* adopted the same approach as *The Star*. It took a strong position especially in insisting that those involved in corruption must be charged in order to protect our democracy. A *Natal Witness* editorial (29 Nov 2004 p.10) noted that "...during the 10 years of democracy we've already travelled some distance down

the road to endemic corruption and that although much of this is exposed and punished, there are "sacred cows" or untouchables here".

Theme 4: Shaik judgment

To what extent did the selected print media support or oppose the Shaik judgment?

In determining whether the selected print media supported or opposed the Shaik judgment, the researcher analysed all the editorials that appeared after the judgment. It must, however, be noted that *The Star* did not have an editorial on this theme, so the researcher considered only the three other newspapers.

All three newspapers advocated that the decision by the judge needed to be respected. *Natal Witness* editorial (8 June 2005, p.8) posited that the responses of the African National Congress Youth League (ANCYL) that the judge was an apartheid judge were "immature". *Natal Witness* (08 June 2005, p.8) further stated that critics need to be made aware of the workings of the legal process ... and that Judge Squires did not work alone; he was assisted by two assessors and repeatedly used the pronoun "we". From this it is clear that *Natal Witness* was supportive of the judgment and encouraged critics to respect the independence of the South African judicial system. *Sowetan* took a similar position. We can "sympathize with Shaik's family for all the troubles they have endured but we "cannot feel sorry for the man who has been found to have used underhand tactics because he wanted to win at all costs". The closing comments of the *Sowetan* editorial were "As a young nation, South Africa has to be vigilant so as not to be tainted with the brush of corruption. The Shaik trial has been a transparent demonstration of justice" (09 June 2005, p12).

The Citizen (04 June 2005, p.10) editorial supported the Shaik judgment, insisting and adding that the "whole arms procurement process must be more thoroughly investigated and the person best suited for the position is Judge Squires". In another editorial about Shaik's judgment, *Sowetan* (07 June 2005, p.10), urged The Congress of South African Trade Unions (COSATU), African National Congress Youth League (ANCYL) and the

Young Communist League (YCL) not to insult the judiciary system. In its editorial, the *Sowetan* clearly stated that it supported the notion that Zuma should have been put on trial or should have his day in court but rejected with contempt the racist attack on the judge and found the suggestion that a black judge would have absolved Zuma insulting in the extreme.

6.1.2 Analysis of front pages

In analysing the front page stories of the four newspapers, the researcher used thematic analysis which allowed her to look into various themes/patterns as they frequently appeared. Thematic development is a kind of pattern-finding process where we identify a type of occurrence by virtue of its being perceived as an underlying "common form" found in different contexts. It is important to analyse themes emerging from reports because themes provide an indication of what was considered important by journalists when covering the news on the Schabir Shaik trial.

Thematic analysis of news discourse is significant given that the theme is the most prominent and informational part of a news text. The theme is usually expressed in the headline of the story. As a prominent feature of news texts, headlines both summarise and give the story a focus. The following section analyses themes as they appear in the front page articles of the Schabir Shaik trial in all four selected newspapers.

Theme 1: Dramatisation of the trial

| Theme | Dramatisation of the trial |
|-----------------------------|----------------------------|
| Total number of occurrences | 16 |

The study found that most of the articles about the Schabir Shaik trial were episodic/dramatic, focusing on events that are entertaining with little representation of the context. The most frequently occurring theme was dramatisation of the trial with a total of 16 occurrences.

The first article by *The Star* (12 Oct 2004, p.1), depicting the dramatisation theme was titled "The Shaik Show starts ... but the judge to decide if it will be on TV". It is interesting to note that the term "show" has been used by almost all four newspapers in their reporting. The use of this word is an attempt by the newspapers to present the news in an interesting way and this type of reporting undermines responsible reporting (Croteau & Hoynes 2003:58 & Hallin 2000:259). This story took up the major part of the paper. The first paragraph alerted the readers to the beginning of the Schabir Shaik fraud and corruption trial. It further highlighted the names of those that accompanied Mr Shaik on the first day of the trial. The style of writing reinforces that the main aim was to create excitement and spark debate.

The picture presented in the front page stories is contrary to McManus's view (1994:219) that this page should be the "newsiest" page in the newspaper". During the first week of the trial most newspapers announced the trial as if it were some kind of a show. For example, *The Star* (14 October 2004, p1) had a big headline: "The Big Showdown". This type of headline is questionable as it is written in an entertaining and dramatic rather than informative manner and is done to please advertisers. This is in line with what is said by Strelitz and Steenveld (2005:265) that newspapers pander to the lowest common denominator of public taste; they generally fail to provide information that assists citizens to make informed political judgments.

Another article reported in an episodic/dramatic fashion titled "Shaik's Mighty Fall" appeared on page 1 of *The Star* on the 9th of June 2005. The article is reminiscent of a soapie. The use of the phrase "mighty fall" implies that Schabir Shaik fell in a very big way. The article foregrounds the Schabir Shaik appeal court case. Emphasis is placed on the statement that he would be forced to give up his expensive lifestyle in exchange for that of a prisoner. What is more prominent is that half of the article does not describe the trial itself but emphasises what he would be losing should he really end up in jail as sentenced to 15 years in jail on corruption charges.

This six-paragraph article gives the readers the views of legal experts who believe that Shaik's sentence was the heaviest ever handed down for corruption by a South African court. Although the article is about Schabir Shaik, it gives only the views of Golden Miles Bhudu, a former inmate who heads the South African Prisoner organisation for human rights. This article may be viewed as one-sided and unfair to Schabir Shaik as he is not given an opportunity to give his side of the story.

Towards the end, the article elaborates the views of Golden Miles Bhudu who posits that although Schabir Shaik has been given such a heavy fine, he may still be able to live like a king because he can afford to. Mr Bhudu continues that "regardless of the rules and regulations, if Schabir Shaik ends up in prison he will be able to go out of the jail's back door when he wants to" (*The Star*, 09 June 2005, p.1).

A similar headline was used on page 1 by *Natal Witness* on the 12th of October 2004 entitled "The Shaik Show Opens". The approach is similar to that of *The Star*. The following sentence illustrates without a doubt that reporters were covering the trial as some kind of film or soapie: "At the top of the stairs at the entrance of the court buildings, Shaik, stopped, turned to the members of the media and smiled" (*Natal Witness*, 12 October 2004, p.1).

The *Sowetan* also appears to have followed the same "entertaining" approach. The page 1 headline on the 11th of October 2004 – "The Drama Begins in Parliament" – assumes that all readers are aware of the Schabir Shaik trial. During the continuation of the trial, it is clear that the same approach was being followed by all four papers.

Dramatisation is also evident in the article entitled "Ins, Outs of Schabir Shaik" (*Sowetan*, 11 Oct 2004, p.1). The article begins by foregrounding that the trial has all the elements of the soapie. The journalist, Waghiled Misbach, warns that it is easy for readers and the media to be side-tracked from the essence of what the trial is all about – corruption in the multibillion rand arms deal. The article states in the second paragraph that within the Schabir Shaik trial there are subplots and political consequences for

deputy president Jacob Zuma, who the prosecution argued, had a "corrupt relationship" with Schabir Shaik.

In closing this theme, it generally appears that the articles mentioned above are more slanted towards market driven than informative journalism. These findings are in inline which what was noted by Gillmor (2004:24) in chapter three that market driven journalism delineates the limits and the extent to which the media can be open to dissident voices and encourage genuine debate. Market driven journalism gathers an audience "not to inform it, but to sell it to advertisers" (Bagdikian 2004:247). This is not good for democracy in that the public loses an opportunity to engage in constructive debates and readers consequently become what Gillmor (2004:24) calls "a mass of shallow citizenry who can be turned into a dangerous more easily than an informed one".

Theme 2: Personalisation of Schabir Shaik

Most articles covering the Schabir Shaik trial are personality oriented rather than trial informative. Specifically, they are more geared to referring to Shaik's attributes than to the actual meaning of the case. He has been referred to both negatively and positively. In the beginning of the trial, Shaik was presented as some form of a movie star, and a hero. The most widely used term when referring to him was "Durban business man". This reference has appeared more than 56 times in the articles analysed. He was also referred to as the "wealthy Durban business man" (*The Star*, 17 Oct 2003, p.1), Zuma's close friend (*The Star*, 26 Aug 2003,p1), Zuma's financial adviser (*The Star*, 26 Aug 2003, p.1) and deputy president's financial advisor (*The Star*, 08 June 2005, p.1);

According to Thetela (2001:363), personalisation is not always complimentary. This is evident in that Shaik was frequently presented very negatively during the trial. Some newspapers like *The Star* referred to him as "The Fraudster" (08 June 2005, p.1); "Fraud and corruption accused" (*Natal Witness*, 25 Aug 2005, p1); "corrupted businessman" (*The Citizen*, 27 Aug 2003, p.1); "controversial business man" (*The Citizen*, 11 Oct 2004, p.1); "Durban's arms dealer" (*The Citizen*, 03 Feb 2004, p1) and "an exam cheat" (*Natal*

Witness, 12 Nov 2004, p1). It must be noted that Schabir Shaik was also referred to by the judge as an "an unreliable witness" (*Natal Witness*, 02 June 2005, p.1). Such words from the judge indicate the possibility that Shaik cheated his way to the top and that perhaps his dishonesty was greater that has been evident.

The media also paid a great deal of attention to the way in which Schabir Shaik was dressed and the cars he drove. For example, "Shaik arrived at court in a black BMW ..." (*Natal Witness*, 03 Nov 2004, p.1). A BMW is a symbol of wealth and status. In another article by *Natal Witness* (12 October 2004, p.1), a reference was made to the clothes that he wore – "Dressed in a dark blue striped suit and a crisp shirt". His portrayal as a movie star fits in with the trial being presented as some sort of drama.

Special focus was given to the way in which Schabir behaved during the court proceedings. *Natal Witness* reported: "Shaik turns scorn on auditors (*Natal Witness*, 25 Feb 2005, p.1). A similar headline was "Shaik's agitation intensified when he admitted that Zuma lied to Mandela" (01 March 2005). His outbursts were highlighted in various newspapers. For example, *Natal Witness* (12 March 2005, p.1) reported that he shouted at his legal team: "I'll sort you out". Again: "Shaik's agitation began when Downer inquired about payments" (*The Star*, 01 March 2005).

Towards the end of the trial, the media portrayed Shaik no longer as a hero but as a zero – and helpless. Some of the newspapers reported that "Shaik remained calm (*Natal Witness*, 01 March 2005, p.1). Another similar headline was "Shaik remained expressionless (*Natal Witness*, 09 June 2005, p1). In an article entitled "Shaik begs for mercy", he was portrayed as very helpless and despairing. In another it was reported: "... as he awaits sentencing, fraudster is defiant, remorseless and scared" (*The Star*, 08 June 2005, 1). This section has shown that the media may be responsible for creating a dramatised version of a person from a hero to zero though the way it reports.

Theme 3: Homogenisation of content

This section checks how many newspapers used the same standard, similar angle etc. During the coverage of the trial, it generally appeared that news workers work according to certain procedures. The researcher noted that there was homogenised content across newspapers and found cases of duplication within particular editions of the conglomerate- owned newspapers. An example of this is *The Star* (08 April 2005) and *Natal Witness* (08 April 2005). Both newspapers had the same story titled "End in sight for Shaik" and used the same headline, same picture and verbatim description. The information was supplied by one source: namely, SAPA.

The researcher noted that all four newspapers adopted a similar approach when announcing that the Schabir Shaik trial was starting. This was identified by the use of the word "show" across all the publications: These are some of the headlines that were published:

- The Shaik **show** opens (*Natal Witness* 12 Oct 2004, p.1)
- The big **show**down (*The Star*, 14 Oct 2004, p.1)
- The drama begins " (Sowetan, 11th October 2004, p.1)
- The Shaik **Show** starts (The Star, 12 Oct 2004, p.1).

These findings are in line which what was noted in the literature review on chapter three that sharing news content/views across publications does not promote originality but breeds homogeneity in newspapers that results in publications losing their individuality. Croteau and Hoynes (2001:153) insist that homogenisation of content can be the outcome of minimising risks and maximising profits, which results in very little innovation and a great deal of imitation. This kind of newspapering is not good for democracy and it is a characteristic of market driven journalism. As discussed in chapter 3, market driven journalism denies citizens the opportunity of being exposed to a wide range of ideas and perspectives.

The uniformity shown above begs the question: Is this journalism practice the result of sloppy editing or is it the result of insufficient copy?

The following section takes a look at the pictures that were used during the trial.

6.1.3 The analysis of pictures

Hartley (1982:181) writes that pictures play a crucial role in the construction of meaning in a story. In some instances, newspapers use similar photographs. The purpose of this section is to check to what extent this happens and also the significance of pictures used during the trial. The study found that not many pictures were used but this may be attributed to the fact that the researcher was using Sabinet (an online database). However, she managed to find at least four photographs and it became evident that *The Star* and *Natal Witness* used similar pictures. In these four photographs – all head and shoulders only – Schabir Shaik is dressed formally. One can see a tie, a shirt and jacket. The clothes on their own signify the expected dress of any business man. The formality that accompanied Schabir Shaik symbolised that he was a well-behaved man. This is of course contrary to the fact that he was facing charges of fraud and corruption.

The photo taken from *The Star* (22 Oct 2004, p.1) is a close-up picture of a smiling Schabir Shaik and it makes one wonder how a man accused of such serious misconduct could smile. It is of course probable that he was trying to maintain an appearance of confidence and unconcern. It may also be a possibility to assume that the selectors of the photographs chose the photographs for satirical reasons, to detach sympathy and to make their readers unsympathetic towards Schabir Shaik for what he was alleged to have done. In both instances, the photographs are placed in the middle of the articles, surrounded by the text. This depiction may put emphasis on how Schabir Shaik was entrapped by the arms deal but the main concern of this study is that the photo used is strikingly identical with the one used in other newspapers under analysis. These findings are in line with the literature review on chapter three that one of the reasons for content conformity in the media output is the use of one source by both print and electronic media and heavy

reliance on external news providers. McManus 2004:111) posits that as it is practically impossible for reporters to cover every story, most media houses rely largely on their reporters, correspondents and news agencies which are responsible for covering events as they happen around the world. It is assumed that facts are self-validating; people with more facts are, therefore, likely to be more accurate. Legitimate organisations are regarded as being inherently more correct because of the procedures they follow to protect the institution and people who come into contact with them. These findings are also in line with Vermeer's (1995:9) view that over-reliance on these limited news sources is a constraint on the independent functioning of the media. The following section analyses the sources that were used for front page stories.

6.1.4 Analysis of sources used in the front page articles

The hypothesis for this section is that newspapers rely more on sources than on their own reporters. News sources are very important in analysing news frames. As stated in the literature, news sources refer to people to whom journalists turn for information; they play an important role in determining which events become news and which are ignored. For a general definition, this research takes Gan's (1979:80) perspective of news sources as people "whom journalists observe or interview". In addition, one must note that these sources can provide information in conjunction with members of the newspaper as a news origination. But newspaper staffs also choose the sources they deem suitable for their readers. From Gans's perspective, it can generally be assumed that news sources are not mutually exclusive from the audience of the newspaper.

Owing to organisational pressures such as deadlines, limited budget and need for objectivity, reporters find themselves relying too heavily on some sources especially news agencies. To answer the research question, the researcher examined sources that were used during the coverage of the Schabir Shaik trial and to what extent they differed. The hypothesis is that some newspapers use similar sources. It is generally assumed that the use of different sources means diversity which refers to the spectrum of opinions that are expressed on a particular subject. Throughout the Schabir Shaik trial, it has generally

appeared that sources may be categorised into newspaper reporters, SAPA, and agencies such as Durban Bureau, legal sources, family sources, spokespersons and others.

Although journalists talk about the importance of using diverse sources to increase diversity in the media content, it is clear that most newspapers rely largely on news agencies to supply them with news. News sources for each story were analysed and their names were noted as they appeared in the articles. Second, the researcher then coded the articles using the Excel program. The coding schedule was designed to guide the classification of certain categories of analysis.

The reason for using SAPA rather that the newspaper's own staff is that it is generally known that newspapers work according to strict deadlines and some journalists find it very difficult to keep up with the demands to deliver newsworthy items on time. This is in line with McManus' argument (2004:111) when he posits that it is practically impossible for reporters to cover every story. Vermeer (1995:9) warns that over-reliance on limited news sources is a constraint on the independent functioning of the media. The use of agencies may result in homogenised media content and does not promote originality. To illustrate this point, consider the story published by both *The Star* and *Natal Witness* (08 April 2005, p.1). It is amazing to note the strikingly similar wording of the articles from these two different newspapers. It, however, became apparent that both stories were supplied by the South African Press Agency (SAPA).

In an article entitled "Blame shifted unto Shaik" (*The Star*, 27 Aug 2003, p.1), written by Jeremy Michaels and Anfre Koopman (*The Star* reporters), the leader of the Democratic Alliance, Tony Leon said: "The institution of democracy is under threat". This was supported by a statement by the leader of the United Democratic Movement, Bantu Holomisa: "Zuma must go immediately". The UDM leader is quoted more than any other source in the article. In an article about the same subject by *The Citizen* (27 August 2003, p.1), the headline announces: "More pressure on Zuma to quit". Both articles were published on the same day and both appeared on the front pages of their papers and it appears that both used Bantu Holomisa, the UDM leader, as their source. In the article by

The Citizen, written by *The Citizen* reporter, Eleanor Momberg, Bantu Holomisa has been referred to almost throughout the article.

This supports the notion that because newspapers are limited to small sources they end up sharing them with the result that the news is similar across newspapers. It must be reiterated that relying on similar sources reduces diversity. Another article entitled "Zuma and Shaik face new charges" (*The Star*, 26 Aug 2003, p.1) written by Margie Inggs, *The Star* reporter, deals with the new charges that were being added to both the Zuma and Shaik cases.

6.1.5 Analysis of metaphors

During the analysis the use of metaphors was also checked. Metaphors are important to consider in this study. There are many ways of discovering metaphors such as from propositions, modals and so on. The researcher will refer to only obvious metaphors. Gardy (1997:8) posits that "metaphor occurs when a unit of discourse is used to refer unconventionally to an object, concept, process, quality, relationship or world to which it does not conventionally refer to colligates. Metaphors are not just devices to paint language; they are used with the aim of offering significant comparisons or connections that help to enhance, clarify, challenge and even reinforce existing ideas and concepts (Carter & Goddard 1997). Checking metaphors is very important. Beard (2000) argues that metaphors are not just figures of speech but are deeply engrafted in one's construction of the world and the way one's world is constructed by others. Furthermore, Kess (1999) argues that metaphors are used to pre-empt the possibility of other alignments. Some of the metaphors that were used follow:

Metaphor 1: Arms Scandal rolls on (*Natal Witness* 11 Jan 2005, p.10)

The expression "Arms Scandal rolls on" is a metaphor in the sense that an arms scandal cannot actually "roll" in the physical sense; the expression denotes that the trial is continuing. The use of the word "roll" is another indication that this trial was perceived as some sort of an episode or drama, hence the use of the word "roll".

Metaphor 2: "Chippy" lands in hot water (*The Citizen*, 20 Nov 2001, p.1).

The phrase "hot water" denotes "trouble". Chippy is Schabir Shaik's brother who worked for the Department of Defence as the Chief of Acquisition and Procurement. He was apparently in trouble for alleged misconduct relating to the alleged passing on of classified information including Cabinet minutes of meetings where the arms purchase was discussed to his brother Schabir Shaik who had an interest in companies bidding for the arms tender. The information apparently assisted Schabir Shaik to gain an unfair advantage over other bidders.

Another metaphorical headline is "Shaik down: is Chippy next?" (*The Citizen*, 17 Nov 2001, p.1). The word "down" refers to Shaik's arrest. The first line of this article also contains a metaphor: "Shamin "Chippy" Shaik may be the next head to roll as the Scorpions' investigation into the arms deal irregularities gains momentum ..." The phrase "head to roll" has been used here to imply that he might be the next in line to be arrested. It is a reference to the nobles who literally lost their heads during the French Revolution. Similar to it is another headline entitled "Shaik's mighty fall" (*The Star*, 10 June 2005, p.1). The "fall" refers to Adam and Eve's fall from grace in the Garden of Eden. Shaik's fall is "mighty" as he was wealthy and prominent in SA's affairs and ended up in prison. Another metaphoric headline is the "Shabby Shaik" trial (*The Citizen*, 21 Oct 2004, p. 28).

Metaphor 3: Succession looking Shaiky (*The Citizen*, 18 Oct 2004, p.14).

The word "Shaiky" refers to Schabir Shaik's surname that has been used figuratively by newspapers to refer to various meanings such as "unsure", "trembling". In this instance, "Shaiky" means "uncertain, doubtful, unsure". The editorial commences with: "It is no coincidence that with the Schabir Shaik corruption trial under way, there is a flurry of interest in who will succeed Thabo Mbeki". The editorial goes on to say that whether Zuma is guilty or not, the ANC needs to be worried that Jacob Zuma's name has been dented in a serious way. A similar headline is entitled "Zuma given shaky advice". This headline – like the previous ones mentioned – is a metaphor based on Schabir Shaik's

name. Another metaphor: "in a bombshell admission, he told the court that he made payments" appeared in *The Citizen* on the 14th of October 2004 on page 1. In this metaphor, the admission is likened to a bomb explosion – surprising, unanticipated and out of the blue.

Metaphor 4: "The Shaik show starts but the judge still to decide if it will be on TV"

This headline of *The Star* informed readers that the Shaik trial had started. It likens the trial to a show, a dramatic presentation. This sort of headline entertains more than it informs. This is suggested by the use of the word "show". As noted in chapter three, the news media has increasingly become part of the entertainment industry instead of providing a forum for informed debate of key issues of public concern. (Franklin 2008:13)

The headline of the *Natal Witness* article that covered the same issues had a strikingly similar headline: "The Shaik show opens". As argued in chapter 3, content diversity is important in order to strengthen democracy. One of the reasons for such headline conformity may be pack journalism. (Schudson 2003:139) defines pack journalism as a notion whereby journalists gather and report news in a group, thereby sharing ideas and adopting similar angles (Schudson 2003:188; Shoemaker and Reese 1991: 221).

Metaphor 5: "Arms deal: Chippy shaiking in his shoes"

The headline is a metaphor based on the name of the individual (Chippy Shaik) involved and implicated in the article. This pun is based on the literal word "trembling or shivering" Chippy's surname (Shaik) is used as a verb "shaking" implying quivering or trembling as he is afraid of being charged next. Other figurative terms that were used include the following: "Succession looking Shaiky" – it simply means that there was a lot of uncertainty as to who would become the next president after President Thabo Mbeki especially because Zuma's name was linked to the arms deal corruption charges that were facing Schabir Shaik.

The use of all the above metaphors clearly confirms what was noted in the literature review on chapter three that there has been a loss of depth in stories and displacement of informative news with "softer" reporting. This point is well illustrated in Underwood and Stamm's survey (1992:307). In this study it is argued that there is a shift of interest. Newspapers are more interested in pleasing than informing readers. Hadland (2005:13) believes that both market driven journalism and commercialisation pose a threat to editorial independence because both lead to suppression of news that may upset advertisers. With similar view is Bagdikian (2000:XV) when he observes that serious journalism has been neglected in favour of trivial and banal content and, consequently, there is superficialisation of public discourse that results in degradation of news.

6.1.6 Analysis of news values/frames

News values provide researchers with how news should be covered. Watson (2007:112) argues that news values provide rules for news selection. News values are the essential elements of every good story and it is important for every reporter to check them when analysing stories. Hartley (1982:79) posits that news values give a good idea of the kind of event that will survive the selection process. The lists of values provide clues to the priority different stories will be given – a story that fulfils many of these conditions, will obviously be bigger. The purpose of this section is to outline which values were prominent during the coverage of the Schabir Shaik trial. In most of the news articles analysed, it generally appeared that in each article there was an inclusion of the following values: proximity, controversy, personal influence, frequency, continuity, the unexpected, personalisation, negativity, threshold, novelty, unpredictability, relevance and altruistic democracy.

Proximity: *Natal Witness* – originally based in the same area as Shaik – published more articles on the trial than on any other story because of its *proximity* value.

Controversy: The trial was also *controversial*. The more controversial a story, the bigger the chance for it to be published. Even limited *controversy* has relatively high news

value. The Schabir Shaik trial was controversial and was, therefore, included in most news coverage.

Personal influence: Shaik and Zuma were/are both influential citizens.

Frequency: Every week there was a story covering the trial.

Continuity: Hartley (1982:78) writes that it is important for a story to have a long life span. The case of the Schabir Shaik trial was very useful to the daily newspapers because people/readers had to be updated continuously on what was happening.

The unexpected: One can also say that the case was unexpected as it implicated two prominent citizens: Schabir Shaik and Jacob Zuma. As shown in the literature (chapter 3), people with economic means are considered to be of more news value than ordinary citizens. Gans (1979:26) notes that ordinary working citizens only receive media coverage when they are victims of accidents such as mine disasters and construction accidents. According to Hartley (1982:78) unexpected news is normally unleashed via expected contexts.

Personalisation: most articles were personality oriented rather than trial informative.

Negativity: The Schabir Shaik trial had a strong aspect of negativity. As stated in the literature, Hartley (1982:79) maintains that bad news is actually good news. Issues that are bad are likely to be given more space than purely good news. Good news tends to be boring and is consequently not given enough coverage.

Threshold: The threshold of the Schabir Shaik trial was also right. As explained in chapter three "threshold" refers to the size of the event. The bigger the story, the more added drama is needed to keep it going. During the Schabir Shaik trial, there were many unfolding small trials. *Natal Witness* (25 Feb 2005:1) reported as follows: Shaik has following – out with his legal team He also had a heated exchange of words with his legal team which became a headline in its owned entitled "I will sort you out" (*Natal Witness*, 12 March 2005, p.1).

Novelty: New, unusual or strange events attract readers and sell newspapers. This was a big story which resulted from the combined efforts of police and journalistic investigation and so it had the added value of *novelty*.

The unpredictability and rarity of an event are also important in news reports.

Relevance: The case was relevant because it concerned South African matters well known to readers. "Relevancy" refers to the local importance of the news reports.

Altruistic democracy: As noted in chapter three, the altruistic democracy value assumes that news media should publish stories that are largely based on public interest and public service. The main reason why the case had so much interest from the public is that both Shaik and Zuma are prominent people and as noted in chapter three, journalists tend to cover stories in a way that always differentiates winners and losers – they tend to have little patience with losers (Gans 1979:43).

6.2 Conclusion

Using quantitative and qualitative content analysis, the study has revealed that newspapers follow market driven journalism, writing stories in an entertaining way. It appears that the Schabir Shaik trial was dramatised by all four selected newspapers. The number of occurrences of dramatisation indicates that for newspapers this case was not about giving critical analysis to the readers but it was more about what sells.

The analysis of front page stories indicates that sharing of news by newspapers breeds homogeneity. The study has also shown that news agencies tend to be predominantly selected for stories and this is not in accordance with good journalism. Good journalists are expected to use competing sources to show both sides of a story. In fact it seems that newspapers take similar approaches when covering the news.

In the analysed editorials, it generally appears that some newspapers did not remain objective in the coverage of the Schabir Shaik trial. This is not bad because it is in line with Croteau and Hoynes (1997:108) when positing that objectivity is "an ideal that is unattainable". The *Citizen*, for example, took a subjective stand and even pre-empted Schabir Shaik's guilt, implying that Jacob Zuma was also guilty. The researcher feels that it was wrong for a newspaper to take that particular stand because Jacob Zuma has not been charged. A newspaper's role is to provide information and assist readers by providing critical analysis. Most newspapers in the study failed to offer a fair and

balanced analysis of the trial. It was argued in chapter three that responsible journalism calls for media that can provide the public with critical analysis thereby enabling them (the public) to make informed choices.

One can also conclude that the Schabir Shaik trial exploited a number of additional news values not mentioned in the list above such as reference to elite persons and institutions, human interest and tragedy, consequence, impact, national importance, scandal and proximity. The subsequent chapter concludes as well as offers some suggestions for further research.

CHAPTER 7 CONCLUSIONS

7.0 INTRODUCTION

This study set out to analyse conformity in the selected print media as a threat to democracy. I tried to establish what causes newspapers to cover stories in a similar fashion across publications. This analysis was done by looking at the way in which the Schabir Shaik trial was covered by the four selected daily newspapers. Themes were selected from the front pages. Editorials were also analysed where it appeared that some newspapers did not remain objective when covering the stories. The analysis of front page stories indicated that the sharing of news and reliance on similar sources across publications breeds homogeneity.

This chapter sums up the key issues that arose during the study of conformity in the selected print media as a threat to democracy. Conclusions based on the findings of the investigation, as well as the literature review, are presented. In this chapter, the aim and theories used are re-examined to see if the research has achieved the objectives presented at the beginning of the study. Possible research projects that could emanate from this research will also be suggested.

7.1 Conclusions on Literature Review

Chapter two provided discussion on the role and functioning of the media. The emphasis was on the role of the media in democracy. The concept of democracy and that of a particular medium were explored. It was stated that in the context of democracy the media is expected to play a watchdog role by exposing corrupt officials from both government and the private sector who are guilty of misusing power conferred on them by their portfolios. Following on from this, the chapter highlighted that the media is also expected to interpret and analyse information for citizens. In a democracy ordinary citizens are, according to Waldhal (2004:13), not in a position to understand and interpret

what is happening in a political sphere. It is sometimes necessary for journalists to interpret complicated matters on a level that ordinary citizens may understand and use. The researcher was mainly interested in the question: To what extent are democratic principles put to work in the four newspapers?

The media are often condemned by various authors for failing to fulfil these roles. According to liberal theory, meaningful political discourse can only take place if the public is provided with a diverse range of information and this is apparently not happening. To a large extent the media often neglect the views, struggles and aspirations of the poor and marginalised.

Chapter two also outlined the Habermasian conception that the public sphere is still essential in understanding the role that must be played by the media in a democracy. This concept has been criticised as being elitist and idealised. Also that it overestimates the degeneration of rational critical debate in mass communication. Despite this criticism, the concept is important in understanding the link between communication and politics; even more so, the problem of media regulation, ownership and government interference.

The section on analysis of African media as it relates to the concept of public sphere in chapter two is also important. Through this section, it emerged that the media in Africa is faced with a problem of the decline of the public sphere; journalists in Africa also fail to provide alternative voices when reporting news. From this point, it can generally be argued that the media is still a force that continues to shape, define and establish democracy in all African countries. In spite of this significant role, there has been very little that the media has done under the new democratic dispensation sweeping the continent. The following section provides some critical reflections about chapter three.

Chapter three outlined the theoretical perspectives underlying the production of news and the qualities which are used as guiding principles to judge stories as meriting inclusion or omission in the newspapers as news. These theories posit that the media is not a neutral purveyor of information – it pre-selects and constructs information for specific audiences.

The chapter also highlighted the significance of news values and their placement in front page stories. It further focused on the operation of the media in the processes of producing and disseminating news.

Through chapter three, it emerged that news production is a process that involves many facets that ultimately affect the output. Despite the claims of many newspapers, the chapter highlighted that news does not mirror reality – it is largely shaped by frames that seek to reflect part of reality. It showed that news values – a significant consideration in the process of selection – are criteria used in newsroom practice to select from the several contending items those few that are potentially news. Having established this factor, the study went on to examine the kind of news sources that are preferred for the front pages.

It emerged during chapter three that news sources are very important in analysing news frames. The use of sources is closely tied to "balanced" reporting and, as demonstrated in the previous chapters, all four newspapers in the study failed to provide a balanced reportage of the Schabir Shaik corruption trial. Balanced reporting demands that opposing sources be allocated a space for their views in a news story. Throughout the Schabir Shaik trial, it appeared that sources could be categorised into newspaper reporters, SAPA and other agencies such as Durban Bureau, legal sources, family sources and spokespersons. As stated in the literature, "news sources" refers to people to whom journalists turn for information. These sources play an important role in determining which events are newsworthy and which are ignored. The research question was for the researcher to check to what extent the use of sources during the coverage of the Schabir Shaik trial differed.

The hypothesis that the some newspapers use similar sources was proven correct. Although journalists always talk about the importance of using varied sources to increase diversity in the media content, it was clear that most of the four newspapers studied relied largely on news agencies to supply them with news. It was amazing to discover the strikingly similar wording of the articles from different newspapers. It is apparent that the wider the spectrum of sources, the wider the range of opinions expressed. These findings

are in line with the literature review on chapter three that another reason for content conformity in the media output is the use of one source by both print and electronic media and heavy reliance on external news providers. McManus 2004:111) posits that as it is practically impossible for reporters to cover every story, most media houses rely largely on their reporters, correspondents and news agencies which are responsible for covering events as they happen around the world. It is assumed that facts are self-validating; people with more facts are, therefore, likely to be more accurate. Legitimate organisations are regarded as being inherently more correct because of the procedures they follow to protect the institution and people who come into contact with them.

These findings are also in line with Vermeer's (1995:9) view that over-reliance on these limited news sources is a constraint on the independent functioning of the media.

It also appears that most articles analysed were slanted towards market driven journalism. This was in line with Gillmor (2004:24) when he posits that market driven journalism delineates the limits and the extent to which the media can be open to dissident voices and encourage genuine debate. Through this chapter, it was noted that journalists were writing news to sell, not inform. Such behaviour was noted in the front page articles by the constant use of the word "show" when referring to the trial. This is not good for democracy in that the public loses an opportunity to engage in constructive debate and readers consequently become what Gillmor (2004:24) calls "a mass of shallow citizenry" instead of being an informed one". Bagdikian (2004:247) concurs with Gillmor when he claims that this kind of journalism gathers an audience "not to inform it, but to sell it to advertisers".

It became apparent in this chapter that journalists face various pressures, structures and norms and the mentioned factors have a large influence on the news content. It also emerged that journalists prefer elite sources compared to ordinary sources and that preference for certain sources is an outcome of processes and routines followed during production. In their endeavour to beat deadlines, news workers follow routines when performing their jobs. The next section discusses the conclusions based on chapter four.

Chapter four noted that the South African print media is characterised by different newspaper firms that are mainly conglomerates. It also discussed the history of South African print media and noted that a lot has been written on South African print media as compared to broadcasting. The chapter further revealed that before democracy, the media was focused on fighting apartheid. The end of apartheid meant that most newspapers including the four under investigation had to re-define their role in line with democratic principles. Although elections on 27th April 1994 marked the official end of the apartheid system and the beginning of democracy, the initial years of democracy were largely marked by a mutual mistrust between the media and government with the latter accusing the media of reluctance to embrace the government of the day.

What was of great importance to this study was the role that *The Citizen, Sowetan, The Star* and *Natal Witness* are playing in our democracy. As demonstrated in chapter four, these four newspapers aim at different target publics. *The Citizen* is published six days a week and distributed mainly in Gauteng. The newspaper is co-owned by Johnnic Communications and CTP/Caxton and is aimed at white readers. *The Star* is published in Johannesburg and distributed throughout South Africa, with most sales in Gauteng. Once aimed exclusively at the white market, today over 50% of the *Star*'s readers are black, *Sowetan* is mainly aimed at black readers.

The difference in these newspapers was also highlighted when analysing their editorials. *Sowetan* is a mild critic of the ANC government and its officials and insisted on stating facts on the principle of innocence until proven guilty while *The Citizen* was quick to judge and even took a strong stand that anyone implicated in corruption is not fit to be the president of the country. The *Star* on the other hand, managed to inform and maintain some form of neutrality. *Natal Witness* was surprisingly harsh in its reporting considering that Schabir Shaik is from Durban and a more positive approach might have been expected.

The next section looks at whether the methodology chosen managed to provide some direction in terms of answering the research questions posed at the beginning of the study.

7.2 Conclusions on Methodology

This study employed content analysis as the method of data collection, and one can say this was indeed a viable choice for this particular study because the method was able to assist the researcher in addressing research questions posed at the beginning of the study. As content analysis is a method that is very systematic and less prone to subjective procedures, it can be concluded that the results of this study not only depict high standards of reliability but are also valid.

The researcher can confirm, after using this method, that it did provide many advantages. It was indeed inexpensive because available text was obtainable from Sabinet, an online database. Second, it was easy to obtain the data itself through the use of key words and following the simple directions indicated in the database. Third, the method was indeed unobtrusive and did not interfere with people's lives, which means ethical problems relating to dealing with human beings were avoided.

This research method provided a way of obtaining data to measure the frequency and extent and meaning of messages. A progression of category construction: sampling; data collection; data analysis; data-coding; interpretation and quantitative content analysis was used to determine the objective content of messages of pre-defined categories.

As discussed earlier, this study employed thematic content analysis in the classification of articles and editorials into categories. Through the use of thematic analysis, the study also managed to analyse front page stories and editorials retrieved from the Sabinet database. Not many editorials were obtained but the number obtained was sufficient for analysis.

The next section provides detailed critical reflections about the key findings.

7.3 Conclusions on Findings

The analysis of front page stories indicated that sharing of news/relying on similar sources across publications breeds homogeneity. Croteau and Hoynes (2001:153) insist that homogenisation of content can be the outcome of media minimising risks and maximising profits, which results in very little innovation and a great deal of imitation. This kind of news reporting is a characteristic of market driven journalism which denies citizens the opportunity of being exposed to a wide range of ideas and perspectives.

The analysed editorials indicated that newspapers were not analysing the Schabir Shaik trial in a fair and balanced manner. Some editorials were taking sides in their analysis. According to Croteau and Hoynes (2001:6) the media, watchdogs of freedom, have a special task of providing independent information to citizens especially in editorials because they represent the internal position of the newspaper.

An editorial conveys the views of the editor and (by extension) the newspaper as a whole. In this the researcher tried to ascertain whether the media fulfilled its role of providing critical analysis. It must be stated that some editors took a subjective stand when reporting the Schabir Shaik trial. *The Citizen*, for example, took a subjective stand and even pre-empted the outcome of the trial before the decision of the judge could be heard (*The Citizen*, 04 June 2005:1). It generally appeared in the editorials that were studied that some failed to provide a fair and balanced analysis of the trial; this is contrary to the editorial code of ethics.

The researcher further established that news values are a significant consideration in the process of selecting news. Having established this factor, the study also went on to check what news sources were used by the four selected newspapers when writing their front page stories. The findings in both quantitative and qualitative studies (see chapter five) show that journalists use similar sources and other voices (sources) are not afforded an opportunity to be heard, especially those of females.

One can conclude that although the way the Schabir Shaik trial was reported created interest in many people and contained many news values, it also exploited a number of news values like personalisation, negativity, reference to elite persons and institutions, drama, human interest, tragedy and so forth. Other news values that could be identified included consequence, impact, national importance, scandal because it was exposing the secrets of highly placed individuals in society (Schabir Shaik and Jacob Zuma), elite personalities, interest and proximity. This was a big story which resulted from a combined effort of police and journalistic investigation and so it had the added value of novelty.

The studied articles revealed that sources used during the Schabir Shaik trial were not representative of all civil society. The findings in chapter 6 show that there was a preference for selecting elite sources. The voices of the people at grassroots level in the front pages were scant. None of the newspapers dared to obtain the views of civil society. Most of the sources selected were largely influenced by the type of trial that was being covered. Attention was given to this particular issue of under-representation of grassroots sources in this study because of the social and demographic significance of their contribution to society and the assumed importance of media in democracy to which *The Citizen, The Star, Sowetan and Natal Witness* subscribes and that can also be witnessed in their editorial policies.

Contrary to what is expected from the media, journalists and editors are not motivated to promote diverse views in the news and this compromises diversity in the media. It generally appeared in the analysed data that the lesser the number of sources used, the lesser the number of opinions that were being portrayed. One can, therefore, conclude that the use of sources was not satisfactory and balanced.

It has been argued by various authors that responsible journalism calls for media that can provide the public with diverse items of information thereby enabling them (the public) to make informed choices. As argued by Curran (see chapters 1 and 2), political discourse

can only take place if the public is provided with a wide range of information (Curran 2002:217). The researcher noted that in some inside pages there were instances where reporters quoted ordinary citizens. It was beyond the scope of this study to examine the types of sources that were chosen in those pages. As stated in chapter 5 as well as at the beginning of the study, the researcher only focused on the front pages and editorials.

After having established that there are imbalances in terms of selecting sources for front page stories, it is clear that such actions have negative implications in terms of the newspapers in the study. First, it may mean to the readers that those groups chosen are irrelevant and relatively unimportant in terms of offering valuable information in a democratic society. Second, it may appear that *The Citizen, The Star, Sowetan and Natal Witness* are promoting the views of a certain class of people with political power, while still claiming to represent different perspectives in news reporting. Third, the four newspapers, by narrowing news sources to such a small sector of society, do not give a rich picture of social reality which is made up of diverse groups of people that represent varying views and interests. Within a democratic paradigm of news reporting, the news media ought to present a true picture of society in terms of selecting news topics, writing news topics, writing news and analysis of front page stories. That means widening the scope of news and source selection, reporting in a balanced fashion and analysing issues critically.

Chapter six also provided evidence that qualified claims and criticism discussed in chapters two and three of the literature review. Findings show that the media is not fulfilling its role as outlined in chapter two. The conclusion is that the selected newspapers did not cover the case as critically as expected from an open democracy. They failed to offer a significant forum for public debate and engagement even in a trial as big as this was. Habermas's concept of public sphere allows citizens to interact, study and debate on public issues. To re-institute the public sphere notion, the media will have to encourage public participation so that cases of such public interest can be covered in a more balanced manner and be open to the public for discussion and reflection.

The next section provides the limitations of the study.

7.4 Limitations of the Study

As the study focused on only the four selected newspapers, one cannot generalise the results to all the South African newspapers which were not represented.

The use of front pages was good in terms of controlling the study but it limited the researcher because some articles which would have provided an interesting analysis appeared on inside pages of the newspapers. The study focused on print media only; it would have been interesting to check how radio stations covered the stories about this case.

Another limitation is that the researcher does not know how to use the SPSS program. These days most researchers use sophisticated statistical software packages called Statistical Programme for Social Sciences (SPSS) for analysing data. The researcher could not use this package because she is not familiar with it.

The researcher was also not able to obtain the particular photograph used by the newspapers during the trial but managed to obtain four others. It would have been very interesting to conduct a deeper analysis of the photographs used to discover whether the newspapers all used the same photo/s provided by news agencies so that deadlines could be met.

Finally, it would have added some value to conduct one or two interviews. Potton (1990) and Lindlof (1995) posit that qualitative interviews allow the researcher to enter other people's worlds. It is, however, important to note that the results and findings of this study are still of importance in spite of these shortcomings. Some of these challenges may themselves be answered in future research on this subject. The following section covers further research that may be conducted.

7.5 On Further Research

A comparative study on the same topic would be very interesting to explore. As mentioned in the literature, one of the major drawbacks is that objectivity of the results is in most cases contested because results depend on the interpretation of the investigator. Should the investigator be dishonest, the results will not be honest. A study is needed that will include other methodology such as critical discourse analysis. This would help in obviating the weaknesses of content analysis.

While this study obtained good insights into the significance of selecting balanced news sources, the findings are only important in as far as they are related to front pages. It should, therefore, not be assumed that a variety of news sources is also missing on the inside pages of the newspapers. A study that looks at the entire content of the newspapers would add an interesting dimension.

The research project only investigated conformity in the print media. The same study of electronic media would also provide an interesting dimension. Furthermore, the research investigated conformity in the print media by using content analysis only. Further study could expand this to include interviews and observations conducted within the newsroom. Finally, a comparative study sampling a number of media houses might bring interesting insights into the general outlook on media practice.

This chapter has highlighted issues for further research. As stated in the beginning the aim of this study was not to accuse the media but was to take a closer look at how some of the newspapers handled a significant trial. It is hoped that the insights of this study will be valuable elsewhere in terms of understanding how the media functions.

BIBLIOGRAPHY

Adhikari, M. 2000. *You Have a Right to know: South* 1987-1994, L and Adhikari, M 2000. <u>South Africa's Resistance Press: Alternative Voices in the last Decade under Apartheid</u>. Ohio: Ohio University Press. Chapter 8.

Altheide, D.1996. *Qualitative media analysis research methods*. Series 38, Thousand Oaks: Sage Publications.

Ansah, P. 1998. *In search for a role for the African media in the democratic process*. Africa Media Review. 2(2): pp.1 - 6. Nairobi: ACCE.

Bagdikian, B. 1983. The media monopoly. Boston: Beacon Press Books.

Bagdikian, B. 2000. The media monopoly. Boston: Beacon Press Books.

Bagdikian, B. 2004. *The media monopoly*. Boston: Beacon Press Books.

Baker, A. 2002. *Media, markets and democracy*. Cambridge: Cambridge University Press.

Bennett, T. 1982. *Theories of the media. Theories of society and the media.* London: Routledge.

Bennet, W.L. 1996. What is news? The construction of political reality. <u>In</u> Bennett, W.L. *News. The politics of illusion*. Routledge: London.

Berger, G. 2005. Current challenges. In changing the fourth estate essays on South African journalism. Edited by Adrian Hadland. South Africa: HSRC Press.

Berger, G. 1993. *What criteria for measuring diversity?* Retrieved March 12, 2005 from the World Wide Web: http://www.journ.ru.ac.za?staff/guy/index.html

Berger, G. (1998). Media and democracy in Southern Africa. <u>In</u> *Review of South African Political Economy*, No 78: pp. 599 - 610.

Berger, G. 2000. *The media, research and racial representation in post-apartheid South Africa*. African Studies Association conference, Nashville

Berger, G. 2001. *Problematising race for journalists: Critical reflections on the HARC*. Inquiry into Media Racism Rhodes University, South Africa, October 2001. Accessed http://journ.ru.ac.za/staff/guy/Research/racism.

Berger, G. 2002. De-racialisation, democracy and development: Transformation of the South African media 1994 – 2000. <u>In</u> Tomasselli K. and Dunn H. (eds). *Media*,

democracy and renewal in Southern Africa: International Academic Publishers: Colorado.

Berger, G. 2005. They look like newspapers. Sunday Times 1 May 2005 p.3.

Berger, G. & Kanyegerire, A. 2002. *A media content, advertising and circulation*. Retrieved March 12, 2004 from the World Wide Web: http://www.journ.ru.ac.za/staff/guy.fulltext.circ.doc

Bertrand, I. & Hughes, P. 2005. *Media research methods: Audiences, institutions, texts.* New York: Palgrave Macmillan.

Blumer, J.G. 1998. The crisis of public communication. London: Routledge.

Boyce, G. Curran, J. Wingate, P.1978. *Newspaper history: from the 17th century to the present day.* London: Constable.

Calhoun. C. 1992. Habermas and public sphere. Cambridge, Massachusetts: MIT Press

Cohen, B.J. (ed). 2006. International political economy. Aldershot: Ashgate.

Common, M.1995. Sustainability and policy. Cambridge: Cambridge University Press.

Compaine, B.M. and Gomery, D. 2000. Who owns the media? Competition and concentration in the mass media industry. Mahwah, New Jersey: Lawrence Erlbaum Associates.

Corden, W.M. 1952. The Maximisation profit by a newspaper <u>In</u> *Review of Economic Studies*, 20, pp.181-190.

Creswell, J. 2003. *Research design: Qualitative, quantitative and mixed method approaches.* 2nd ed. Thousand Oaks: Sage Publications.

Croteau, D. & Hoynes, W. 1997. *Social inequality and media representation* In *Media/Society: Industries, images and audiences*. London: Pine Forge Press.

Croteau & Hoynes 2000. *The business of media: Corporate media the public interest*. Thousand Oaks: Pine Press.

Croteau, D. & Hoynes W. 2003. *Media Society: Industries, images and audiences*. 3rd ed. London: Sage Publications.

Croteau, D. & Hoynes, W. 2006. *The Business of media: Corporate media and the public interest.* 2nd ed. London: SAGE.

Croteau, D. & Hoynes, W. 2007. *Media society: Industries, images and audiences*, 2nd

edition. Thousand Oaks: Pine Forge Press.

Curran, J. 1990. *The new revisionism in mass communication research: a reappraisal* European journal of communication 5(23), pp.135 – 64.

Curran, J. 1991. Mass media and democracy: A reappraisal. <u>In</u> Curran, J. & Gurevitch, M. 2000. *Mass media and society*. London: Arnold.

Curran, J., Gurevitch, M. 1991. Mass media and society. London: Arnold.

Curran, J. & Seaton, J. 1991. *Power without responsibility: The press and broadcasting in Britain*. 4th ed. London: Routledge.

Curran, J. 2000, Rethinking media and democracy <u>In</u> Curran and Gurevitch (eds) *Mass media and society*. 3rd ed. London: Arnold.

Curran, J. 2000. Rethinking media and democracy <u>In</u> Curran, J. & Gurevitch, M. 2000. *Mass media and society*. London: Arnold.

Curran, J., Gurevitch, M. 2000. Mass media and society. 3rd ed. London: Arnold.

Curran, J. 2002. Media and power. London: Routledge.

Dahl, R. 1971. Polyarchy. London: Yale University Press.

Dahl, R. 1989. Democracy and its critics. London: Yale University Press.

Dahl, R.A. 2005. What political institutions does large-scale democracy require? Political Science Quarterly Volume 120 Number 2, pp.187 - 197.

Dennis, E.E. & Merril, J.C. 1991. *Media debates issues in mass communication*. London: Longman.

Djankov, S., McLeish, C., Nenova, T. & Shleifer, A. 2001. Who owns the media? Draft paper for the World Bank Development, Report.

Doyle, G. 2002. Media ownership. London: Sage Publications Limited.

Du Plooy, G.M. 2001. Communication research: techniques, methods and applications. Kenwyn: Juta.

Du Plooy, GM. 2005. Communication research: techniques, methods and applications. Kenwyn: Juta.

Elliot, D. 1982. Responsible journalism. Beverly Hills: SAGE.

Entman, R.E. 1993. *Framing: Toward clarification of a fractured paradigm.* Journal of Communication, 43 (4), pp.51 - 58.

Fairclough, N, 1989. Language and power. London: Longman.

Fairclough, N, 1992. Discourse and social change. Oxford: Polity.

Fourie, JP. 2001. *Media studies, volume one. Institutions, theories and issues.* South Africa: Juta

Fourie, JP. 2007. *Media studies, volume one. Institutions, theories and issues.* South Africa: Juta.

Franklin 2008. Newszak and news media. London: Arnold.

Fraser, N, 1990. Rethinking the public sphere: A contribution to the critique of actually existing democracy. <u>In</u> C. Calhoun (Ed) *Habermas and the public sphere*. Boston: MT Press.

Gans, H.J. 1980. Deciding what is news. London: Constable.

Garnham, N. 1992. *The media and the public sphere*. <u>In</u> Coulhoun, C. (ed) Calhoun, C. 1992. *Habermas and the public sphere*. Boston: MT Press.

Gassaway, B.M. 1984. The social construction of journalistic reality. University of Missourie: Columbia

Gillmor, D. (2004). We the media. Sebastopol: O'Reilly Media Inc.

Gillwald, A.1993. *The public sphere, the media and democracy*. Transformation (21) pp. 65 -77.

Goffman, E. 1974. Frames analysis. New York: Harper and Row.

Goffman, E. 1974. *Frame analysis: An essay on organisation of experience*. Philadelphia: University of Pennsylvania Press.

Goatly, A. 1997. The language of metaphors. Metaphors: London

Golding, P. & Murdock, G. (eds). 2000. *The political economy of the media* vol.II. Cheltenham: Edward Elgar Publishing Limited.

Graber, D. 2003. *The media and democracy: Beyond myths and stereotypes. Political Science* pp. 139 -160.

Grbesa, M. 2004. Why if all is the public sphere a useful concept? Politicka Misao, vol. XL (5), pp.110-121,

Gunter, B. 2000. *Media research methods: Measuring audiences, reactions and impact.* London: SAGE.

Habermas, J. 1989. *The Transformation of the public sphere*, trans Thomas Berger, with Frederick Lawrence. Cambridge: MIT Press.

Habermas, J. 1992. Further reflections on the public sphere, trans. Thomas Burger in Calhoun, C. (ed)

Hadland, A. & Thorne, K. (eds). 2003. The people's voice: The development and current state of the South African small media sector. Cape Town: HSRC Press.

Hallin, DC, 1994. Covering the Body: The Kennedy Assassination, the Media and the Shaping of the Collective Memory, American Journal of Sociology 100 (2) p549-550.

Hallin, DC. 2000, *Commercialism and Professionalism in the American News Media* In Curran, J. & Gurevitch, M. 2000. Mass media and society. London: Arnold.

Harber, A. 2004. Reflections on journalism in the transition to democracy: Report from South Africa.

Hartley, J.H. 1982. *Understanding news*. London: Routledge.

Herman, E.S. & Chomsky, N. 1994. *Manufacturing consent: The political economy of mass media*. New York: Routledge.

Horwitz, R. 2001. Communication and democratic reform in South Africa. Cambridge: Cambridge University

Jackson, G.S. 1993. Breaking story: The South Africanp. Oxford: Westview Press.

Jacobs, S. 2002. How good is the South African media for democracy? Mapping the South African public sphere after apartheid, World Press Freedom Review 279 - 302: A Special Report. Vienna: International Press.

Jacobs, S. 2004. Media during South Africa's first decade of liberal democracy: Some short impressions, Ecquid Novi vol. 25 (2) pp. 346 -350.

Jensen, K.B. 1987. *News as ideology: Economic statistics and political ritual in television network news.* Journalism of Communication 37:27.

Jensen, K. B. 2002. The qualitative research process In A handbook of media and

communication research: Qualitative and quantitative methodologies. Jensen, K.B. (ed). London and New York: Routledge.

Kareithi, P. 2005. Rethinking the African press: Journalism and the democratic process <u>In</u> Kareithi, P. & Kariithi, N. (eds) *Untold stories: Economic and business reporting in African Media*. Johannesburg: Wits University Press.

Kasoma, F.1995. *The role of the independent media in Africa's change to democracy*. Media, Culture & Society, vol. 17, 5.pp.37 - 555. London: Sage.

Kasoma, F. 1997. *The press and multi-party politics in Africa*. Finland: University of Tampere.

Keane, J. 1991. Media and democracy. Cambridge: Polity Press.

Keane, J. 1993. Media and democracy. Cambridge: Polity Press.

Kisuke, C.S. 2004. An investigation of the role of news values in the selection of news sources in a contemporary third world newspaper: A case study of Daily Nation Newspapers. Thesis. Johannesburg: Rhodes University, department of Journalism and Media Studies.

Kobre, S. 1981. *Reporting news in depth.* Washington: University Press of America: Washington.

Koppel, T.1983. ABC news viewpoint, ABC-TV – Oct -14.

Kress, G.1990. *Linguistic processes in socio-cultural practice*. Manchester: Oxford University Press.

Krippendorff, K, 2004. *Content analysis: An introduction to its methodology*. 2nd ed. London: Sage.

Kupe, T. 2005. The role of independent media in democratic African societies. Challenges and opportunities for the African media In 2001, Globalisation and East African media, pp. 1 - 3.

Kuypers, J. 2002. Press bias and politics. London: Arnold.

Legun, C. 1971. *The mass media – Institutions of the African political systems*. Sweden: Uppsala.

Litchenberg, J. 2000. *In defence of objectivity revisited* In Curran, J. & Gurevitch, M. 2000. *Mass media and society*. London: Arnold.

Lippmann, W. 2000. Newspapers <u>In</u> Grabber, D. (ed) *Media power in politics*.

Washington DC: CQ Press.

Lubbe, H. J. 2007. Comparative analysis of the reportage of the Gugulethu 7 events in the South African press in 1986 and 1997. 21 (1), pp.152 -165.

Lupton, D. 1994. Moral threats and dangerous desires: Aids in the news media in the Australian press in the mid 1990s <u>In</u> *Journal of Sociology of Health and illness*, 21 (1) pp. 37 -53.

Mak' Ochieng, M. 1993. The effect of a changed political culture on press performance: The case of Kenya during the change to multi-party politics and polit thesis. University of Bergen, Department of Mass Communication and Media Studies.

Mandela, N.1994. Address to the International Press Institute, 14 February.

Manning, P. 2001. News and news sources: A critical introduction. London: SAGE.

McChesney, R. 1998. Off limits: An inquiry into the lack of debate over the ownership structure and control of the mass media in US political life. Journal of Mass Communication, 13, pp. 1-9.

McChesney, R. 2000. Rich media, poor, democracy: Communication politics in dubious times. New York: The New Press.

Mckee, A. 2005. *The public sphere: An introduction*. Australia: Cambridge University Press.

McManus, J.H. 1994. Market driven journalism: Let the citizen beware. London: SAGE.

McNair, B. 1995. An introduction to political communication. Routledge: New York.

McNair, B. 1998. The sociology of journalism. London: Arnold.

McNair, B. 2000. *Journalism and democracy: a millennial audit*. Scotland:University of Stirling.

McQuail. D. 2005. Mass communication theory. London: Sage Publications.

McQuail, D. 2005. Theory of media and theory of society. <u>In</u> D. McQuail (ed.) *Mass communication theory: An introduction* (3rd ed.) London: Sage.

Megwa, E. R. 2001. Democracy without citizens: The challenge for South African journalism education. Journalism Studies, vol. 2, number 2, pp. 281 - 299.

Merrit, D. 1995. Public journalism – Defining a democratic art, <u>In Media Studies Journal</u> 9 (3).

Mouton, J.1996. *Understanding social research*. Pretoria: Van Schaik Publishers.

Mouton, J. 2001. *How to succeed in your masters and doctoral studies*. Pretoria: Van Schaik Publishers.

Mukarovsky, J. 1970. Linguistics and literary style. New York: Ann Artor.

Murdock, G. 1982. Large corporations and the control of the communications industries. <u>In</u> Gurevitch, M., Bennet, T., Curran, J., Woolacott, J (eds). *Culture, society and the media*. New York: Methuen.

Murdock, G. 2000. *Reconstructing the ruined tower: Contemporary communications* Questions of Class in J. Curran, Gurevitch, M. 2000. *Mass Media and Society*. London: Arnold.

Mwesige, P.G. 2004. Disseminators, advocates and watchdogs: A profile of Ugandan journalists in the new millennium: Journalism vol. 5 (1) pp. 69 – 96.

Napoli, P.M. 2006. *Media diversity and localism: Meaning and metrics*. London: Lwarence Erlbaum Associates Publishers.

Negrine, R. 1989. *Politics and the mass media in Britain* (2nd ed.) London: Routledge Press.

Neuendorf, K. 2002. The content analysis handbook. London: Sage.

Nyamnjoh, F.B. 2005. *Africa's media: Democracy & politics of belonging*. Pretoria: Unisa Press.

Occiti, J. 1999. Media and democracy. <u>In</u> *Africa. Mutual political bedfellows or implacable arch-foes?* Weatherhead Center for International Affairs: Harvard University.

Oosthuizen, L. M. 2001. A critical assessment of news. <u>In Media studies</u>, vol. 1: *Institutions, theories and issues*, edited by P.J. Fourie. Lansdowne: Juta Education.

Ornebring, H. & Jonson, A. M. 2003. *Television the public sphere? Forty years of current affairs programmes on Swedish television*, European Journal of Communication, (18) 4 pp. 501 - 527.

Ornerbring, H. & Jonsson, A.M. 2004. *Tabloid journalism and the public sphere: A historical perspective on tabloid journalism*. Journalism Studies, (3): pp.283 - 295.

Parenti, M 1993. *Methods of Media Manipulation*. In 20 years of censored news by Jensen, C 1997, Seven Stories Press: New York.

Picard, R.G. 2004. *Commercialism and newspaper quality:* Newspaper Research Journal, 25(1), pp. 4 - 65. http://www.poynter.Org/resource/63500/picard.pdf. Accessed: 2009-01-16.

Pillay, D. 2004. *Media diversity and the contested character of the post-apartheid state*, Politikon (November) 31 (2), pp.167 -184.

Radebe, J. L. 2006. The role of the media in transition to democracy. An analysis of the coverage of the alleged arms deal corruption by Sowetan and the Mail and Guardian, Thesis. Johannesburg: University of Witwatersrand, department of Journalism and Media Studies.

Ramaprasad, J. 2001. *The private and government sides of Tanzanian journalists*. Politics 8 (1) pp.8 - 26.

Ramsden, G.P. 1996. Media coverage of issues and candidates: What balance is appropriate in a democracy, political sciences Quarterly, vol. 111. No 1. p.81.

Roberts, G. 2001. Leaving the readers behind <u>In</u> Kunkel, T. and Layton, C. (eds) *Leaving the readers behind: An age of corporate newspapering*. Fayetteville: University of Arkansas Press.

Robert, M. *Framing toward clarification of fractured paradigm*. Journal of Communication 43 (autumn 1993): pp. 51 - 58.

Rodman, G.1981. *Mass media issues. Analysis and debates*. USRA. USA. http://aclib.sabinet.co.za/about.html.

Ross, E. 1997. The suppression of important news <u>In</u> Golding, P. & Murdock, G. (eds). *The political economy of media* vol. 11 Cheltenham: Edward Publishing Limited.

SAARF. 2008. *All media product survey: Newspaper* readership 2007 - 2008. Available at www.saarf.co.za

Scheufele, B. 2006. *Frames, schemata and news reporting*. Communications (31), pp. 65 - 83

Schlesinger, P. 1978. *Putting "reality" together -* BBC News. Great Britain: Anchor Press.

Schudson, M. 1991. Communication concepts 3. Gatekeeping. New Dehli: Sage.

Schudson, M. 1992. Was there ever a public sphere? <u>In</u> Calhoun, C. 1992. *Habermas and the public sphere*. Boston: MT Press.

Schudson, M. 1997. The sociology of news production. <u>In</u> Dan Berkowitz (ed) *Social meanings of news: A text reader*. California: Sage.

Schudson, M. 2000. The Sociology of News Production Revisited <u>In</u> Curran, J. & Guerevitch, M. (eds). *Media and Society*. California: Sage.

Schudson, M. 2003. *The sociology of news*. New York: WW Norton and Company.

Shoemaker, P. 1997. A new gatekeeping model <u>In</u> Dan Berkowitz (ed.) *Social meanings of news: A text reader.* California: Sage.

Shoemaker, P. & Reese, S.F. 1991. *Influences on content from individual media workers in mediating the message: Theories of influences on mass media content*. New York: Longman.

Shoemaker, P. & Reese, S.F. 1996. *Mediating the message: Theories of influences on mass media content*. New York: Longman.

Siebert, F.S., Pieterson, T. & Schramm, W. 1956. Four theories of the press. University of Illinois Press, Urbana, Urbana.

Sparks, C. 1991. Goodbye, Hildy Johnson: The vanishing 'serious press'. <u>In</u> Dahlgren, P. & Sparks, C. eds. *Communication and citizenship: Journalism in the new media age* pp. 58 - 74. London: Routledge.

Sparks, C. 1992. Popular journalism: Theories and practice. <u>In</u> Dahlgren, P. and Sparks, C. eds. *Journalism and popular culture* pp. 24 – 44. London: Sage.

Sparks, C. 2000. *The popular press and political democracy. Media, culture and society*, (10): pp. 209 - 223. London: Sage.

Splichal, S. 1993. Communication and democracy. New Jersey: Ablex Publishing

Staab, J.F. 1990. *The role of news factors in the news selection: A theoretical reconsideration*. European journal of communication vol. 5 pp. 423 - 443. London, Newbury Park and New Delhi: Sage.

Steenveld, L, 2004. *Transforming the media: A cultural approach* .vol. 1, pp. 92.-114. Steyn, E., De Beer, A.S. 2004. *The level of journalism skills in South African media: A reason for concern within a developing democracy?* Journalism studies vol. 5, No. 3, 2004, pp. 387 - 397.

Stokes, J. 2003. How to do cultural studies. London: Sage.

Stokke, O. 1971. *Mass communication in Africa - freedoms and functions*. Sweden: Uppsala.

Stokke, O. 2000. *Reporting Africa*. Sweden: The Scandinavian Institute of African Studies: Uppsala.

Strelitz, L. & Steenveld, L. 2005. *Thinking about South African tabloid newspapers*. Department of Journalism and Media Studies. Grahamstown: Rhodes University.

Switzer, L. 2000. South Africa's resistance press in perspective <u>In</u> Switzer, L. & Adhikari, M. 2000. *South Africa's resistance press: Alternative voices in the last decade under apartheid* pp. 1 - 75 Ohio: University Press.

Tahsoh, P.S. 2000. Media ownership and control in Cameroon in the information age: A colonial legacy. <u>In</u> M'Bayo, R.T., Onwumechili, C. & Mwankwo, N, (eds). *Press and Politics in Africa*. Lewiston: Edwin Mellen Press.

Tettey, W.J. 2001. The media and democratisation in Africa contributions constraints and concerns of the private press <u>In</u> *Media Culture*, vol 23: pp. 5 -31.

Thetela, P. 2001. Critique discourses and ideology in newspaper reports: a discourse analysis of the South African press reports on the 1998 SADC's military intervention in Lesotho. Discourse and Society 12 (3), pp. 347 - 370.

Thompson, J.B. 1993. The *media and modernity: A social theory of the media*. Stanford: Polity Press.

Thompson, J.B. 1994. *The media and modernity: A social theory of the media.* Stanford: Polity press.

Tomaselli, R. 2005. The role of the public broadcaster in changing the fourth estate. Essays on South African Journalism ed. by Hadland, A. Cape Town: HSRC.

Tomaselli, R. & Tomaselli K.G. 2001 Transformation, nation building and the South African media, 1993 -1999. In Tomaselli, K. & Dunn, H. 2001. *Media, democracy and renewal in South Africa*. Springs: International Academic Publishers.

Tomaselli, T. 1995. *Moving towards democracy: the South African Broadcasting Corporation and the 1994 election.* Media, Culture and Society 17: pp. 577 - 601.

Tomaselli, T. 1996. Reconceptualising public service broadcasting. <u>In Visions and voices: Audio-visual media in the new South Africa</u>, Johannesburg: Open Window Network, pp. 17 - 23.

Tsedu, M. 2002. Sanef's 2002 South African national journalism skills audit: An

introduction. Ecquid Novi 23 (1) pp.5 - 7.

Tuchman, G. 1978. Making news. New York: Free Press.

Tuchman, G, 1991. *Handbook of qualitative methodologies for mass communication*. Research. New York: Free Press.

Underwood, D. 1995. When MBAs rule the newsroom: How the marketers and managers are reshaping today's media. New York: Columbia University Press.

Van Dijk, T. 1988. *New analysis: Case studies of international and national stories in the press*. Hillside, NJ: Lawrence Erbaum Publishers.

Van Dijk, T. 1997. *News analysis: Case studies of international and national stories in the press.* Hillsdale, NJ: Lawrence Erlbaum Publishers.

Waldhal, R. 2004. *Politics and persuasion: Media coverage of Zimbabwe's 2004 election.* Harare: Frontline Publishing.

Wasburn, P.C 2002. The social construction of international news: We're talking about them, they are talking about us. London: Praeger Publishers.

Wasserman, H. & De Beer, A. 2005. Which public? Whose interest? The South Africa media and its role during the first year of democracy, Criticalarts, 19 (1 and 2) pp. 36-51.

Watson, J.2007. *Representing realities: An overview of news framing*. Keio Communication Review No. 29, pp.107 -131.

Weber, R.P. 1990. Basic content analysis. Thousand Oaks, CA: Sage

Wigston, D. (2001). A South African media map. In *Media studies, Vol. 1: Institutions, Theories and Issues*, ed. P.J. Fourie. Lansdowne: Juta Education.

Wigston, D. 2006. A South African media map. In *Media Studies*, vol. 1: Institutions, Theories and Issues, ed. by P.J. Fourie. Lansdowne: Juta Education.

Wilbraham, L. 1995. *Thematic content analysis: Panacea for the ills of intentioned opacity of discourse analysis.* Paper presented at the 1st Annual Qualitative Methods Conference. *A spanner in the works of the factory of truth,* 20 October 1995, University of Witwatersrand, South Africa. Retrieved November 20, 2005 from the World Wide Web: http://www.criticalmethods.org/spanner.htm.

Wilcox, D.L. *Mass media in black Africa: Philosophy and control.* London: Praeger Publishers.

Williams, K. 2003. *Understanding media theory*. London: Arnold.

Wimmer, R.D. & Dominick, J.R. 1991 (3rd ed.) *Mass media research: An introduction*. Belmont, Calif: Wadsworth.

Zegeye, A. & Harris, R. 2002. *Media, identity and the public sphere in post-apartheid South Africa: An introduction*. African & Asian Studies vol. 1 (4) pp. 239 - 254.

Newspaper Articles

Sowetan All front page stories and editorials between 01 January 2001 and

14 June 2005.

Natal Witness All front page stories and editorials between 01 January 2001 and

14 June 2005.

The Citizen All front page stories and editorials between 01 January 2001 and

14 June 2005.

The Star All front page stories and editorials between 01 January 2001 and

14 June 2005.

Internet Sources

www.sabinet.co.za

http://www.southafrica.info/

http://www.journalism.co.za

http://www.saarf.co.za

http://www.criticalmethods.org.

http://en.wikipedia.org/wiki/Schabir_Shaik_trial

http://www.mediclubsouthafrica.com

Audit Bureau of Circulation January 2009 - March 2009

APPENDIX A





| | | | Subscriptions | | | 1 | | | | | |
|---------------------------------|------------|--------------------|----------------|----------|-------------------------------|---------------|------------------------|-------|-------------------------|--------------------------|---|
| Publicat ion Name | Frequency | Cover Price | Individu al | Business | Travel & Comm ercial | Copy Sales | Third Party Bulk | PMIE | Digital Edition s | TOTAL CIRCULATI ON | Correspon ding Previous Period |
| Beeld, (Daily) | MD, Mo-Fr | R 4.90 | 47,042 | 2,809 | | 43,93 2 | 423 | 1,708 | | 95,914 | 105,149 |
| Bukedde | MD, Mo-Sat | UGShs 800.00 | 276 | | | 19,45 2 | | | | 19,728 | 14,401 |
| Burger, Die (Daily) | MD, Mo-Fr | R 4.50/3.4 0 | 32,731 | | 970 | 42,42 1 | 3,926 | 172 | | 80,220 | 84,545 |
| Burger, Die (Daily) | Oos Kaap | | | | | | | | | 14,758 | 17,123 |
| Burger, Die (Daily) | Wes Kaap | | | | | | | | | 65,462 | 67,422 |
| Business Day | MD, Mo-Fr | R 9.50 | 3,986 | 21,379 | 4,456 | 9,636 | 59 | | | 39,516 | 41,975 |
| Cape Argus | MD, Mo-Fr | R 4.50 | 18,721 | 1,254 | | 30,13 4 | 1,266 | 9,177 | | 60,552 | 70,401 |
| Cape Times | MD, Mo-Fr | R 5.20 | 14,696 | 3,606 | | 29,21 2 | | | | 47,514 | 50,621 |
| Citizen, The (Daily) | MD, Mo-Fr | R 4.00 | 3,595 | 4,208 | 9,906 | 48,12 8 | 1,667 | 355 | 13 | 67,872 | 71,324 |
| Daily Dispatch | MD, Mo-Fr | R 4.00 | 3,684 | 1,452 | 49 | 27,22 9 | 505 | | | 32,919 | 32,194 |
| Daily Monitor | MD, Mo-Sat | UGShs 1000.00 | 671 | | | 19,23 1 | | | | 19,902 | 24,065 |
| Daily Nation | MD, Mo-Sat | KShs 35.00 | 746 | | | 162,3 46 | | | | 163,092 | 162,813 |
| Daily News | MD, Mo-Fr | R 4.00 | 21,200 | 479 | 578 | 16,50 0 | 120 | 3,397 | | 42,274 | 52,018 |
| Daily Sun | Mo-Fr | R 2.00 | 131 | | | 507,1 97 | | | | 507,328 | 499,436 |
| Diamond Fields Advertiser | MD, Mo-Fr | R 3.20 | 1,238 | 497 | 500 | 7,208 | | 735 | | 10,178 | 10,138 |
| Herald, The | MD, Mo-Fr | R 4.50 | 6,079 | | | 19,42 3 | | | | 25,502 | 25,978 |
| Isolezwe | MD, Mo-Fr | R 2.30 | 666 | 249 | 95 | 101,4 44 | | | | 102,454 | 99,098 |
| Mercury, The | MD, Mo-Fr | R 5.00 | 14,582 | 2,573 | 1,595 | 17,04 7 | 159 | 1,124 | | 37,080 | 38,916 |
| Namibian, The | MD, Mo-Fr | Nam\$ 3.00 | | 538 | | 25,13 2 | | | | 25,670 | 23,438 |
| New Vision | MD, Mo-Sat | UGShs 1000.00 | 1,220 | | | 31,54 0 | | | | 32,760 | 32,653 |

| The | | | 244,451 | 65,073 | 26,302 | 8 1, 528 , | 13,636 | 50,71 | 13 | 1,929,184 | 1,983,383 |
|----------------------|------------|---------------|---------|--------|--------|----------------------|--------|------------|----|-----------|-----------|
| Witness, | MD, Mo-Fr | R 4.40 | 10,625 | | | 11,30 | 254 | | | 22,187 | 23,187 |
| Volksblad (Daily) | Mo-Fr | R 4.00 | 11,297 | 1,017 | 200 | 14,07 6 | -16 | | | 26,574 | 28,176 |
| Swaziland | MD, Mo-Fr | E 2.33 | 788 | | | 25,85 5 | | | | 26,643 | 24,821 |
| Taifa Leo | MD, Mo-Sat | KShs 25.00 | 18 | | | 11,19 6 | | | | 11,214 | 13,965 |
| Star, The (Daily) | MD, Mo-Fr | R 5.00 | 43,260 | 17,935 | 6,700 | 81,79 8 | 1,202 | 16,31 9 | | 167,214 | 178,294 |
| Sowetan | Mo-Fr | R 3.50 | 1,304 | 5,625 | 1,253 | 104,1 79 | 3,527 | 14,17 2 | | 130,060 | 145,173 |
| Son (Daily) | MD, Mo-Fr | R 2.00 | | | | 107,8 99 | | | | 107,899 | 102,402 |
| Pretoria News | MD, Mo-Fr | R 4.30 | 5,895 | 1,452 | | 15,47 6 | 544 | 3,551 | | 26,918 | 28,202 |

Report Date: Revision: 55.0 Q1 2009 5/19/2009

Daily Newspapers: January 2009 - March 2009 (Summary)

138